UC San Diego UC San Diego Electronic Theses and Dissertations

Title

Combinatorics of intersecting set systems

Permalink https://escholarship.org/uc/item/87t0v7rk

Author O'Neill, Jason

Publication Date 2022

Peer reviewed|Thesis/dissertation

UNIVERSITY OF CALIFORNIA SAN DIEGO

Combinatorics of intersecting set systems

A dissertation submitted in partial satisfaction of the

requirements for the degree

Doctor of Philosophy

 in

Mathematics

by

Jason O'Neill

Committee in charge:

Professor Jacques Verstraëte, Chair Professor Fan Chung Graham Professor Shachar Lovett Professor Andrew Suk

2022

Copyright

Jason O'Neill, 2022

All rights reserved.

The dissertation of Jason O'Neill is approved, and it is acceptable in quality and form for publication on microfilm and electronically.

University of California San Diego

2022

TABLE OF CONTENTS

Dissertation	n Appr	oval Page	iii
Table of Co	ontents		iv
Acknowledg	gement	s	vii
Vita			ix
Abstract of	the Di	issertation	xi
Chapter 1	Introd	luction	1
	1.1	Frequently used notation	2
	1.2	Intersecting families	4
	1.3	Set families with few intersections of odd size	10
	1.4	Cross family intersections	17
	1.5	The Bollobás set pairs inequality	21
	1.6	Intersection Saturation	27
Chapter 2	Non-t	rivial d -wise intersecting families $\ldots \ldots \ldots \ldots \ldots \ldots \ldots$	35
	2.1	Preliminaries	37
	2.2	Structure of d -wise intersecting families $\ldots \ldots \ldots \ldots \ldots$	39
	2.3	The Delta System Method	45
	2.4	Proof of Theorem 2.0.1	47

Chapter 3	Super	saturati	on for Oddtown and Eventown	 •	 	 . 49
	3.1	Proof	of Theorem 1.3.6	 	 	 . 50
	3.2	Proof	of Theorem 1.3.4	 •••	 	 . 52
		3.2.1	The case of $s = 1$. 52
		3.2.2	The case of $s = 2$. 53
Chapter 4	On <i>k</i> -	wise od	dtown problems	 	 	 . 56
	4.1	Hyper	graph Covering Problems	 •••	 	 . 57
	4.2	Proof	of Theorem 1.4.4	 •••	 	 . 60
		4.2.1	Two families	 • •	 	 . 60
		4.2.2	More than two families	 •	 	 . 61
	4.3	Proof	of Theorem 1.4.5	 	 	 . 62
		4.3.1	Upper bound on $f_{k,t}(n)$	 • •	 	 . 63
		4.3.2	Upper bound on $b_{k,t}(n)$. 64
Chapter 5	A gen	eralizat	ion of the Bollobás set pairs inequality	 • •	 	 . 66
	5.1	Proof	of Theorem 1.5.2	 •••	 	 . 67
	5.2	Proof	of Theorem 1.5.4	 	 	 . 73
		5.2.1	Upper bound on $f_{k,t}(n)$. 73
		5.2.2	Lower bound on $f_{k,2}(n)$	 •	 	 . 76
		5.2.3	Lower bound on $f_{k,k}(n)$. 78
Chapter 6	Inters	ection s	aturated set systems	 • • ·	 	 . 80

6.1	Clique partitions	81
6.2	Rödl Nibble	81
6.3	Large Sets of Designs	82
6.4	Proof of Theorem 1.6.3	83
6.5	Proof of Theorem 1.6.4	86
6.6	Proof of Theorem 1.6.5	87
6.7	Proof of Theorem 1.6.6	88
References		90

ACKNOWLEDGEMENTS

I'd like to start by thanking my advisor Jacques Verstraëte for his immense contributions to my mathematical and professional development. Our weekly meetings often contained deep mathematical insights and helpful career advice, which have helped shaped the mathematician that I am today. I would also like to thank my committee members for their support throughout the degree process. The Theory lunch seminar, hosted by Shachar and the UCSD CS Theory group, provided an excellent source of new ideas and stimulating conversations regarding the connections between discrete mathematics and theoretical computer science. I very much enjoyed Andrew's course on discrete geometry and learning about spectral graph theory from Fan. Lastly, I want to thank Ron Graham for his time on my committee and for telling me about his and Peter Frankl's juggling accomplishments, which eased my nerves prior to my advancement to candidacy talk.

There are also many other within the Combinatorics community at UC San Diego who I would like to acknowledge. Brendon, thank you for the interesting courses in algebraic combinatorics and for demonstrating a devotion to teaching that I hope to exhibit throughout my career. I would also like to thank Brendon, Andrew and Jeff for their letters of recommendation. Thank you Ruth, Gwen, Daniël, Jiaxi, Sam, Nicholas, Emily, Max for your commitment to ABACUS. I am very lucky to have had many supportive friends throughout my time in San Diego. Thank you Sam, Samir, Tom, Felipe, Vaki, Eva, Jacob, Vatsa, Greg for making this experience so enjoyable. Thank you Rachel for always making me smile after a long day of usually unsuccessful mathematical research. Finally, I would like to thank my family for their constant love, support and encouragement. I could not have done this without you, Susan, Geoff, Kevin and Peter!

Chapter 2 contains material from [OV21b]: Jason O'Neill and Jacques Verstraëte, "Non-trivial *d*-wise intersecting families", J. Combin. Theory Ser. A, 178, 2021. The dissertation author was one of the primary investigators and authors of this paper.

Chapter 3 contains material from [O'N21]: Jason O'Neill, "Towards supersaturation for oddtown and eventown", arxiv:2109.09925, 2021. The dissertation author was the sole author of this paper.

Chapter 4 contains material from [OV20b]: Jason O'Neill and Jacques Verstraëte, "A note on k-wise oddtown problems", arxiv:2011.09402, 2020. The dissertation author was one of the primary investigators and authors of this paper.

Chapter 5 contains material from [OV21a]: Jason O'Neill and Jacques Verstraëte, "A generalization of the Bollobás set pairs inequality", The Electronic Journal of Combinatorics, 28(3), 2021. The dissertation author was one of the primary investigators and authors of this paper.

Chapter 6 contains material from [OV20a]: Jason O'Neill and Jacques Verstraëte, "A note on intersection saturation", which is currently being prepared for submission for publication. The dissertation author was one of the primary investigators and authors of this paper.

VITA

2017	Bachelor of Science, University of California Los Angeles
2019	Candidate of Philosophy, University of California San Diego
2022	Doctor of Philosophy, University of California San Diego

PUBLICATIONS

C. Buchanan, A. Clifton, E. Culver, J. Nie, J. O'Neill, P. Rombach, M. Yin, "Odd covers of graphs", arxiv:2202.09822, 2022.

Z. Füredi, D. Mubayi, J. O'Neill, J. Verstraëte, "Extremal problems for pairs of triangles",Journal of Combinatorial Theory Ser. B, 155, 2022.

J. O'Neill, "Towards supersaturation for oddtown and eventown", arxiv:2109.09925, 2021.

J. O'Neill, "On the poset and asymptotics of Tesler matrices", The Electronic Journal of Combinatorics, 25(2), 2018.

J. O'Neill, S. Spiro, "Saturation problems for convex geometric hypergraphs", arxiv:2109.09931, 2021.

J. O'Neill, J. Verstraëte, "A note on k-wise oddtown problems", arxiv:2011.09402, 2020.

J. O'Neill, J. Verstraëte, "Non-trivial *d*-wise intersecting families", J. Combin. Theory Ser. A, 178, 2021.

J. O'Neill, J. Verstraëte, "A generalization of the Bollobás set pairs inequality", The Electronic Journal of Combinatorics, 28(3), 2021.

J. O'Neill, J. Verstraëte, "Intersection saturated set systems" – In preparation

ABSTRACT OF THE DISSERTATION

Combinatorics of intersecting set systems

by

Jason O'Neill

Doctor of Philosophy in Mathematics

University of California San Diego, 2022

Professor Jacques Verstraëte, Chair

In this dissertation, we examine various problems in extremal set theory, which typically entails maximizing the size of a collection of subsets, or set family, given intersection constraints. For instance, the classical Erdős-Ko-Rado theorem (1961) establishes the largest set family of size k subsets of an n element set which is *intersecting* (i.e. has the property that any two sets have at least one element in common). The largest such intersecting family is the collection of all size k subsets that contain a fixed element, which is commonly referred to as *the star*. Answering a question of Erdős, Ko and Rado, Hilton and Milner (1967) determined the largest intersecting set family which is not isomorphic to a sub-collection of the star. This dissertation settles a conjecture of Hilton and Milner (1967) on the largest set family, for each integer $d \ge 3$, of size k subsets of an n element set which has the property that any d sets have at least one element in common and is not isomorphic to a sub-collection of the star. We also consider various combinatorial results on set families with restricted intersection properties. In particular, we prove generalizations of both the *Bollobás two family* theorem and the *Oddtown and Eventown* theorems.

Chapter 1

Introduction

This dissertation studies various problems regarding intersecting collections of subsets of a finite set. Problems in *extremal set theory* typically involve maximizing the size of a collection of subsets, or set family, given set theoretic constraints. Perhaps the most famous theorem in the area is the Erdős-Ko-Rado [EKR61] Theorem on the size of the largest set family which is *intersecting* (i.e. any two members from the set family have at least one element in common). Also within this general framework are the foundational Oddtown and Eventown results, which demonstrate the *linear algebra method*, and establish the largest set family for which all pairwise intersections contain an even number of elements and the sets themselves have odd and even number of elements respectively.

There are also other results in this dissertation which fall outside the typical framework of maximizing the size of a set family subject to set theoretic constraints. In particular, given an integer k at least three, this dissertation explores k set family analogs which are generalizations of Bollobás [Bol65] *two family theorem* and Babai and Frankl's [BF20] *Bipartite Oddtown theorem*. Finally, we also explore a problem on determining how *small* a set family can be provided it satisfies various set theoretic constraints and is maximal with respect to these constraints. We will now establish commonly used notation in order to give a more accurate description of the main results of this dissertation.

1.1 Frequently used notation

We first describe some of the main objects of study in the field:

- Interval notation: Given positive integers $n \ge m$, we use the notation $[m, n] = \{m, m+1, \ldots, n\}$ and in the case where m = 1, we let [n] = [1, n].
- Non-uniform set systems: Given a positive integer $n \ge 1$, we let $2^{[n]} = \{A \subseteq [n]\}$ denote the collection of subsets of [n]. We refer to collections $\mathcal{A} \subseteq 2^{[n]}$ as set systems and use calligraphic font to denote set systems
- k-uniform set systems: Given positive integers $n \ge k \ge 1$, we let $\binom{[n]}{k} = \{A \subseteq [n] :$ $|A| = k\}$. If a set systems $\mathcal{F} \subseteq 2^{[n]}$ also satisfies the property that |F| = k for all $F \in \mathcal{F}$, then we refer to $\mathcal{F} \subseteq \binom{[n]}{k}$ as a k-uniform set system

Given a particular subset $A \subseteq [n]$, we use the following notation to describe objects which correspond to the underlying set:

- Characteristic vector: Given A ⊆ [n], we let v_A ∈ F_n be given by (v_A)_k = 1 if and only if k ∈ A
- Set Complement: Given a subset A ⊆ [n], let A^c := [n] \ A be the complement of A in [n]. Further given a set family A = {A₁,..., A_m} ⊆ 2^[n], let A^c = {A₁^c,..., A_m^c} ⊆ 2^[n] denote the set of corresponding complements.
- Link set system: Given $A \subseteq [n]$ and $\mathcal{F} \subseteq 2^{[n]}$, we let $\mathcal{F}(A) = \{B \subseteq [n] : B \cup A \in \mathcal{F}\}$ denote the link of A in \mathcal{F} and we refer to $|\mathcal{F}(A)|$ as the *degree* of A in \mathcal{F}
- Linear functional: Given $A \subseteq [n]$, let $\chi_A : \mathbb{F}_2^n \to \mathbb{F}_2$ be given by the inner product $w \mapsto \langle w, v_A \rangle$

• Perpendicular Space: Given the vector space \mathbb{F}_2^n with the standard inner product and vector subspace U, let $U^{\perp} = \{v \in \mathbb{F}_2^n : \langle v, u \rangle = 0 \ \forall u \in U\}.$

We also use the following asymptotic notation and numerical expressions:

- For positive integers k ≤ n, let (n)_(k) = (n)(n − 1) · · · (n − k + 1) denote the falling factorial.
- Asymptotic notation: For functions f, g : N → R⁺, f = o(g) if lim_{n→∞} f(n)/g(n) = 0, and f = O(g) and g = Ω(f) if there exists c > 0 such that f(n) ≤ cg(n) for all n ∈ N. If f = O(g) and g = O(f), we write f = Θ(g).

We now briefly describe the main results proven in this dissertation.

1.2 Intersecting families

In this section, we explore the classical literature pertaining to *intersecting set* families. A set family $\mathcal{F} \subseteq 2^{[n]}$ is called *intersecting* if for all $F_1, F_2 \in \mathcal{F}$, the sets F_1 and F_2 have at least one element in common (i.e. $F_1 \cap F_2 \neq \emptyset$). The study of intersecting set families was initiated by the foundational Erdős-Ko-Rado Theorem [EKR61]:

Theorem 1.2.1 (Erdős-Ko-Rado [EKR61]). Let $\mathcal{F} \subseteq 2^{[n]}$ be an intersecting family. Then $|\mathcal{F}| \leq 2^{n-1}$. Further, if $n \geq 2k$ and $\mathcal{F} \subseteq {\binom{[n]}{k}}$ is an intersecting family, then $|\mathcal{F}| \leq {\binom{n-1}{k-1}}$.

Proof. Let $\mathcal{F} \subseteq 2^{[n]}$ be an intersecting family. It then follows that $\mathcal{F}^c \cap \mathcal{F} = \emptyset$ as if $A \in \mathcal{F}^c \cap \mathcal{F}$, then $A, A^c \in \mathcal{F}$; a contradiction. Hence, as $\mathcal{F} \cup \mathcal{F}^c \subseteq 2^{[n]}$ and $|\mathcal{F}^c| = |\mathcal{F}|$,

the result follows by noting that

$$2|\mathcal{F}| = |\mathcal{F}| + |\mathcal{F}^c| = |\mathcal{F} \cup \mathcal{F}^c| \le |2^{[n]}| = 2^n.$$

For the second statement, we let C_n denote the collection of long cycle permutations of [n] arranged in a circle. Given $\mathcal{F} \subseteq {\binom{[n]}{k}}$ which is intersecting and $\sigma \in C_n$, let \mathcal{F}_{σ} denote the sets in \mathcal{F} which appear as cyclically consecutive elements of σ . It then follows that $|\mathcal{F}_{\sigma}| \leq k$. Without loss of generality, let $\sigma = (1, 2, ..., n)$ and suppose $[k] \in \mathcal{F}_{\sigma}$. Setting $A_i = [i, i+k-1]$ and utilizing $n \geq 2k$, it is straightforward to see that $A_i \in \mathcal{F}_{\sigma}$ implies that $i \geq n-k+2$ or $i \leq k$. Consider the pair $\{A_{n-k+i}, A_i\}$ for $i \in [2, k]$. As $A_{n-k+i} \cap A_i = \emptyset$ and \mathcal{F} is an intersecting family, $|\mathcal{F}_{\sigma} \cap \{A_{n-k+i}, A_i\}| \leq 1$. We then recover the desired bound $|\mathcal{F}_{\sigma}| \leq k$. A double counting argument now gives

$$|\mathcal{F}| \cdot k! (n-k)! = \sum_{F \in \mathcal{F}} |\{\sigma \in \mathcal{C}_n : F \in \mathcal{F}_\sigma\}| = \sum_{\sigma \in \mathcal{C}_n} |\{F \in \mathcal{F} : F \in \mathcal{F}_\sigma\}| \le (n-1)! \cdot k.$$

Dividing both sides by k!(n-k)!, we recover

$$|\mathcal{F}| \le \frac{(n-1)! \cdot k}{k!(n-k)!} = \frac{(n-1)!}{(k-1)!(n-k)!} = \binom{n-1}{k-1}.$$

The proof of the second statement was noted by Katona [Kat72] and uses a method commonly referred to as *Katona's circle method*. The original proof by Erdős, Ko and Rado used a method referred to as *shifting* together with an induction argument.

An interesting component of Theorem 1.2.1 is that there are two substantially different intersecting families of size 2^{n-1} . First, we may take the set family $\mathcal{A}_1 = \{A \subseteq [n] : 1 \in A\}$ – this is intersecting as each pairwise intersection contains the element 1. Second, for odd n, we may take the set family $\mathcal{A}_2 = \{A \subseteq [n] : |A| \ge \frac{n-1}{2}\}$ – this is intersecting as the subsets are too large to have two of them be pairwise disjoint and hence they necessarily contain at least one element in common. However, in the k-uniform case we have a drastically different situation where all of intersecting families of size *nearly* $\binom{n-1}{k-1}$ have similar structure.

The set family $\mathcal{F}_1 = \{F \in {\binom{[n]}{k}} : 1 \in F\}$ is an intersecting family of size ${\binom{n-1}{k-1}}$. It is straightforward to see that one can remove a few sets from \mathcal{F}_1 and still have an intersecting family of size *nearly* ${\binom{n-1}{k-1}}$. To rule out this trivial example, Erdős, Ko and Rado defined a set family \mathcal{F} to be *non-trivial* if $\bigcap_{A \in \mathcal{F}} A = \emptyset$. In other words, in a non-trivial set family, for each $i \in [n]$, there exists $X_i \in \mathcal{F}$ so that $i \notin X_i$. Erdős, Ko and Rado asked for the maximum size of a non-trivial intersecting family \mathcal{F} of k-element subsets of [n]. This question was answered by Hilton and Milner [HM67].

Theorem 1.2.2 (Hilton-Milner). Let n > 2k and $k \ge 3$. If $\mathcal{F} \subseteq {\binom{[n]}{k}}$ is a non-trivial intersecting family, then $|\mathcal{F}| \le {\binom{n-1}{k-1}} - {\binom{n-k-1}{k-1}} + 1$.

This may be viewed as a *stability* version of the Erdős-Ko-Rado Theorem, in the sense that an intersecting family of size larger than the bound in Theorem 1.2.2 is necessarily a subfamily of the extremal intersecting family – i.e. there a point in the intersection of all sets in the family. Further, there is a non-trivial intersecting family obtaining the bound in Theorem 1.2.2:

$$\mathcal{HM}(k,2) = \{[2,k+1]\} \cup \{A \in \binom{[n]}{k} : 1 \in A, A \cap [2,k+1] \neq \emptyset\}.$$

Hilton and Milner [HM67] considered an extension of Theorem 1.2.2 to *d*-wise intersecting families: a set family $\mathcal{F} \subseteq 2^{[n]}$ is *d*-wise intersecting if any set of *d* distinct sets in \mathcal{F} have non-empty intersection, with the case d = 2 corresponding to intersecting families. In the case where $\mathcal{F} \subseteq 2^{[n]}$ is a non-trivial *d*-wise intersecting family, there is a natural extension to the family of all sets which contain a fixed element. Observe that the following set family is non-trivial *d*-wise intersecting:

$$\mathcal{A}(d) = \{ A \subseteq [n] : |A \cap [d+1]| \ge d \}.$$

Brace and Daykin [BD71] proved that \mathcal{A}_d is the largest such set family:

Theorem 1.2.3. Let $\mathcal{A} \subseteq 2^{[n]}$ be a non-trivial d-wise intersecting family. Then $|\mathcal{A}| \leq |\mathcal{A}(d)| = (d+2)2^{n-d-1}$.

In the case of k-uniform non-trivial d-wise intersecting families, there are two natural extensions to the set families $\mathcal{A}(d)$ and $\mathcal{HM}(k, 2)$. These set families were constructed by Hilton and Milner [HM67]. Moreover, Hilton and Milner [HM67] conjectured that for large enough n the extremal non-trivial d-wise intersecting family of k-sets in [n], up to isomorphism, is one of the following two families:

$$\mathcal{A}(k,d) = \{A \in \binom{[n]}{k} : |A \cap [d+1]| \ge d\}$$
$$\mathcal{H}\mathcal{M}(k,d) = \{A \in \binom{[n]}{k} : [d-1] \subseteq A, A \cap [d,k+1] \neq \emptyset\} \cup \{[k+1] \setminus \{i\} : i \in [d-1]\}.$$

In Chapter 2, we prove the conjecture of Hilton and Milner (1967):

Theorem 1.2.4 (O'N-Verstraëte). Let k, d be integers with $2 \le d < k$. Then there exists $n_0(k, d)$ such that for $n \ge n_0(k, d)$, if \mathcal{F} is a non-trivial, d-wise intersecting family of k-element subsets of [n], then

$$|\mathcal{F}| \le \max\{|\mathcal{HM}(k,d)|, |\mathcal{A}(k,d)|\}.$$

The defining property of non-trivial families $\mathcal{F} \subseteq 2^{[n]}$ is the fact that for each element $i \in [n]$, there exists at least one set $X_i \in \mathcal{F}$ so that $i \notin X_i$. This notion of avoiding singletons can be extended to avoiding sets of larger size. Given a set system $\mathcal{F} \subseteq 2^{[n]}$, the *transversal number*, denoted $\tau(\mathcal{F})$, is the minimum positive integer s so that there exists a set $S \in {[n] \choose s}$ where $S \cap F \neq \emptyset$ for all $F \in \mathcal{F}$. As such, given a set family $\mathcal{F} \subseteq 2^{[n]}$ with $\tau(\mathcal{F}) = s + 1$, for each $S \in {[n] \choose s}$, there exists $X_S \in \mathcal{F}$ with $S \cap X_S = \emptyset$.

In this language of transversal numbers, a set family $\mathcal{F} \subseteq 2^{[n]}$ is non-trivial if and only if $\tau(\mathcal{F}) \geq 2$. Further, a (non-empty) set family $\mathcal{F} \subseteq {\binom{[n]}{k}}$ is intersecting if and only if $\tau(\mathcal{F}) \leq k$ (where each set $F \in \mathcal{F}$ has the defining property). There has been substantial research on intersecting families (i.e. d = 2) with $\tau(\mathcal{F}) \geq 3$ and it would interesting to explore these larger transversal problems in the *d*-wise intersecting setting:

Problem 1.2.5. Given $n > k > d \ge 3$ integers, what is the maximum size of a d-wise intersecting family $\mathcal{F} \subseteq {\binom{[n]}{k}}$ with transversal number at least three?

In this setting, the set family $\mathcal{A}^3(k,d) = \{A \in {\binom{[n]}{k}} : |A \cap [2d+1]| \ge 2d-1\}$ is a d-wise intersecting family with $\tau(\mathcal{A}^3(k,d)) = 3$. Further, it is not hard to see that $\mathcal{A}_3(k,d)$ is a natural generalization of $\mathcal{A}(k,d)$. Moreover, there exists a natural generalization of

the set family $\mathcal{HM}(k,d)$ in this setting. Let $\mathcal{HM}^3(k,d) := \mathcal{F}_1 \sqcup \mathcal{F}_2 \sqcup \mathcal{F}_3$ where

$$\mathcal{F}_{1} = \left\{ A \in \binom{[n]}{k} : [2d-3] \subseteq A, e \subseteq A \text{ for some } e \in \binom{[2d-2, k+2]}{2} \right\}$$
$$\mathcal{F}_{2} = \left\{ [k+2] \setminus e \right\}_{e \in \binom{[2d-3]}{2}}$$
$$\mathcal{F}_{3} = \left\{ A \in \binom{[n]}{k} : |A \cap [2d-3]| = 2d-4, |A \cap [2d-2, k+2]| = k-2d+4 \right\}$$

We now conjecture the following analog of Theorem 1.2.4 to the setting with transversal number at least three:

Conjecture 1.2.6. Let n be sufficiently large and $d \ge 3$ with $2d - 1 \le k$. If $\mathcal{F} \subseteq {\binom{[n]}{k}}$ is a d-wise intersecting family with $\tau(\mathcal{F}) \ge 3$, then $|\mathcal{F}| \le \max\{|\mathcal{A}^3(k,d)|, |\mathcal{HM}^3(k,d)|\}$

Let us now explain the condition of $2d - 1 \leq k$ in Conjecture 1.2.6. Suppose 2d - 1 > k and hence $k \leq 2d - 2$ and that $\mathcal{F} \subseteq {\binom{[n]}{k}}$ is *d*-wise intersecting with $\tau(\mathcal{F}) \geq 3$. Without loss of generality, we may assume $[k] \in \mathcal{F}$. We may now find $e_1, \ldots, e_{d-1} \in {\binom{[k]}{2}}$ so that $e_1 \cup \cdots \cup e_{d-1} = [k]$. For each e_i , there exists $X_i \in \mathcal{F}$ so that $e_i \cap X_i = \emptyset$ by the condition $\tau(\mathcal{F}) \geq 3$. It now follows that $[k] \cap X_1 \cap \cdots \cap X_{d-1} = \emptyset$; which contradicts the *d*-wise intersecting condition on \mathcal{F} .

In the non-uniform setting, the Brace-Daykin theorem determines the size of the largest d-wise intersecting family with transversal number at least two. It would be interesting to explore an analogous statement for transversal number at least three:

Problem 1.2.7. Given $n > d \ge 3$ integers, what is the maximum size of a d-wise intersecting family $\mathcal{F} \subseteq 2^{[n]}$ with transversal number at least three?

There is a very natural candidate for Problem 1.2.5 which we conjecture is the largest such set family:

Conjecture 1.2.8. Let n be sufficiently large and $d \ge 3$. If $\mathcal{F} \subseteq 2^{[n]}$ is a d-wise intersecting family with $\tau(\mathcal{F}) \ge 3$, then $|\mathcal{F}| \le |\mathcal{A}^3(d)|$ where $\mathcal{A}^3(d)$ is defined as:

$$\mathcal{A}^{3}(d) = \{ A \subseteq [n] : |A \cap [2d+1]| \ge 2d-1 \}.$$

Intersecting set systems are required to have non-empty pairwise intersections. However, there are also interesting combinatorial results when you require the pairwise intersections to all have an even number of elements in common.

1.3 Set families with few intersections of odd size

In this section, we explore the classical Oddtown and Eventown problems on set systems for which all pairwise intersections have an even number of elements. A set family $\mathcal{A} \subseteq 2^{[n]}$ follows *oddtown rules* if the sizes of all sets in \mathcal{A} are odd and distinct pairs of sets from \mathcal{A} have an even number of elements in common.

Theorem 1.3.1 (Berlekamp [Ber69] and Graver [Gra75]). Let $\mathcal{A} \subseteq 2^{[n]}$ satisfy oddtown rules. Then $|\mathcal{A}| \leq n$.

Proof. Let $\mathcal{A} = \{A_1, A_2, \dots, A_m\} \subseteq 2^{[n]}$ be a set family which satisfies odd town rules. For each A_i , let $v_i = v_{A_i}$ denote the characteristic vector of A_i in \mathbb{F}_2^n . Further, let $\langle \cdot, \cdot \rangle$ denote the standard inner product in \mathbb{F}_2^n . It then follows that $\langle v_i, v_i \rangle = 1$ for all $i \in [m]$ as $|A_i|$ is odd. It also follows that $\langle v_i, v_j \rangle = 0$ for all $i \neq j \in [m]$ as $|A_i \cap A_j|$ is even. We now claim that the set $\{v_1, \ldots, v_m\}$ is linearly independent in \mathbb{F}_2^n . Let $0 = \epsilon_1 v_1 + \cdots + \epsilon_m v_m$ and observe that

$$0 = \langle \epsilon_1 v_1 + \dots + \epsilon_m v_m, v_i \rangle = \epsilon_i \langle v_i, v_i \rangle = \epsilon_i.$$

As such, it follows that $m \leq n$ as desired.

There are many constructions of n sets which satisfy odd town rules and hence show that the above theorem is best possible. We highlight the following constructions which will play a role in later discussions:

$$\mathcal{O}_1 = \{\{i\} : 1 \le i \le n\} \text{ and } \mathcal{O}_2 = \{A \in \binom{[n]}{3} : A \subseteq [4i - 3, 4i]\}$$

where $|\mathcal{O}_2| = n$ when 4|n.

Alternatively, a collection \mathcal{A} of even-sized subsets of an n element set follows eventown rules if all pairs of sets from \mathcal{A} have even-sized intersections.

Theorem 1.3.2 (Berlekamp [Ber69] and Graver [Gra75]). Let $\mathcal{A} \subseteq 2^{[n]}$ satisfy eventown rules. Then $|\mathcal{A}| \leq 2^{\lfloor n/2 \rfloor}$.

Proof. Let $\mathcal{A} = \{A_1, \ldots, A_m\} \subseteq 2^{[n]}$ satisfy eventown rules. As before, we let $v_i = v_{A_i}$ denote the characteristic vector in \mathbb{F}_2^n and consider $W = \operatorname{span}\{v_1, \ldots, v_m\}$. For each $w \in W$, it follows that $\langle w, v_i \rangle = 0$ for all characteristic vectors v_i . Hence, $W \subseteq W^{\perp}$. Noting the classical linear algebra result that $\dim(W) + \dim(W^{\perp}) = n$, it thus follows that $\dim(W) \leq \lfloor n/2 \rfloor$. The result then follows as $|W| \leq 2^{\lfloor n/2 \rfloor}$.

There are non-isomorphic eventown families which have $2^{\lfloor n/2 \rfloor}$ sets. In fact, by examining the proof of Theorem 1.3.2 we immediately obtain:

Corollary 1.3.3. Let $\mathcal{A} \subseteq 2^{[n]}$ satisfy eventown rules and $|\mathcal{A}| < 2^{\lfloor n/2 \rfloor}$. Then there exists $X \in 2^{[n]}$ so that $\mathcal{A} \cup \{X\}$ satisfies eventown rules.

As such, any maximal eventown family is maximum. Further, this allows us to construct many non-isomorphic eventown families of size $2^{\lfloor n/2 \rfloor}$. One such eventown construction, given $B_i = \{2i - 1, 2i\}$, is as follows:

$$\mathcal{E} = \left\{ \bigcup_{i \in J} B_i : J \in 2^{\left[\lfloor n/2 \rfloor\right]} \right\}.$$

The oddtown and eventown problems are foundational results which highlight the linear algebra method [BF20] in extremal combinatorics. There have been numerous extensions and variants of these results in the literature [OV20b, SV18, Vu99, SV05, Vu97, FO83, DFS83]. It is also worth noting that one can equivalently ask about extremal problems where sets of even (odd) size have an odd sized pairwise intersections. Given an extremal oddtown construction and adding an auxiliary element to each set in the family, we recover a set family of sets of even size for which each pair of distinct sets have an odd number of elements in common. This results in a natural duality between the oddtown and eventown problems and those which requires pairwise odd sized intersections. As such, we restrict our study in this dissertation to the classical Oddtown and Eventown problems.

Many problems in extremal combinatorics involve maximizing the size of an object

which satisfies combinatorial constraints. In this setting, we necessarily have at least one instance of the forbidden combinatorial constraint if we have one more than the maximum size. Supersaturation problems entail the *minimum* number of instances of the forbidden combinatorial constraint if we have more than the maximum size. The notion of *supersaturation* stems from the fact that, in certain instances, given one more than the extremal amount, we are guaranteed *many* instances of the forbidden property. In other words, the forbidden property saturates as we get slightly more than the maximum.

The first instance of supersaturation in extremal combinatorics comes from triangle-free graphs. Mantel [Man07] proved that any *n*-vertex triangle-free graph has at most $\lfloor n^2/4 \rfloor$ edges. This result is easily seen to be best possible by taking a complete bipartite graph with part sizes nearly equal. The Erdős-Rademacher problem in extremal combinatorics involves the minimum number of triangles in an *n*-vertex graph with $\lfloor n^2/4 \rfloor + s$ edges. Erdős [Erd62] proved that there are at least $s \cdot \lfloor n/2 \rfloor$ triangles in an *n*-vertex graph with $\lfloor n^2/4 \rfloor + s$ edges for $s \leq 3$.

In Chapter 3, we explore the *supersaturation* versions of the oddtown and eventown problems: given slightly more than n odd-sized subsets (resp. $2^{\lfloor n/2 \rfloor}$ even-sized subsets) of an n element set, how many pairs of sets from our collection must have an odd number of elements in common?

For a collection, or set family \mathcal{A} , let $op(\mathcal{A})$ denote the number of distinct pairs of sets $A, B \in \mathcal{A}$ for which $|A \cap B|$ is odd. For convenience, suppose n = 2k = 4land let X_1, \ldots, X_l be pairwise disjoint sets with $X_i = \{x_{1,i}, x_{2,i}, x_{3,i}, x_{4,i}\}$. Given X_i , let

$$A_{2i-1} = \{x_{1,i}, x_{2,i}\}, A_{2i} = \{x_{3,i}, x_{4,i}\}, B_{2i-1} = \{x_{1,i}, x_{4,i}\}, \text{ and } B_{2i} = \{x_{2,i}, x_{3,i}\}.$$
 Define

$$\mathcal{E}_1 = \left\{ \bigcup_{j \in J} A_j : J \subseteq [k] \right\} \quad \text{and} \quad \mathcal{E}_2 = \left\{ \bigcup_{j \in J} B_j : J \subseteq [k] \right\}.$$
(1.1)

Observe that both \mathcal{E}_1 and \mathcal{E}_2 are extremal (i.e. largest) eventown families and moreover for each $X \in \mathcal{E}_2 \setminus \mathcal{E}_1$, $\operatorname{op}(\mathcal{A} \cup \{X\}) = 2^{k-1}$. For $1 \leq s \leq 2^k - 2^l$, let $\mathcal{E}_1(s)$ denote a set family which contains s elements from $\mathcal{E}_2 \setminus \mathcal{E}_1$ together with \mathcal{E}_1 . It then follows that $|\mathcal{E}_1(s)| = 2^k + s$ and $\operatorname{op}(\mathcal{E}_1(s)) = s \cdot 2^{k-1}$. Our main result is that we are able to show this is best possible when s = 1 and s = 2:

Theorem 1.3.4. Let $n \ge 1$. If $\mathcal{A} \subseteq 2^{[n]}$ is a collection of sets of even size with $|\mathcal{A}| \ge 2^{\lfloor n/2 \rfloor} + 1$, then $\operatorname{op}(\mathcal{A}) \ge 2^{\lfloor n/2 \rfloor - 1}$. Moreover, if $\mathcal{A} \subseteq 2^{[n]}$ is a collection of sets of even size with $|\mathcal{A}| \ge 2^{\lfloor n/2 \rfloor} + 2$, then $\operatorname{op}(\mathcal{A}) \ge 2^{\lfloor n/2 \rfloor}$.

Given $3 \leq s \leq 2^{\lfloor n/2 \rfloor} - 2^{\lfloor n/4 \rfloor}$, we believe that the set family $\mathcal{E}_1(s)$ minimizes the number of pairwise intersections of odd size amongst all families of $2^{\lfloor n/2 \rfloor} + s$ sets of even size and conjecture:

Conjecture 1.3.5. Let $n \ge 1$. Fix $3 \le s \le 2^{\lfloor n/2 \rfloor} - 2^{\lfloor n/4 \rfloor}$. If $\mathcal{A} \subseteq 2^{[n]}$ is a collection of sets of even size with $|\mathcal{A}| \ge 2^{\lfloor n/2 \rfloor} + s$, then $\operatorname{op}(\mathcal{A}) \ge s \cdot 2^{\lfloor n/2 \rfloor - 1}$.

For oddtown, recall the extremal constructions \mathcal{O}_1 and \mathcal{O}_2 . For $1 \leq s \leq n$, let $\mathcal{O}_1(s)$ be any set family consisting of \mathcal{O}_1 together with exactly s sets from \mathcal{O}_2 . It then follows that $|\mathcal{O}_1(s)| = n + s$ and $\operatorname{op}(\mathcal{O}_1(s)) = 3s$. Hence there are set families with slightly more than n odd sets and few pairwise odd-sized intersection which in some sense

demonstrates that supersaturation results for odd town do not occur. In any case, we are able to prove this is best possible for s = 1:

Theorem 1.3.6. Let $n \ge 1$ and $\mathcal{A} \subseteq 2^{[n]}$ be a collection of sets of odd size with $|\mathcal{A}| \ge n+1$. Then $\operatorname{op}(\mathcal{A}) \ge 3$.

It is also worth noting that there exist extremal examples to Theorem 1.3.6 which do not contain an extremal oddtown family. To see this, let n = 5 and take $\mathcal{X} = \{X_1, \ldots, X_6\}$ where $X_1 = \{1, 2, 3\}, X_2 = \{1, 4, 5\}, X_3 = \{1, 2, 4\}, X_4 = \{1, 3, 5\}, X_5 = \{1, 3, 4\}, \text{ and } X_6 = \{1, 2, 5\}.$ Then $\operatorname{op}(\mathcal{X}) = 3$. Given $2 \leq s \leq n$, we believe that the set family $\mathcal{O}_1(s)$ minimizes the number of pairwise intersections of odd size amongst all families of n + s sets of odd size and conjecture:

Conjecture 1.3.7. Let $n \ge 1$ and fix $1 \le s \le n$. If $\mathcal{A} \subseteq 2^{[n]}$ is a collection of sets of odd size with $|\mathcal{A}| \ge n + s$, then $op(\mathcal{A}) \ge 3s$.

There are numerous extensions of the oddtown and eventown problems. In particular, Deza, Frankl, and Singhi [DFS83] used *multilinear polynomials* to give an extension of the Oddtown theorem to arbitrary primes:

Theorem 1.3.8 (Deza-Frankl-Singhi [DFS83]). Let p be prime and let $L \subseteq [0, p-1]$ be so that |L| = s. If $\mathcal{F} \subseteq 2^{[n]}$ is such that for all $F_1 \neq F_2 \in \mathcal{F}$, $|F_1 \cap F_2| \in L$ and $|F_1| \notin L$, then $|\mathcal{F}| \leq {n \choose 0} + \cdots + {n \choose s}$.

The bound that Theorem 1.3.8 gives for odd town (i.e. when p = 2 and s = 1) is almost tight, but Theorem 1.3.8 also handles many more cases. It would be interesting to extend the supersaturation result in Theorems 1.3.4 and 1.3.6 to the more general p and L setting:

Problem 1.3.9. Let p be prime and let $L \subseteq [0, p-1]$ with |L| = s. What is the minimum number of pairwise intersections not in $L \pmod{p}$ over all set families $\mathcal{F} \subseteq 2^{[n]}$ with $|\mathcal{F}| \ge {n \choose 0} + \dots + {n \choose s} + 1$? What is it amongst families with $|\mathcal{F}| \ge {n \choose 0} + \dots + {n \choose s} + t$ for a positive integer t > 1?

There are also results on families $\mathcal{F} \subseteq {\binom{[n]}{k}}$ which, modulo a prime p, have restricted pairwise intersections. In this area, the monumental result is the Frankl-Wilson Theorem [FW81] which states:

Theorem 1.3.10 (Frankl-Wilson [FW81]). Let p be prime and let $L \subseteq [0, p-1]$ with $k \notin L \pmod{p}$ and |L| = s. If $\mathcal{F} \subseteq {\binom{[n]}{k}}$ is such that for all $F_1 \neq F_2 \in \mathcal{F}$, $|F_1 \cap F_2| \in L \pmod{p}$, then $|\mathcal{F}| \leq {\binom{n}{s}}$.

Analogous to Problem 1.3.9 asking for a supersaturation version of Theorem 1.3.8, it would also be interesting to explore supersaturation versions of the well studied Frankl-Wilson Theorem:

Problem 1.3.11. Let p be prime and let $L \subseteq [0, p-1]$ with $k \notin L \pmod{p}$ and |L| = s. What is the minimum number of pairwise intersections not in $L \pmod{p}$ over all set families $\mathcal{F} \subseteq {\binom{[n]}{k}}$ with $|\mathcal{F}| \ge {\binom{n}{s}} + 1$? What is it amongst families with $|\mathcal{F}| \ge {\binom{n}{s}} + t$ for a positive integer t > 1?

1.4 Cross family intersections

In this section, we explore a natural extension to problems in extremal set theory with intersection constraints on the pairwise intersections from a set family. The sets families $\mathcal{A} \subseteq {\binom{[n]}{k}}$ and $\mathcal{B} \subseteq {\binom{[n]}{k}}$ are said to be *cross-intersecting* if $A \cap B \neq \emptyset$ for all $A \in \mathcal{A}$ and $B \in \mathcal{B}$. There is a rich history of research on cross-intersecting set families initiated from Hilton and Milner [HM67] who proved the following:

Theorem 1.4.1 (Hilton-Milner [HM67]). Let $n \ge 2k$. If $\mathcal{A} \subseteq \binom{[n]}{k}$ and $\mathcal{B} \subseteq \binom{[n]}{k}$ are cross intersecting, then

$$|\mathcal{A}| + |\mathcal{B}| \le {\binom{n}{k}} - {\binom{n-k}{k}} + 1.$$

It is straightforward to see that Theorem 1.4.1 is best possible by taking $\mathcal{A} = \{[k]\}$ and $\mathcal{B} = \{A \in {[n] \choose k} : A \cap [k] \neq \emptyset\}$. In addition to being a cross family variant of the Erdős-Ko-Rado theorem, Theorem 1.4.1 is a key lemma in the proof of Theorem 1.2.2. Pyber [Pyb86] proved another classical result on the maximum such product:

Theorem 1.4.2 (Pyber [Pyb86]). Let $n \ge 2k$. If $\mathcal{A} \subseteq \binom{[n]}{k}$ and $\mathcal{B} \subseteq \binom{[n]}{k}$ are cross-intersecting, then

$$|\mathcal{A}||\mathcal{B}| \le \binom{n-1}{k-1}^2$$

In Section 1.3, we explored the classical oddtown problem of set families consisting of sets of odd size for each distinct pairs of sets have an even number of elements in common. The generalization of this classical result to the cross family setting is the *bipartite oddtown theorem*, which appears in Babai and Frankl [BF20, Exercise 1.1.8]: **Theorem 1.4.3** (Babai, Frankl [BF20]). Let $\mathcal{A} = \{A_1, A_2, \dots, A_m\} \subset 2^{[n]}$ and $\mathcal{B} = \{B_1, B_2, \dots, B_m\} \subset 2^{[n]}$ be families where $|A_i \cap B_i|$ is odd for all $i \in [m]$ and $|A_i \cap B_j|$ is even for all $i \neq j \in [m]$. Then $m \leq n$.

Proof. Recall that v_{A_i} and v_{B_j} are the characteristic vectors of A_i and B_j respectively. Set A to be the $m \times n$ matrix whose m rows consist of the vectors $v_{A_1}, \ldots, v_{A_m} \in \mathbb{F}_2^n$. Similarly, set B to be the $n \times m$ matrix whose m columns consist of the vectors $v_{B_1}, \ldots, v_{B_m} \in \mathbb{F}_2^n$. The bipartite oddtown conditions implies that $AB = I_m$ where I_m is the $m \times m$ identity matrix. Noting the standard linear algebra fact that $\operatorname{rank}(AB) \leq \min\{\operatorname{rank}(A), \operatorname{rank}(B)\}$, we recover the desired bound that $m \leq n$.

To see that Theorem 1.4.3 is a generalization of the Oddtown theorem, let $\mathcal{A} = \{A_1, \ldots, A_m\} \subseteq 2^{[n]}$ satisfy oddtown rules. We now define $\mathcal{B} = \{B_1, \ldots, B_m\} \subseteq 2^{[n]}$ so that $B_i = A_i$ for all $i \in [m]$. It then follows that \mathcal{A} and \mathcal{B} satisfy the condition of Theorem 1.4.3 and hence the conclusion also implies the Oddtown result. Although the proof of Theorem 1.4.3 is different than the proof of Oddtown theorem, it still falls under the general *linear algebra method*. It is also worth noting that translating the proof of oddtown with slight alterations gives an upper bound of n + 1 for the bipartite oddtown problem, where n+1 comes from dimension of the vector space of multilinear polynomials of degree at most one (see also the proof of Theorem 1.3.8 [ABS91, DFS83]).

In Section 1.2, we explored an extension of a classical result on pairwise intersections to the *d*-wise intersecting setting. In this section, we similarly explore an extension of Theorem 1.4.3 to $k \ge 3$ families. We use the letter k to denote the number of set families as opposed to d as our methods involve reducing the problem to a k-uniform hypergraph problem for which the notation k is more standard. Our main result is as follows:

Theorem 1.4.4. Let $(\mathcal{A}_1, \mathcal{A}_2, \dots, \mathcal{A}_k)$ be set families of an n element set with $\mathcal{A}_j = \{A_{j,i} : 1 \leq i \leq m\}$ where $|\bigcap_{j=1}^k A_{j,i_j}|$ is even if and only if $i_1, \dots, i_k \in [m]$ are all distinct. Then $m = O(n^{1/\lfloor k/2 \rfloor})$.

By employing a connection to a hypergraph covering problem and a hypergraph extension (see Alon [Alo86], Cioabă-Kündgen-Verstraëte [CKV09], and Leader-Milićević-Tan [LMT18]) of the Graham-Pollak Theorem [GP72], we also show that Theorem 1.4.4 is best possible. Moreover, we prove the following:

Theorem 1.4.5. Let t, k be integers with $t \ge 2$ and $2t - 2 \le k$. If $(\mathcal{A}_1, \mathcal{A}_2, \dots, \mathcal{A}_k)$ are set families of an n element set with $\mathcal{A}_j = \{A_{j,i} : 1 \le i \le m\}$ where $|\bigcap_{j=1}^k A_{j,i_j}|$ is even if and only if at least t of the i_j are distinct, then $m = O(n^{1/(t-1)})$.

Theorem 1.4.5 is seen to be best possible by again using a connection to a hypergraph covering problem. In the case where $\mathcal{A} = \mathcal{A}_1 = \cdots = \mathcal{A}_k$, we recover a variation of the classical oddtown problem. For integers t < k, a family $\mathcal{A} \subseteq 2^{[n]}$ is said to follow (k,t)-oddtown rules if the intersection of any d distinct sets in \mathcal{A} is odd if d < t and even if $t \leq d \leq k$.

Corollary 1.4.6. Let $t, k \in \mathbb{N}$ be so that $2 \leq t \leq k$ and $2t - 2 \leq k$. If $\mathcal{A} \subseteq 2^{[n]}$ follows (k, t)-oddtown rules, then $|\mathcal{A}| = O(n^{1/(t-1)})$.

Corollary 1.4.6 is best possible by considering $\mathcal{A} = \{A_i\}_{i \in [n]}$ where $A_i = \{F \in \binom{[n]}{t-1} : i \in F\}$ and where n is so that the binomial coefficients corresponding to the size of the d-wise intersections are odd for d < t. When t < k and 2t - 2 > k, we are able to show that $m = O(n^{1/(k-t+1)})$ (see Section 4.3), and conjecture that a stronger bound holds:

Conjecture 1.4.7. Let t, k be integers with $t \ge 2$ and 2t - 2 > k. If $(\mathcal{A}_1, \mathcal{A}_2, \dots, \mathcal{A}_k)$ are set families of an n element set with $\mathcal{A}_j = \{A_{j,i} : 1 \le i \le m\}$ where $|\bigcap_{j=1}^k A_{j,i_j}|$ is even if and only if at least t of the i_j are distinct, then $m = O(n^{1/\lfloor k/2 \rfloor})$.

The bound $m = O(n^{1/\lfloor k/2 \rfloor})$ comes from a connection to a hypergraph covering problem (see Section 4.1) and would imply that Conjecture 1.4.7 if true is tight. By adding an element to *every* set in a family, the size of *d*-wise intersections switch parity and we get "reverse" analogs of Theorems 1.4.4 and 1.4.5.

Many results in extremal set theory regarding even/odd sized intersections have been extended to arbitrary primes p. Theorem 1.3.8 and Theorem 1.3.10 are canonical examples of this in the non-uniform and uniform cases respectively. For $k \ge 3$, Szabó and Vu [SV05] consider k-wise oddtown problems wherein the size of the sets are odd and the sizes of intersections of k distinct sets are even. Grolmusz and Sudakov [FS04] study k-wise (p, L)-intersecting set systems where the size of distinct k-wise intersections are in $L \pmod{p}$. Similarly, we propose the following extension to Theorem 1.4.4 for primes p > 2. **Conjecture 1.4.8.** Let p be prime and $(\mathcal{A}_1, \mathcal{A}_2, \ldots, \mathcal{A}_k)$ be set families of an n element set with $\mathcal{A}_j = \{A_{j,i} : 1 \leq i \leq m\}$ where $|\bigcap_{j=1}^k A_{j,i_j}|$ is nonzero modulo p if and only if $i_1, \ldots, i_k \in [m]$ are all distinct. Then $m = O(n^{(p-1)/\lfloor k/2 \rfloor})$.

Füredi and Sudakov [FS04, Lemma 3.1] used multilinear polynomials to prove if $\mathcal{A} = \{A_1, \ldots, A_m\}$ and $\mathcal{B} = \{B_1, \ldots, B_m\}$ are families of subsets of an *n* element set where $|A_i \cap B_j|$ is nonzero modulo *p* if and only if $i \neq j$, then

$$m \le \sum_{i=0}^{p-1} \binom{n}{i}.$$
(1.2)

This verifies Conjecture 1.4.8 when k = 2. It then follows that (1.2) is asymptotically best possible by taking $\mathcal{A} = {[n] \choose p-1}$ and \mathcal{B} to be the corresponding complements.

1.5 The Bollobás set pairs inequality

In Section 1.4, we considered cross-intersecting families and proved bounds on the number of sets from each family provided the set families satisfied underlying intersection properties. Although many of the classical results in the field fit under this guise, there are also instances of classical results in which we can prove much more. One particular instance of this is the Bollobás set pairs inequality or two families theorem [Bol65]:

Theorem 1.5.1. (Bollobás) Let $\mathcal{A} = \{A_1, A_2, \dots, A_m\}$ and $\mathcal{B} = \{B_1, B_2, \dots, B_m\}$ be families of finite sets, such that $A_i \cap B_j \neq \emptyset$ if and only if $i, j \in [m]$ are distinct. Then

$$\sum_{i=1}^{m} \binom{|A_i \cup B_i|}{|A_i|}^{-1} \le 1.$$
(1.3)

Proof. Set $X = (A_1 \cup B_1) \cup \cdots \cup (A_m \cup B_m)$ to be the ground set and suppose |X| = n. For each $i \in [m]$, define

$$\mathcal{C}_i = \{\pi : X \to [n] : \max_{x \in A_i} \pi(x) < \min_{x \in B_i} \pi(x)\}$$

to be the collections of permutations which order all of A_i before B_i . It then follows that $C_i \cap C_j = \emptyset$. To see this, suppose that $\pi \in C_i \cap C_j$ and without loss of generality suppose that π is so that

$$\max_{x \in A_i} \pi(x) \le \max_{x \in A_j} \pi(x). \tag{1.4}$$

As $A_i \cap B_j \neq \emptyset$, there exists an element $y \in A_i \cap B_j$. As a result, by definition of $\pi \in \mathcal{C}_i \cap \mathcal{C}_j$ and (1.4),

$$\pi(y) \le \max_{x \in A_i} \pi(x) \le \max_{x \in A_j} \pi(x) < \min_{x \in B_j} \pi(x) \le \pi(y).$$

It therefore follows that $\pi(y) < \pi(y)$; a contradiction. A straightforward counting argument then gives

$$|\mathcal{C}_i| = \binom{n}{|A_i \cup B_i|} |A_i|! |B_i|! (n - |A_i \cup B_i|)! = n! \cdot \binom{|A_i \cup |B_i|}{|A_i|}^{-1}.$$

Thus, noting that the C_i are pairwise disjoint, we recover

$$n! \ge \sum_{i=1}^{m} |\mathcal{C}_i| = \sum_{i=1}^{m} n! \cdot {\binom{|A_i \cup |B_i|}{|A_i|}}^{-1}$$

and the result follows by dividing through by n!.

For convenience, we refer to a pair of families \mathcal{A} and \mathcal{B} satisfying the conditions of Theorem 1.5.1 as a *Bollobás set pair*. Bollobás set pairs are also interesting for their corresponding one family problem. A set system $\mathcal{A} \subseteq 2^{[n]}$ is called an *antichain* if for all $A, B \in \mathcal{A}, A \nsubseteq B$ and $B \nsubseteq A$ (i.e. no two sets in the family have one contained in the other). Given an antichain $\mathcal{A} \subseteq 2^{[n]}$, it follows that $(\mathcal{A}, \mathcal{A}^c)$ is a Bollobás set pair. An immediate corollary of Theorem 1.5.1 is that the number of sets in a Bollobás set pair is at most $\binom{n}{\lfloor n/2 \rfloor}$. Further, this also implies the classical result of Sperner [Spe28] on the size of the largest antichain. As such, Theorem 1.5.1 is substantially stronger than a the typical extremal set theory result on the maximum number of subsets set families can have if they satisfy an intersections constraint.

The inequality (1.3) is tight, as we may take the pairs (A_i, B_i) to be distinct partitions of a set of size a + b with $|A_i| = a$ and $|B_i| = b$ for $1 \le i \le {a+b \choose a}$. The latter inequality was proved for a = 2 by Erdős, Hajnal and Moon [EHM64], and in general has a number of different proofs [Han64, JP71, Kat74, Lov77a, Lov77b]. A geometric version of Theorem 1.5.1 was proved by Lovász [Lov77a, Lov77b], who showed that if A_1, A_2, \ldots, A_m and B_1, B_2, \ldots, B_m are respectively *a*-dimensional and *b*-dimensional subspaces of a linear space and dim $(A_i \cap B_j) = 0$ if and only if $i, j \in [m]$ are distinct, then $m \le {a+b \choose a}$. Theorem 1.5.1 has been generalized in a number of different directions in the literature [Fra82, F84, KKK15, Lov07, Tal04, Tuz85].

In this dissertation, we give a generalization of Theorem 1.5.1 from the case of two families to $k \ge 3$ families of sets with conditions on the k-wise intersections. For $2 \le t \le k$, a Bollobás (k, t)-tuple is a sequence $(\mathcal{A}_1, \mathcal{A}_2, \ldots, \mathcal{A}_k)$ of set families $\mathcal{A}_j =$ $\{A_{j,i} : 1 \le i \le m\}$ where $\bigcap_{j=1}^k A_{j,i_j} \ne \emptyset$ if and only if at least t of the indices i_1, i_2, \ldots, i_k are distinct. We refer to m as the *size* of the Bollobás (k, t)-tuple. Let $[m]_{(t)}$ denote the set of sequences of t distinct elements of [m] and fix a surjection $\phi : [k] \to [t]$. For $\sigma \in [m]_{(t-1)}$, set $\sigma(t) = \sigma(1)$ and define $A_{1,\sigma}(\phi) = \bigcap_{j:\phi(j)=1} A_{j,\sigma(1)}$ and, for $2 \leq j \leq t$, we define

$$A_{j,\sigma}(\phi) = \bigcap_{h:\phi(h)=j} A_{h,\sigma(j)} \setminus \bigcup_{h=1}^{j-1} A_{h,\sigma}(\phi).$$

Using this notation, we generalize (1.3) as follows:

Theorem 1.5.2. Let $k \ge t \ge 2$ and $m \ge t$, let $\phi : [k] \to [t]$ be a surjection, and let $(\mathcal{A}_1, \mathcal{A}_2, \ldots, \mathcal{A}_k)$ be a Bollobás (k, t)-tuple of size m. Then

$$\sum_{\sigma \in [m]_{(t-1)}} \left(\begin{vmatrix} A_{1,\sigma}(\phi) \cup A_{2,\sigma}(\phi) \cup \cdots \cup A_{t,\sigma}(\phi) \\ |A_{1,\sigma}(\phi)| & |A_{2,\sigma}(\phi)| & \cdots & |A_{t,\sigma}(\phi)| \end{vmatrix} \right)^{-1} \le 1.$$
(1.5)

We show in Chapter 5 that this inequality is tight for all $k \ge t = 2$, but do not have an example to show that this inequality is tight for any t > 2.

For $n \ge k \ge t \ge 2$, let $\beta_{k,t}(n)$ denote the maximum m such that there exists a Bollobás (k, t)-tuple of size m consisting of subsets of [n]. Then (1.3) gives $\beta_{2,2}(n) \le {\binom{n}{\lfloor n/2 \rfloor}}$ which is tight for all $n \ge 2$. Letting $H(q) = -q \log_2 q - (1-q) \log_2(1-q)$ denote the standard binary entropy function, we prove the following theorem:

Theorem 1.5.3. For $k \geq 3$ and large enough n,

$$\frac{1}{k} \le \frac{\log_2 \beta_{k,2}(n)}{n} \le H\left(\frac{1}{k}\right) \le \frac{\log_2(ke)}{k}.$$
(1.6)

For $k \geq t \geq 3$ and large enough n,

$$\frac{\log_2 e}{\binom{k}{t-1}(t+1)t^{t-1}} \le \frac{\log_2 \beta_{k,t}(n)}{n} \le \frac{2}{\binom{k}{t-1}(t-1)^{t-3}}.$$
(1.7)

This determines $\log_2 \beta_{k,2}(n)$ up to a factor of order $\log_2 k$ and $\log_2 \beta_{k,t}(n)$ up to a factor of order t^3 . We leave it as an open problem to determine the asymptotic value of $(\log_2 \beta_{k,t}(n))/n$ as $n \to \infty$ for any $k \ge 3$ and $t \ge 2$. A natural source for lower bounds on $\beta_{k,t}(n)$ comes from the probabilistic method – see the random constructions in Section 5.2.1 which establish the lower bounds in Theorem 1.5.3. To prove Theorem 1.5.3, we use a natural connection to hypergraph covering problems.

Theorem 1.5.1 has a wide variety of applications, from saturation problems [Bol65, MS15] to covering problems for graphs [Han64, Orl77], complexity of 0-1 matrices [Tar75], geometric problems [AK85], counting cross-intersecting families [FK18], crosscuts and transversals of hypergraphs [Tuz85, Tuz94, Tuz96], hypergraph entropy [KM88, Sim95], and perfect hashing [FK84, GR19]. In this section, we give an application of our main results to hypergraph covering problems. For a k-uniform hypergraph H, let f(H) denote the minimum number of complete k-partite k-uniform hypergraphs whose union is H. In the case of graph covering, a simple connection to the Bollobás set pairs inequality (1.3) may be described as follows. Let $K_{n,n} \setminus M$ denote the complement of a perfect matching $M = \{x_iy_i : 1 \leq i \leq n\}$ in the complete bipartite graph $K_{n,n}$ with parts $X = \{x_1, x_2, \ldots, x_n\}$ and $Y = \{y_1, y_2, \ldots, y_n\}$. If H_1, H_2, \ldots, H_m are complete bipartite graphs in a minimum covering of $K_{n,n} \setminus M$, then let $A_i = \{j : x_i \in V(H_j)\}$ and $B_i = \{j :$ $y_i \in V(H_j)\}$. Setting $\mathcal{A} = \{A_i\}_{i \in [m]}$ and $\mathcal{B} = \{B_i\}_{i \in [m]}$, it is straightforward to check that $(\mathcal{A}, \mathcal{B})$ is a Bollobás set pair, and Theorem 1.5.1 applies to give

$$f(K_{n,n} \setminus M) = \min\{m : \binom{m}{\lceil m/2 \rceil} \ge n\}.$$
(1.8)

In a similar way, Theorem 1.5.2 applies to covering complete k-partite k-uniform hypergraphs. Let $K_{n,n,\ldots,n}$ denote the complete k-partite k-uniform hypergraph with parts $X_i = \{x_{ij} : j \in [n]\}$ for $i \in [k]$. Let $H_{k,t}(n)$ denote the subhypergraph consisting of hyperedges $\{x_{1,i_1}, x_{2,i_2}, \ldots, x_{k,i_k}\}$ such that at least t of the indices i_1, i_2, \ldots, i_k are distinct, and set $f_{k,t}(n) = f(H_{k,t}(n))$. Then there is a one-to-one correspondence between Bollobás (k, t)-tuples of subsets of [m] and coverings of $H_{k,t}(n)$ with m complete k-partite k-graphs. We let $\beta_{k,t}(m)$ be the maximum size of a Bollobás (k, t)-tuple of subsets of [m], so that

$$f_{k,t}(n) = \min\{m : \beta_{k,t}(m) \ge n\}.$$
(1.9)

This correspondence together with Theorem 1.5.2 will be exploited to prove

$$f_{k,2}(n) \ge \min\{m : \binom{m}{\lceil m/k \rceil} \ge n\}$$
(1.10)

which is partly an analog of (1.8). More generally, we prove the following theorem:

Theorem 1.5.4. For $k \geq 3$ and large enough n,

$$\frac{k}{\log_2(ke)} \le \frac{1}{H(\frac{1}{k})} \le \frac{f_{k,2}(n)}{\log_2 n} \le k.$$
(1.11)

For $k \geq t \geq 3$ and large enough n,

$$\binom{k}{t-1}\frac{(t-1)^{t-3}}{2} \le \frac{f_{k,t}(n)}{\log_2 n} \le \frac{(t+1)t^{t-1}}{\log_2 e} \binom{k}{t-1}.$$
(1.12)

The bounds on $\beta_{k,t}(n)$ in Theorem (1.5.3) follow immediately from this theorem and (1.9). Equation (1.12) gives the order of magnitude for each $t \ge 3$ as $k \to \infty$, but for t = 2, Equation (1.11) has a gap of order $\log_2 k$. From (1.10), we obtain $\beta_{k,2}(n) \le {n \choose \lfloor n/k \rfloor}$. It is perhaps unsurprising that the asymptotic value of $f_{k,t}(n)/\log_2 n$ as $n \to \infty$ is not known for any k > 2, since a limiting value of $f(K_n^k)/\log_2 n$ is not known for any k > 2– see Körner and Marston [KM88] and Guruswami and Riazanov [GR19].

Although there has been substantial research on Bollobás set pairs and various generalization, there are still many possible directions for future research. For example, it would be interesting to explore the problem where we allow for more permissible values for $|A_i \cap B_i|$; i.e. a two family version of Theorem 1.3.8:

Problem 1.5.5. Let p be prime and $L \subset [0, p - 1]$. A set pair $(\mathcal{A}, \mathcal{B})$ is (p, L)-cross intersecting if $\mathcal{A} = \{A_1, A_2, \ldots, A_m\} \subset 2^{[n]}$ and $\mathcal{B} = \{B_1, B_2, \ldots, B_m\} \subset 2^{[n]}$ are so that $A_i \cap B_j \in L \pmod{p}$ if and only if $i, j \in [m]$ are distinct. What is the largest (p, L)-cross intersecting set pair?

In the next section, we explore one of Bollobás' original applications of Theorem 1.5.1.

1.6 Intersection Saturation

In Section 1.3, we explored supersaturation problems where there are many instances of a forbidden intersection provided we have more than the extremal amount. In this section, we explore a related concept known as saturation: what is the minimum number of sets in a set system which avoids an intersection condition, but adding any edge necessarily creates a forbidden intersection. More formally, a set family $\mathcal{F} \subset {[n] \choose k}$ is (n, k, t)-saturated if $|A \cap B| \neq t$ for all $A, B \in \mathcal{F}$ and for all $C \in {\binom{[n]}{k}}$ with $C \notin \mathcal{F}$, there exists $A \in \mathcal{F}$ so that $|A \cap C| = t$.

The motivation for intersection saturation is from the classical theory of saturation numbers for graphs. Given a forbidden graph F, an *n*-vertex graph $G \subset {\binom{[n]}{2}}$ is Fsaturated if G does not contain any copies of F and any edge $e \in {\binom{[n]}{2}} \setminus G$ is so that $G \cup e$ contains a copy of F. The saturation number of F, denoted sat(n, F) is the minimum number of edges in an *n*-vertex F-saturated graph. In this language, the underlying object of study in this section is the hypergraph analog of the saturation number when F is the hypergraph consisting of two edges with t vertices in common.

While saturation problems for graphs have been extensively studied, there is not much known regarding hypergraph saturation. Perhaps somewhat surprisingly, we understand hypergraph saturation for cliques $\binom{[r]}{k}$ quite well:

Theorem 1.6.1 (Bollobás [Bol65]). Fix $n \ge r \ge k$. Then

$$sat\left(n, \binom{[r]}{k}\right) = \binom{n}{k} - \binom{n-r+k}{k}.$$

Proof. First, we consider the set system

$$\mathcal{F} = \{A \in \binom{[n]}{k} : A \cap [r-k] \neq \emptyset\}.$$

It follows that \mathcal{F} does not contain a copy of $\binom{[r]}{k}$ as such a copy would necessarily contain at least k elements disjoint from [r-k] and hence at least one set disjoint from [r-k]. To see that \mathcal{F} is saturated, let $X \in \mathcal{F}^c$ so that $X \cap [r-k] = \emptyset$. Then $\mathcal{F} \cup \{X\}$ contains a copy of $\binom{[r]}{k}$ on the vertex set $X \cup [r-k]$. This shows the upper bound sat $(n, \binom{[r]}{k}) \leq \binom{n}{k} - \binom{n-r+k}{k}$. For the lower bound, let $\mathcal{H} \subset {\binom{[n]}{k}}$ be saturated with respect to ${\binom{[r]}{k}}$. Then let $\mathcal{A} = {\binom{[n]}{k}} \setminus \mathcal{H}$ and set $\mathcal{A} = \{A_1, \ldots, A_l\}$. For each A_i , it follows that $\mathcal{H} \cup \{A_i\}$ contains a copy of ${\binom{[r]}{k}}$. Let V_i denote the vertex set of this copy and set $B_i = [n] \setminus V_i$. It then clearly follows that $A_i \cap B_i = \emptyset$ for all $i \in [l]$. Noting that each V_i contains exactly one k-set from \mathcal{A} (namely $A_i \subset V_i$), it follows that $A_j \cap B_i \neq \emptyset$ for all $j \neq i \in [l]$. As such, $(\mathcal{A}, \mathcal{B})$ is a Bollobás set pair with each set in \mathcal{A} having size k and each set in \mathcal{B} having size n - r. Using Theorem 1.5.1, we recover

$$|\mathcal{A}| = l \le \binom{(n-r)+k}{k}$$

Noting that $\mathcal{A} = \binom{[n]}{k} \setminus \mathcal{H}$, we obtain the desired lower bound that

$$|\mathcal{H}| \ge \binom{n}{k} - \binom{n-r+k}{k}.$$

It turns out that a good candidate from minimally (n, k, t)-saturated set systems are Steiner systems. For positive integers t < k < n, a *Steiner system* $S = S(n, k, t) \subset {\binom{[n]}{k}}$ is a set system where for all $T \in {\binom{[n]}{t}}$, there exists a unique $F \in S$ so that $T \subset F$. The existence of Steiner systems, and more broadly designs, is a rich combinatorial area which has been extensively studied [Wil72, Kee18, GKLO20, JX83, Tie91] in the literature.

Given a Steiner system S(n, k, t), it is straightforward to see that given any distinct $S_1, S_2 \in S(n, k, t), |S_1 \cap S_2| \neq t$ as otherwise there would a *t*-set contained in *both* S_1 and S_2 . Moreover, when we add a set $C \in {\binom{[n]}{k}}$ but $C \notin S(n, k, t)$, it follows that we create an intersection of size *at least t*. Moreover, we actually create ${\binom{k}{t}}$ sets of size *t* which are contained two sets in our new system. It therefore turns out that for many integer pairs

(k, t), we necessarily have that all Steiner systems $\mathcal{S}(n, k, t)$ are (n, k, t)-saturated and as such we define

$$w(n,k,t) := \min\left\{\frac{|\mathcal{F}| \cdot \binom{k}{t}}{\binom{n}{t}} : \mathcal{F} \subset \binom{[n]}{k} \quad \text{is an } (n,k,t)\text{-saturated set system}\right\}.$$
(1.13)

Up to re-scaling, w(n, k, t) is the saturation number of the k-uniform hypergraph consisting of two edges intersecting in exactly t places. For more on saturation numbers of hypergraphs, see the survey of Faudree, Faudree, and Schmitt [FFS11].

A good lower bound on w(n, k, t) comes from a hypergraph Turán problem. Let $K_k^{(t)}$ denote the *t*-uniform clique on *k* vertices and $ex_t(n, K_k^{(t)})$ be the maximum number of edges of an $K_k^{(t)}$ -free *t*-uniform hypergraph on *n* vertices. An averaging argument gives that $\pi(K_k^{(t)}) := \lim_{n\to\infty} ex_t(n, K_k^{(t)})/{n \choose t}$ exists for all k > t > 2. Hypergraph Turán problems are often studied via the dual problem of determining the minimum size $\mathcal{F} \subset {n \choose t}$ so that every $A \in {n \choose k}$ contains an edge of \mathcal{F} . In this setting, the limiting density is denoted by t(k, t) so that $t(k, t) + \pi(K_k^{(t)}) = 1$. Although the limiting value is not known for any k > t > 2, Turán [Tur41] conjectured the following for t = 3:

Conjecture 1.6.2 (Turán [Tur41]). Let $k \ge 3$. Then $\pi(K_k^{(3)}) = 1 - 4/(k-1)^2$.

Towards a lower bound on w(n, k, t), let $\mathcal{F} \subset {\binom{[n]}{k}}$ be an (n, k, t)-saturated set system and $\partial_t(\mathcal{F}) = \{T \in {\binom{[n]}{t}} : \exists F \in \mathcal{F}, T \subset F\}$. Then ${\binom{[n]}{t}} \setminus \partial_t(\mathcal{F})$ is $K_k^{(t)}$ -free as otherwise we may add that k-set to \mathcal{F} without creating an intersection of size exactly t; contradicting the fact that \mathcal{F} is (n, k, t)-saturated. Thus, $|\binom{[n]}{t} \setminus \partial_t(\mathcal{F})| \leq \pi(K_k^{(t)})\binom{n}{t}$. Noting that $|\partial_t(\mathcal{F})| \leq |\mathcal{F}| \cdot {k \choose t}$, and dividing through by ${n \choose t}$, we recover:

$$w(n,k,t) \ge 1 - \pi(K_k^{(t)}) = t(k,t).$$
(1.14)

There has been substantial research towards hypergraph Turán numbers for cliques $K_k^{(t)}$. However, in the case where t > 3, there are not even conjectured limiting values for $\pi(K_k^{(t)})$ for many integers pairs (k, t) to match that of Conjecture 1.6.2. This is perhaps somewhat surprising as Theorem 1.6.1 determines the corresponding saturation number exactly. For more on the vast history of hypergraph Turán problems, see the surveys written by Keevash [Kee11] and Sidorenko [Sid95].

Using a result of Füredi [F15] on the stability of extremal K_k -free graphs, we are able to prove that the bound in (1.14) is not tight when k > t and t = 2.

Theorem 1.6.3. Let n be sufficiently large, then $w(n, k, 2) \ge (1 + k^{-2}) \cdot t(k, 2)$.

A Steiner system S(n, k, t) is (n, k, t)-saturated (and hence $w(n, k, t) \leq 1$) if each clique partition of $\binom{[k]}{t}$ into smaller cliques contains at least one *t*-set. This need not be true in general as the Fano plane exhibits a clique partition of $\binom{[7]}{2}$ into cliques of size three. As a result, we construct a (n, k, t)-saturated set system with $\Theta(n^t)$ edges:

Theorem 1.6.4. For all $1 \le t < k$ and $q \in \mathbb{R}$ such that $k = q^2 + q + t - 1$,

$$w(n,k,t) \le \binom{k + \lfloor q - 1 \rfloor}{\lfloor q - 1 \rfloor}.$$
(1.15)

The bound in (1.15) is unlikely to be tight and perhaps there even exists Steiner systems which are $o(n^t)$ sets away from being (n, k, t)-saturated. Moreover, there are also cases when w(n, k, t) < 1. Pikhurko [Pik01] proved that $w(n, 4, 3) \le 4/9$ which is best possible if Conjecture 1.6.2 is true. We are able to extend the work of Pikhurko [Pik01] when (k, t) = (5, 4):

Theorem 1.6.5. For n sufficiently large, $w(n, 5, 4) \leq 5/8$.

Using the Rödl Nibble, we obtain a construction of a (n, k, k - 2)-saturated set system for $k \in \{6, 7\}$:

Theorem 1.6.6. Let *n* be sufficiently large. Then $w(n, 7, 5) \le 1/2$ and $w(n, 6, 4) \le 3/4$.

A similar construction as in Theorem 1.6.6 gives $w(n, 11, 8) \le 67/128$ and it seems likely that this can be extended to other integer pairs (k, t). However, all integer pairs for which we were able to show w(n, k, t) < 1 are so that 2t > k and it would interesting to find a construction which works for $t \le 2k$:

Problem 1.6.7. Find an integer pair (k, t) with $2t \leq r$ for which w(n, k, t) < 1.

It would also be interesting to study intersection saturation in the more general setting where a set family has a list of permissible intersection sizes. To this end, let $L \subset [0, k - 1]$. Then $\mathcal{F} \subset {\binom{[n]}{k}}$ is an (n, k, L)-saturated set system if for each $A, B \in \mathcal{F}$, $|A \cap B| \in L$ and for each $C \in \mathcal{F}^c$, there exists $A \in \mathcal{F}$ with $|A \cap C| \notin L$. Deza, Erdős, and Frankl [DEF78] studied the problem of extremal, i.e. largest, (n, k, L)-saturated set systems and a good survey of the problem may be found in [FT18]. Let s(n, k, L) denote the size of the smallest (n, r, L)-saturated system. In this paper, we considered the case of $L = [0, k - 1] \setminus \{t\}$ and showed that $s(n, k, [0, r - 1] \setminus \{t\}) = \Theta(n^t)$. Moreover, letting $2 \le t < s \le r - 1$ be integers and setting $L = [0, t - 1] \cup [s, k - 1]$, it is not too hard to show that $\mathcal{F} = \partial_k(\mathcal{S}(n, 2k - s, t))$ is an (n, k, L)-saturated set system which together with an argument similar to (1.14) gives $s(n, k, [0, r - 1] \setminus \{t\}) = \Theta(n^t)$.

When L = [0, t - 1], it is straightforward to see Steiner systems S(n, k, t), if they exist, are (n, k, L)-saturated set systems of maximal size. Moreover, the same argument as in (1.14) gives:

$$s(n,k,[0,t-1]) \cdot \binom{k}{t} \ge t(k,t) \cdot \binom{n}{t}$$
(1.16)

As Theorem 1.6.3 shows that (1.14) is not sharp for $k \ge 3$ and $L = [0, k - 1] \setminus \{2\}$, we similarly believe that (1.16) is not always sharp.

Problem 1.6.8. Find an integer pair (k, t) for which (1.16) is not sharp. That is, show there exists an $\epsilon = \epsilon(k, t) > 0$ where $s(n, k, [0, t - 1]) \cdot {k \choose t} \ge (1 + \epsilon) \cdot t(k, t) \cdot {n \choose t}$.

A good candidate for Problem 1.6.8 is (k, t) = (5, 4). Sidorenko [Sid81] conjectured that $\pi(K_5^{(4)}) = 11/16$ which if true would imply that $w(n, 5, 4) \ge 5/16$ while Theorem 1.6.5 gives that $w(n, 5, 4) \le 5/8$. Another good candidate for Problem 1.6.8 is (k, t) =(7, 4) where taking two vertex disjoint S(n/2, 7, 4) is a $(n, 7, \{0, 1, 2, 3\})$ -saturated set system which gives $s(n, 7, \{0, 1, 2, 3\}) \cdot {7 \choose 4} \le 1/8 \cdot {n \choose 4}$ whereas Sidorenko [Sid21] recently showed $t(7, 4) \le .0866$.

For a subset $L \subset [0, k-1]$, let m(L) be the smallest integer from [0, k-1] not in L. It is straightforward to see that $s(n, k, L) = \Omega(n^{m(L)})$ by a similar argument as in (1.14) when $0 \in L$ and this vacuously holds otherwise. We conjecture that this lower bound is best possible:

Conjecture 1.6.9. Let $k \ge 1$ and $L \subset [0, k - 1]$. Then $s(n, k, L) = \Theta(n^{m(L)})$.

In order to prove Conjecture 1.6.9, it suffices to, for each $k \ge 1$ and $L \subset [0, k - 1]$, give a construction of an (n, k, L)-saturated set system of size $\Theta(n^{m(L)})$. Although we were able to verify Conjecture 1.6.9 for all $k \le 4$ and when L = [0, t-1] and $L = [0, t-1] \cup [s, k]$, Conjecture 1.6.9 might be too optimistic as the order of magnitude of extremal (n, k, L)saturated set systems is wide open. See [DEF78, FT18, Fra13] for more on extremal (n, k, L)-saturated set systems.

Chapter 2

Non-trivial d-wise intersecting

families

There have been many recent directions [BD71, HK17, Kup19, FHHZ18, FS04, GS02] in the classical Hilton-Milner theory generalizing Theorem 1.2.2. In this chapter, we prove Theorem 1.2.4, which resolved the conjecture of Hilton and Milner [HM67]. Our proof actually gives a bit more and in order to state our theorem, we need the following additional non-trivial *d*-wise intersecting family $\mathcal{B}(k,d) = \mathcal{B}_1(k,d) \cup \mathcal{B}_2(k,d)$ where

$$\mathcal{B}_1(k,d) = \{B \in \binom{[n]}{k} : |B \cap [d-1]| = d-2, [d,k] \subseteq B\}$$
$$\mathcal{B}_2(k,d) = \{B \in \binom{[n]}{k} : [d-1] \subseteq B, B \cap [d,k] \neq \emptyset\}.$$

The role of this family is in the stability for non-trivial *d*-wise intersecting families of *k*element sets when 2d < k, in which case $|\mathcal{A}(k, d)| \leq |\mathcal{B}(k, d)| \leq |\mathcal{H}\mathcal{M}(k, d)|$ with equalities if and only if k = 3 and d = 2 and are all isomorphic when d = k.

Theorem 2.0.1. Let k, d be integers with $2 \le d < k$. Then there exists $n_0(k, d)$ such that for $n \ge n_0(k, d)$, if \mathcal{F} is a non-trivial, d-wise intersecting family of k-element subsets of [n], then

$$|\mathcal{F}| \le \max\{|\mathcal{HM}(k,d)|, |\mathcal{A}(k,d)|\}.$$

Furthermore, if $d \geq 3$ and $2d \geq k$ and $|\mathcal{F}| > \min\{|\mathcal{HM}(k,d)|, |\mathcal{A}(k,d)|\}$, then \mathcal{F} is isomorphic to a subfamily of $\mathcal{HM}(k,d)$ or $\mathcal{A}(k,d)$. If $d \geq 3$, 2d < k, and $|\mathcal{F}| > |\mathcal{B}(k,d)|$, then \mathcal{F} is isomorphic to a subfamily of $\mathcal{HM}(k,d)$.

It is worth noting that the sizes of these families are given by

$$\begin{aligned} |\mathcal{A}(k,d)| &= (d+1)\binom{n-d-1}{k-d} + \binom{n-d-1}{k-d-1} \sim (d+1)\binom{n}{k-d} \\ |\mathcal{H}\mathcal{M}(k,d)| &= \binom{n-d+1}{k-d+1} - \binom{n-k-1}{k-d+1} + d - 1 \sim (k-d+2)\binom{n}{k-d}. \end{aligned}$$

This is particularly interesting as for large values of n, $|\mathcal{HM}(k, d)|$ and $|\mathcal{A}(k, d)|$ are both substantially smaller than the trivial *d*-wise intersecting family of size $\binom{n-1}{k-1}$. We use the Delta system method to prove Theorem 2.0.1 in the case where $k \ge d \ge 3$. When d = 2, stability versions of Theorem 1.2.2 appear in Han and Kohayakawa [HK17] and Kostochka and Mubayi [KM17]. The Delta system method is a powerful tool in extremal combinatorics that initially appeared in Deza, Erdős and Frankl's [DEF78] study of (n, k, L)-systems. It has also been used by Frankl and Füredi [FF87] in Chvátal's problem of avoiding *d*-simplicies as well as by Füredi [Für14, FJS14] on the problem of embedding expansions of forests in *r*-graphs for $r \ge 4$. The application of the Delta system method here gives $n_0(k, d) = d + e(k^2 2^k)^{2^k}(k - d)$.

2.1 Preliminaries

In this section, we will prove some basic facts and structural results pertaining to non-trivial *d*-wise intersecting families. We will first show, as was initially done by Hilton and Milner in [HM67], that there cannot be a *k*-uniform non-trivial *d*-wise intersecting family for d > k.

Lemma 2.1.1. [HM67] Let $n \ge d > k$, then there does not exist a d-wise intersecting non-trivial $\mathcal{F} \subseteq {\binom{[n]}{k}}$.

Proof. Fix $A \in \mathcal{F}$. For each $a \in A$, there exists $X_a \in \mathcal{F}$ so that $a \notin X_a$ by the definition of non-trivial. Then $A \cap \bigcap_{a \in A} X_a = \emptyset$ which is a contradiction.

A similar argument as in Lemma 2.1.1 also gives an upper bound on the m-wise intersections from a non-trivial d-wise intersecting family.

Lemma 2.1.2. Let $m \ge 1$ be a positive integer. Given a non-trivial d-wise intersecting family $\mathcal{F} \subseteq {\binom{[n]}{k}}$ and $A_1, \ldots, A_m \in \mathcal{F}$,

$$|\bigcap_{i=1}^{m} A_i| \ge d - (m - 1).$$

Proof. Suppose not, and that there exists $A_1, \ldots, A_m \in \mathcal{F}$ with $A = \bigcap_{i=1}^m A_i$ and |A| < d - (m-1). For each $a \in A$, by the non-triviality of \mathcal{F} , we may find $X_a \in \mathcal{F}$ so that $a \notin X_a$ and hence violate the *d*-wise intersecting property of \mathcal{F} .

In the case where d = k, Hilton and Milner [HM67] noted that $\binom{[k+1]}{k}$ is a non-trivial k-wise intersecting family and proved this is the only such example up to isomorphism.

Proposition 2.1.3. [HM67] If $n \ge k+1$ and $\mathcal{F} \subseteq {\binom{[n]}{k}}$ is a k-wise intersecting non-trivial family, then \mathcal{F} is isomorphic to ${\binom{[k+1]}{k}} = \mathcal{A}(k,k)$.

Proof. Let $A \in \mathcal{F}$, then without loss of generality we may assume that A = [k]. Observe that there exists $A_1 \in \mathcal{F}$ so that $1 \notin A_1$ and by Lemma 2.1.2, $|A \cap A_1| = k - 1$. Without loss of generality, let $A_1 \setminus A = \{k + 1\}$. Then for each $i \in [2, k]$, there exists $A_i \in \mathcal{F}$ so that $i \notin A_i$. Next, Lemma 2.1.2 yields that $|A \cap A_i| = k - 1$ and $|A_1 \cap A_i| = k - 1$ and as a result $A_i \cap [k + 2, n] = \emptyset$. Putting these all together we get that $\binom{[k+1]}{k} \subseteq \mathcal{F}$ and then noting that no other k-sets may be added to $\binom{[k+1]}{k}$ while preserving the k-wise intersection property yields the desired result.

2.2 Structure of *d*-wise intersecting families

We start by defining some terminology regarding Delta systems that may be found in the survey written by Mubayi and Verstraete [MV16]. A *Delta system* is a hypergraph Δ such that for all distinct $X_1, X_2 \in \Delta, X_1 \cap X_2 = \bigcap_{E \in \Delta} E$. We let $\Delta_{k,s}$ be a k-uniform Delta system with s edges and define core $(\Delta) := \bigcap_{E \in \Delta} E$. Let $\mathcal{F} \subseteq {[n] \choose k}$ and $X \subseteq [n]$, then define the *core degree* of X in \mathcal{F} to be the size of the largest Delta system contained in \mathcal{F} which has core equal to X. That is, let

$$d_{\mathcal{F}}^{\star}(X) := \max\{s : \exists \Delta_{k,s} \subseteq \mathcal{F} \text{ so that } \operatorname{core}(\Delta_{k,s}) = X\}.$$

In this section, we will examine the collection of *d*-sets with large core degree with respect to a non-trivial *d*-wise intersecting family. We will show that this collection of *d*-sets is necessarily isomorphic to a subfamily of one of the corresponding collections of *d*sets in the extremal examples $\mathcal{HM}(k,d)$ and $\mathcal{A}(k,d)$. Moreover, given enough *d*-sets with large core degree, we show that $|\mathcal{F}|$ is less than or equal to max{ $|\mathcal{A}(k,d)|, |\mathcal{HM}(k,d)|$ }.

By Lemma 2.1.2, $|A \cap B| \ge d-1$ for all $A, B \in \mathcal{F}$ in a non-trivial *d*-wise intersecting family \mathcal{F} , and hence $d_{\mathcal{F}}^{\star}(X) \le 1$ whenever |X| < d-1. Moreover, since |X| < d-1 < k, it follows that X cannot be the core of a delta system $\Delta_{k,1}$ and thus $d_{\mathcal{F}}^{\star}(X) = 0$. We will now show that (d-1)-sets cannot have large core degree in non-trivial *d*-wise intersecting families.

Lemma 2.2.1. Let $k \ge d \ge 3$ and $\mathcal{F} \subseteq {\binom{[n]}{k}}$ be non-trivial d-wise intersecting and $X \in {\binom{[n]}{d-1}}$. Then $d_{\mathcal{F}}^{\star}(X) < k$.

Proof. Without loss of generality, suppose that X = [d-1] is so that $d_{\mathcal{F}}^{\star}(X) \geq k$. Thus, there exists $\Delta_{k,k} = \{F_1, \ldots, F_k\} \subseteq \mathcal{F}$ so that $\operatorname{core}(\Delta_{k,k}) = [d-1]$. Next, by the nontriviality of \mathcal{F} , for each $j \in [d-1]$, there exists $X_j \in \mathcal{F}$ so that $j \notin X_j$.

Now, when $d \ge 4$, since $F_1 \cap F_2 = [d-1]$ and \mathcal{F} is *d*-wise intersecting, for $3 \le m \le d-1$,

$$F_1 \cap F_2 \cap \left(\bigcap_{j \in [d-1] \setminus \{m\}} X_j\right) = \{m\}.$$

As a result, $|X_m \cap [d-1]| = d-2$ and hence $|X_m \cap (F_i \setminus [d-1])| \ge 1$ for all $i \in [k]$ by Lemma 2.1.2. This yields a contradiction as

$$|X_m| \ge |X_m \cap [d-1]| + \sum_{j=1}^k |X_m \cap (F_i \setminus [d-1])| > k.$$

When d = 3, the result follows similarly by considering the cases where $2 \in X_1$ and $2 \notin X_1$.

We are interested in *d*-sets which have large core degree since they intersect elements of our family \mathcal{F} in many places. To this end, we say $D \in \binom{[n]}{d}$ has large core degree if $d_{\mathcal{F}}^{\star}(D) \geq k$.

Lemma 2.2.2. Let $k \ge d \ge 3$ and $\mathcal{F} \subseteq {\binom{[n]}{k}}$ be non-trivial d-wise intersecting and $D \in {\binom{[n]}{d}}$ have large core degree. Then $|A \cap D| \ge d - 1$ for all $A \in \mathcal{F}$.

Proof. Seeking a contradiction, suppose there exists such a $D \subseteq [n]$ and $A \in \mathcal{F}$ so that $|A \cap D| < d - 1$. By definition of large core degree, there exists a Delta system $\Delta_{k,k} = \{F_1, \ldots, F_k\} \subseteq \mathcal{F}$ so that $\operatorname{core}(\Delta_{k,k}) = D$. By Lemma 2.1.2, $|A \cap (F_j \setminus D)| \ge (d-1) - |A \cap D|$ for all $j \in [k]$. This gives a contradiction as

$$|A| \ge |A \cap D| + \sum_{j=1}^{k} |A \cap (F_j \setminus D)| > k.$$

Given a family \mathcal{F} , we let $\mathcal{S}_d(\mathcal{F})$ be the possibly empty collection of *d*-sets with large core degree in \mathcal{F} . We first show that $\mathcal{S}_d(\mathcal{F})$ is a (d-1)-intersecting family; i.e. the intersection of any pair of sets in the family has size at least (d-1).

Lemma 2.2.3. Let $k \ge d \ge 3$ and $\mathcal{F} \subseteq {\binom{[n]}{k}}$ be a non-trivial d-wise intersecting family. Then $\mathcal{S}_d(\mathcal{F}) \subseteq {\binom{[n]}{d}}$ is a (d-1)-intersecting family.

Proof. Let $d \geq 3$. Suppose there exists $D_1, D_2 \in \mathcal{S}_d(\mathcal{F})$ so that $|D_1 \cap D_2| \leq d-2$. By definition, there exists $\Delta_{k,k}^1 = \{F_1, \ldots, F_k\} \subseteq \mathcal{F}$ and $\Delta_{k,k}^2 = \{G_1, \ldots, G_k\} \subseteq \mathcal{F}$ so that $\operatorname{core}(\Delta_{k,k}^i) = D_i$ for i = 1, 2. Note that there necessarily exists $F_i \in \Delta_{k,k}^1$ so that $|(F_i \setminus D_1) \cap D_2| = 0$. By Lemma 2.1.2, $|(F_i \setminus D_1) \cap (G_j \setminus D_2)| \geq (d-1) - |D_1 \cap D_2|$ for any $1 \leq j \leq k$. This gives a contradiction since

$$|F_i| \ge |D_1 \cap D_2| + \sum_{j=1}^k |(F_i \setminus D_1) \cap (G_j \setminus D_2)| > k.$$

We now note that for $n \ge k(k-d) + d$, the two extremal families $\mathcal{A}(k,d)$ and $\mathcal{HM}(k,d)$ are so that:

$$\mathcal{S}_d(\mathcal{A}(k,d)) = \binom{[d+1]}{d}$$

$$\mathcal{S}_d(\mathcal{HM}(k,d)) = \{A \in \binom{[k+1]}{d} : [d-1] \subseteq A\}$$

As a result of Lemma 2.2.3, we are interested in $S \subseteq {\binom{[n]}{d}}$ which are (d-1)intersecting. In the proof of Theorem 2.0.1, we will iteratively find *d*-sets with large core

degree. The following structural type result yields that the collection of these sets is necessarily isomorphic to a subfamily of the collection of *d*-sets with large core degree in the extremal families $\mathcal{A}(k, d)$ and $\mathcal{HM}(k, d)$.

Lemma 2.2.4. Let $k \ge d \ge 3$ and $\mathcal{F} \subseteq {\binom{[n]}{k}}$ be a non-trivial d-wise intersecting family. Then $\mathcal{S}_d(\mathcal{F})$ is isomorphic to a subfamily of ${\binom{[d+1]}{d}}$ or $\{D \in {\binom{[n]}{d}} : [d-1] \subseteq D\}$. In the latter case, it also follows that $\mathcal{S}_d(\mathcal{F})$ is isomorphic to a subfamily of $\{D \in {\binom{[k+1]}{d}} : [d-1] \subseteq D\}$.

Proof. Given distinct $F_1, F_2 \in \mathcal{S}_d(\mathcal{F})$, we have $|F_1 \cap F_2| = d - 1$ and hence without loss of generality, we may assume that $F_1 = [d]$ and $F_2 = [d - 1] \cup \{d + 1\}$. Now, we let

$$S_1 := \{ F \in S_d(F) \setminus \{F_1, F_2\} : |F \cap [d-1]| = d-2 \}$$

and note that if $F \in S_1$, then $\{d, d+1\} \subseteq F$ as S is (d-1)-intersecting. We then let

$$\mathcal{S}_2 := \{F \in \mathcal{S}_d(\mathcal{F}) \setminus \{F_1, F_2\} : |F \cap [d-1]| = d-1\}$$

so that $S_d(\mathcal{F}) = S_1 \cup S_2 \cup \{F_1, F_2\}$. For all $F \in S_1$ and for all $G \in S_2$, $|F \cap G| = d - 2$ and thus if $S_2 \neq \emptyset$, then $S_1 = \emptyset$. This completes the proof of the first statement.

For the second statement, seeking a contradiction, and without loss of generality, suppose that $\{D \in \binom{[k+2]}{d} : [d-1] \subseteq D\} \subseteq \mathcal{S}_d(\mathcal{F})$. Then, as \mathcal{F} is non-trivial, there exists $X_1 \in \mathcal{F}$ so that $1 \notin X_1$. By Lemma 2.2.2, it follows that $[d, k+2] \subseteq X_1$ and hence $[2, k+2] \subseteq X_1$ which is a contradiction. As a result of Lemma 2.2.4, we think of $\mathcal{S}_d(\mathcal{F}) \subseteq \binom{[k+1]}{d}$. We will now show that if a non-trivial *d*-wise intersecting family $\mathcal{F} \subseteq \binom{[n]}{k}$ has a particular structure of *d*-sets with large core degree, then $|F| \leq |\mathcal{HM}(k, d)|$.

Lemma 2.2.5. Let $k \ge d \ge 3$ and $\mathcal{F} \subseteq {\binom{[n]}{k}}$ be a non-trivial d-wise intersecting family. If $\{D \in {\binom{[k]}{d}} : [d-1] \subseteq D\} \subseteq \mathcal{S}_d(\mathcal{F}), \text{ then } |\mathcal{F}| \le |\mathcal{HM}(k,d)|.$ Moreover, if $|\mathcal{F}| > |\mathcal{B}(k,d)|,$ then \mathcal{F} is necessarily isomorphic to some subfamily of $\mathcal{HM}(k,d)$.

Proof. We have that $D_x := [d-1] \cup \{x\} \in \mathcal{S}_d(\mathcal{F})$ for all $x \in [d, k]$. As a result of Lemma 2.2.2, for all $A \in \mathcal{F}$, $|A \cap [d-1]| \ge d-2$. We let $\mathcal{F}_1 := \{A \in \mathcal{F} : |A \cap [d-1]| = d-2\}$ and $\mathcal{F}_2 := \{A \in \mathcal{F} : |A \cap [d-1]| = d-1\}$. For each $i \in [d-1]$ and each $X_i \in \mathcal{F}$ so that $i \notin X_i$, Lemma 2.2.2 gives that $[k] \setminus \{i\} \subseteq X_i$ since $D_x := [d-1] \cup \{x\} \in \mathcal{S}_d(\mathcal{F})$ for all $x \in [d, k]$.

We now have two cases based on the collection over $i \in [d-1]$ of non-empty 1-uniform link graphs of $[k] \setminus \{i\}$. First, we consider the case where these 1-uniform link graphs do not all have size one consisting of the same vertex. That is, we may find $\{X_1, \ldots, X_{d-1}\} \subseteq \mathcal{F}$ so that $i \notin X_i$ for each $i \in [d-1]$ and so that

$$\bigcap_{i=1}^{d-1} X_i = [d, k].$$

Let $A \in \mathcal{F}_1$ so that $|A \cap [d-1]| = d-2$, then by Lemma 2.2.2 it follows that $[d,k] \subseteq A$. Next, for $A \in \mathcal{F}_2$, if $A \cap [d,k] = \emptyset$, then $A \cap (X_1 \cap \cdots \cap X_{d-1}) = \emptyset$; a

contradiction. Thus $A \cap [d, k] \neq \emptyset$. Putting these together, it follows that:

$$\mathcal{F}_1 \subseteq \{A \in \binom{[n]}{k} : |A \cap [d-1]| = d-2, [d,k] \subseteq A\}$$

$$\mathcal{F}_2 \subseteq \{A \in \binom{[n]}{k} : [d-1] \subseteq A, A \cap [d,k] \neq \emptyset\}.$$

Thus $|\mathcal{F}| = |\mathcal{F}_1| + |\mathcal{F}_2| \le |\mathcal{B}(k,d)| \le |\mathcal{HM}(k,d)|.$

Next, suppose all of the 1-uniform link graphs all have size one consisting of the same vertex (say k + 1). Thus for all $i \in [d - 1]$, there exists a unique set $X_i \in \mathcal{F}$ so that $i \notin X_i$ and that $X_i = [k+1] \setminus \{i\}$. Hence $\mathcal{F}_1 = \{[k+1] \setminus \{i\} : i \in [d-1]\}$. Now, let $A \in \mathcal{F}_2$ be so that $[d-1] \subseteq A$ and suppose that $A \cap [d, k+1] = \emptyset$. Then, $A \cap X_1 \cap \cdots \cap X_{d-1} = \emptyset$; a contradiction. Thus

$$\mathcal{F}_2 \subseteq \{A \in {[n] \choose k} : [d-1] \subseteq A, A \cap [d, k+1] \neq \emptyset\}$$

and hence $|\mathcal{F}| = |\mathcal{F}_1| + |\mathcal{F}_2| \le |\mathcal{HM}(k, d)|$. The second part of the statement holds since if $|\mathcal{F}| > |\mathcal{B}(k, d)|$, we cannot be in the first case and are necessarily in the second case for which \mathcal{F} is isomorphic to a subfamily of $\mathcal{HM}(k, d)$.

We now will prove the analog of Lemma 2.2.5 when $S_d(\mathcal{F})$ is isomorphic to a subfamily of $\binom{[d+1]}{d}$.

Lemma 2.2.6. Let $k \ge d \ge 3$ and $\mathcal{F} \subseteq {\binom{[n]}{k}}$ be a non-trivial d-wise intersecting family. Given that $|\mathcal{S}_d(\mathcal{F})| \ge 3$ and $\mathcal{S}_d(\mathcal{F}) \subseteq {\binom{[d+1]}{d}}$, then $\mathcal{F} \subseteq \mathcal{A}(k, d)$.

Proof. Let $D_1, D_2, D_3 \in \mathcal{S}_d(\mathcal{F})$ be distinct *d*-sets with large core degree. By Lemma 2.2.2 and Lemma 2.2.4, we may assume $D_i = [d+1] \setminus \{i\}$ and that $|A \cap D_i| \ge d-1$ for all $A \in \mathcal{F}$ for i = 1, 2, 3. This then implies that $|A \cap [d+1]| \ge d$ for all $A \in \mathcal{F}$ and thus the result follows.

2.3 The Delta System Method

Given a k-partite hypergraph $\mathcal{H} \subseteq {\binom{[n]}{k}}$ with parts X_1, \ldots, X_k , we let the projection of $E \subseteq [n]$ be

$$\operatorname{proj}(E) = \{i : E \cap X_i \neq \emptyset\}.$$

Let $\mathcal{H}_E = \{X \cap E : X \in \mathcal{H}\}$ be the trace of E on \mathcal{H} and let the *intersection pattern* of E on \mathcal{H} be

$$I_{\mathcal{H}}(E) = \{ \operatorname{proj}(A) : A \in \mathcal{H}|_E \}.$$

We are now able to state Füredi's intersection semilattice lemma [Für83].

Lemma 2.3.1. [Für83] For fixed $s, k \in \mathbb{N}$, there exists c := c(s, k) so that for all $\mathcal{H} \subseteq {\binom{[n]}{k}}$ there exists a k-partite $\mathcal{H}^* \subseteq \mathcal{H}$ with $|\mathcal{H}^*| > c|\mathcal{H}|$ and a $\mathcal{J} \subseteq 2^{[k]}$ so that the following hold:

- 1. For all $A_1, A_2 \in \mathcal{J}, A_1 \cap A_2 \in \mathcal{J}$ (i.e., \mathcal{J} is intersection closed).
- 2. For all $E \in \mathcal{H}^*$ that $I_{\mathcal{H}^*}(E) = \mathcal{J}$.
- 3. For all $X \cap E \in \mathcal{H}^*|_E$, $d^*_{\mathcal{H}^*}(X \cap E) \ge s$.

Given $\mathcal{H} \subseteq {\binom{[n]}{k}}$ so that it satisfies the conclusions of Lemma 2.3.1, we say that \mathcal{H} is (s, \mathcal{J}) -homogeneous and if there exists an $s \in \mathbb{N}$ so that \mathcal{H} is (s, \mathcal{J}) -homogeneous, we

say that \mathcal{H} is \mathcal{J} -homogeneous. Let $\mathcal{J} \subseteq 2^{[k]}$, then the rank of \mathcal{J} , denoted $\rho(J)$ is defined as

$$\rho(\mathcal{J}) := \min\{|E| : E \subseteq [k] : d_{\mathcal{J}}(E) = 0\}$$

where $d_{\mathcal{J}}(E)$ is the *degree* of the set E in \mathcal{J} .

Let $\mathcal{F} \subseteq {\binom{[n]}{k}}$ be non-trivial and *d*-wise intersecting. Applying Lemma 2.3.1 with s = k to large subfamilies $\mathcal{H} \subseteq \mathcal{F}$ when *n* sufficiently large gives a particular intersection structure \mathcal{J} . To this end, let $n_0(k,d) := d + e(k^2 2^k)^{2^k}(k-d)$ and we let $c_k := c(k,k) > (k^2 2^k)^{-2^k}$.

Lemma 2.3.2. Let $\mathcal{H} \subseteq \mathcal{F}$ be \mathcal{J} -homogeneous and so that $|\mathcal{H}| \geq \binom{n-d}{k-d}$ where $n > n_0(k, d)$. Then

$$\mathcal{J} \subseteq \bigcup_{l=d}^{k-1} \binom{[k]}{l}.$$

Moreover, $|\mathcal{J} \cap {\binom{[k]}{d}}| = 1$ so that there exists $D \in \mathcal{S}_d(\mathcal{F})$ as $d^*_{\mathcal{F}}(D) \ge d^*_{\mathcal{H}}(D) \ge k$.

Proof. Lemma 2.1.2 gives that $|A \cap B| \ge d - 1$ for all $A, B \in \mathcal{F}$ which yields that

$$\mathcal{J} \subseteq \bigcup_{l=d-1}^{k-1} \binom{[k]}{l}.$$

Lemma 2.2.1 yields that $\mathcal{J} \cap {\binom{[k]}{d-1}} = \emptyset$ and hence

$$\mathcal{J} \subseteq \bigcup_{l=d}^{k-1} \binom{[k]}{l}.$$

Now, since \mathcal{J} is intersection closed, $|\mathcal{J} \cap {\binom{[k]}{d}}| \leq 1$. If $\mathcal{J} \cap {\binom{[k]}{d}} = \emptyset$, then without loss of generality, suppose that [d+1] is the inclusion minimal element of \mathcal{J} . Suppose there exists $X \in \mathcal{J}$ so that $[d+2,k] \subseteq X$, then there is an $i \in [d+1]$ so that $i \notin X$ as $[k] \notin \mathcal{J}$. Now, $X \cap [d+1] \in \mathcal{J}$, but $|X \cap [d+1]| < (d+1)$ so we necessarily have that $d_{\mathcal{J}}([d+2,k]) = 0$. As a result, $\rho(\mathcal{J}) \le k - d - 1$. Next, if $\mathcal{H}^{\star} \subseteq {\binom{[n]}{k}}$ is \mathcal{J} -homogeneous with $\rho(\mathcal{J}) = r$, then $|\mathcal{H}^{\star}| \le {\binom{n}{r}}$ (see [MV16]). Thus, $|\mathcal{H}^{\star}| \le {\binom{n}{k-d-1}}$. However, for $n > n_0(k,d)$,

$$|\mathcal{H}^{\star}| > c_k |\mathcal{H}| \ge c_k {n-d \choose k-d} > {n \choose k-d-1}.$$

In the case where the minimal element has size strictly larger than d+1, the proof is similar as $\binom{n}{k-d-1} > \binom{n}{k-a}$ for all a > d+1 when $n \ge n_0(k,d)$.

2.4 Proof of Theorem 2.0.1

In this section, we will prove Theorem 2.0.1 by repeated applications of Lemma 2.3.1 and Lemma 2.3.2 and the structural results from Section 2.2.

Proof of Theorem 2.0.1. As a result of Lemma 2.2.5 and Lemma 2.2.6, it suffices to show that $|\mathcal{S}_d(\mathcal{F})| \geq 3$ and that we either have $\mathcal{S}_d(\mathcal{F})$ contains $\{A \in \binom{[k]}{d} : [d-1] \subseteq A\}$ with $|\mathcal{S}_d(\mathcal{F})| = k - d + 1$ or $\mathcal{S}_d(\mathcal{F})$ is isomorphic to a subfamily of $\binom{[d+1]}{d}$. An application of Lemma 2.3.1 and Lemma 2.3.2 yields a *d*-set D_1 which has large core degree. We now consider

$$\mathcal{H}_1 := \{ A \in \mathcal{F} : D_1 \nsubseteq A \}$$

and again applying Lemma 2.3.1 and Lemma 2.3.2 yields a *d*-set D_2 which has large core degree and $D_1 \neq D_2$. We can iteratively apply Lemma 2.3.1 and Lemma 2.3.2 *s* times to get $\{D_1, \ldots, D_s\} \in S_d(\mathcal{F})$ where the particular value of *s* depends on $|\mathcal{F}|$. That is, we need s to be small enough so that the number of sets contained in at least one of $\{D_1, \ldots, D_s\}$ is less than $|\mathcal{F}|$.

In the case where 2d < k, we may suppose that $|\mathcal{F}| > |\mathcal{B}(k,d)| > |\mathcal{A}(k,d)|$ and we take s = k - d + 1. Noting that s > d + 1 then yields that $\mathcal{S}_d(\mathcal{F})$ is not isomorphic to a subfamily of $\binom{[d+1]}{d}$. Lemma 2.2.4 and Lemma 2.2.5 then yield that \mathcal{F} is isomorphic to a subfamily of $\mathcal{HM}(k,d)$.

In the case where 2d = k, we may suppose that $|\mathcal{F}| > |\mathcal{A}(k,d)| > |\mathcal{B}(k,d)|$ and also take s = k - d + 1 where we note k - d + 1 = d + 1. Noting that $|\mathcal{F}| > |\mathcal{A}(k,d)|$ then yields that $\mathcal{S}_d(\mathcal{F})$ is not isomorphic to $\binom{[d+1]}{d}$. Lemma 2.2.4 and Lemma 2.2.5 then yield that \mathcal{F} is isomorphic to a subfamily of $\mathcal{HM}(k,d)$.

In the case where $2d \ge k + 1$, we may suppose that $|\mathcal{F}| > |\mathcal{HM}(k,d)| > |\mathcal{B}(k,d)|$. When d < k - 1, we take $s = k - d + 1 \ge 3$ and when d = k - 1, and an Inclusion-Exclusion argument yields that we may take s = 3. In both of these cases, noting that $|\mathcal{F}| > |\mathcal{HM}(k,d)|$ then yields that $\mathcal{S}_d(\mathcal{F})$ is not isomorphic to a subfamily of $\{A \subseteq [k] : [d-1] \subseteq A\}$ by Lemma 2.2.5. Thus we may use Lemma 2.2.4 and Lemma 2.2.6 to get that \mathcal{F} is isomorphic to a subfamily of $\mathcal{A}(k,d)$.

This Chapter contains material from: J. O'Neill, J. Verstraëte, "Non-trivial *d*-wise intersecting families", J. Combin. Theory Ser. A, 178, 2021. The dissertation author was one of the primary investigators and authors of this paper.

Chapter 3

Supersaturation for Oddtown and

Eventown

3.1 Proof of Theorem 1.3.6

It suffices to consider $\mathcal{A} = \{A_1, \ldots, A_{n+1}\}$ with $|A_i|$ odd for all $i \in [n+1]$ and to prove that we cannot have $op(\mathcal{A}) \leq 2$. Let $v_i := v_{A_i} \in \mathbb{F}_2^n$ for each $i \in [n+1]$ and consider the vector space

$$V = \{ (\epsilon_1, \dots, \epsilon_{n+1}) \in \mathbb{F}_2^{n+1} : \epsilon_1 v_1 + \dots + \epsilon_{n+1} v_{n+1} = 0 \}.$$
 (3.1)

Without loss of generality, it suffices to consider three cases:

Case 1: There exists one such odd intersection pair which we may assume is $\{A_1, A_2\}$. Let $(\epsilon_1, \ldots, \epsilon_{n+1}) \in V$ be nonzero. Taking inner products with each v_i and $\epsilon_1 v_1 + \cdots + \epsilon_{n+1} v_{n+1}$, we obtain $\epsilon_3 = \epsilon_4 = \cdots = \epsilon_{n+1} = 0$ and $\epsilon_1 = \epsilon_2$. If $\epsilon_1 = \epsilon_2 = 1$, then $0 = v_1 + v_2$ which implies $A_1 = A_2$; a contradiction.

Case 2: There exists two such odd intersection pairs which we may assume are $\{A_1, A_2\}$ and $\{A_1, A_3\}$. It then follows that $\{v_2, \ldots, v_{n+1}\}$ is linearly independent. Consider the nonzero vector $(\epsilon_1, \ldots, \epsilon_{n+1}) \in V$. If $\epsilon_1 = 0$, then $0 = \epsilon_2 v_2 + \cdots + \epsilon_{n+1} v_{n+1}$ and hence $\epsilon_2 = \epsilon_3 = \cdots = \epsilon_{n+1} = 0$; a contradiction. Thus, we must have $\epsilon_1 = 1$. Taking inner products with each v_i and $\epsilon_1 v_1 + \cdots + \epsilon_{n+1} v_{n+1}$, we obtain $\epsilon_1 + \epsilon_2 = 0$ and $\epsilon_1 + \epsilon_3 = 0$ and $\epsilon_4 = \epsilon_5 = \cdots = \epsilon_{n+1} = 0$. Using that $\epsilon_1 = 1$, it follows that $\epsilon_1 = \epsilon_2 = \epsilon_3 = 1$. As such, $v_1 + v_2 + v_3 = 0$ which implies that each element in $A_1 \cup A_2 \cup A_3$ is in exactly two of $\{A_1, A_2, A_3\}$. Thus, for each $x \in A_1$, it follows that $x \in A_2$ or $x \in A_3$ but not both and hence we reach a contradiction as

$$|A_1| = |A_1 \cap A_2| + |A_1 \cap A_3| = 1 + 1 = 0 \pmod{2}.$$

Case 3a: There exists two such odd intersection pairs which we may assume are $\{A_1, A_2\}$ and $\{A_3, A_4\}$ and dim(V) = 2. By the rank-nullity theorem,

$$\dim(\operatorname{span}\{v_1,\ldots,v_{n+1}\})=n-1$$

and observe that $\{v_1, v_3, v_5, v_6, \ldots, v_{n+1}\}$ is an orthogonal basis. Let $(\epsilon_1, \ldots, \epsilon_{n+1}) \in V$ be nonzero. For all $i \geq 5$, taking inner products with v_i and $\epsilon_1 v_1 + \cdots + \epsilon_{n+1} v_{n+1}$, we obtain $\epsilon_i = 0$. For $i \leq 4$, taking inner products with v_i and $\epsilon_1 v_1 + \cdots + \epsilon_{n+1} v_{n+1}$ we recover that $0 = \epsilon_1 + \epsilon_2$ and $0 = \epsilon_3 + \epsilon_4$. Hence $\epsilon_1 = \epsilon_2 \in \{0, 1\}$ and $\epsilon_3 = \epsilon_4 \in \{0, 1\}$. Suppose $\epsilon_1 = \epsilon_2 = 1$ and $\epsilon_3 = \epsilon_4 = 0$. Then $0 = \epsilon_1 v_1 + \cdots + \epsilon_{n+1} v_{n+1} = v_1 + v_2$ and we reach a similar contradiction as in Case 1. Similarly, we cannot have $\epsilon_1 = \epsilon_2 = 0$ and $\epsilon_3 = \epsilon_4 = 1$. As such, the only nonzero vector in V is $(1, 1, 1, 1, 0, \ldots, 0)$ which contradicts the dimension of V.

Case 3b: There exists two such odd intersection pairs which we may assume is $\{A_1, A_2\}$ and $\{A_3, A_4\}$ and $\dim(V) = 1$. The reasoning from Case 3a gives that the vector space $V = \{0, (1, 1, 1, 1, 0, \dots, 0)\}$. By the rank-nullity theorem, span $\{v_1, \dots, v_{n+1}\} = \mathbb{F}_2^n$. Clearly any basis for \mathbb{F}_2^n consisting of n vectors from $\{v_1, \dots, v_{n+1}\}$ cannot contain all of the elements in $\{v_1, v_2, v_3, v_4\}$ as $v_1 + v_2 + v_3 + v_4 = 0$. Without loss of generality, assume that $\{v_2, \dots, v_{n+1}\}$ forms a basis for \mathbb{F}_2^n and consider $U = \text{span}\{v_3, v_4\}$. It then follows that $\dim(U^{\perp}) = n - 2$ and $\{v_2, v_5, v_6, \dots, v_{n+1}\}$ forms a basis for U^{\perp} . As such, since $v_1 \in U^{\perp}$, we have $v_1 = \alpha_2 v_2 + \alpha_5 v_5 + \dots + \alpha_{n+1} v_{n+1}$. However, this is a contradiction as $v_1 = v_2 + v_3 + v_4$ and $\{v_2, \dots, v_{n+1}\}$ is a basis for \mathbb{F}_2^n .

3.2 Proof of Theorem 1.3.4

3.2.1 The case of s = 1

Let $\mathcal{A} = \{A_1, \ldots, A_m\} \subseteq 2^{[n]}$ be a collection of even-sized subsets where $m = 2^{\lfloor n/2 \rfloor} + 1$ and suppose $\mathcal{A}' \subseteq \mathcal{A}$ is a maximal (with respect to \mathcal{A}) eventown subfamily of \mathcal{A} . If $|\mathcal{A}'| \leq 2^{\lfloor n/2 \rfloor - 1}$, then since for each $A \in \mathcal{A} \setminus \mathcal{A}'$, there exists $B \in \mathcal{A}'$ for which $|A \cap B|$ is odd,

$$\operatorname{op}(\mathcal{A}) \ge |\mathcal{A} \setminus \mathcal{A}'| \ge 2^{\lfloor n/2 \rfloor} + 1 - 2^{\lfloor n/2 \rfloor - 1} > 2^{\lfloor n/2 \rfloor - 1}.$$

If $|\mathcal{A}'| = t > 2^{\lfloor n/2 \rfloor - 1}$, then without loss of generality assume that $\mathcal{A}' = \{A_1, \ldots, A_t\}$. Setting $v_i = v_{A_i} \in \mathbb{F}_2^n$ as the characteristic vector of A_i , we can consider the vector space $W = \operatorname{span}\{v_1, \ldots, v_t\}$. Then, $W \subseteq W^{\perp}$ and since $\dim(\mathbb{F}_2^n) = \dim(W) + \dim(W^{\perp})$, it follows that $\dim(W) \leq \lfloor n/2 \rfloor$. Further, as $t > 2^{\lfloor n/2 \rfloor - 1}$, a straightforward counting argument gives that we must have that $\dim(W) = \lfloor n/2 \rfloor$.

For each A_i with i > t, we consider the linear functional $\chi_i : W \to \mathbb{F}_2$ given by $w \mapsto \langle w, v_i \rangle$. By the first isomorphism theorem, it follows that $W/\ker(\chi_i) \cong \mathbb{F}_2$ and $\dim(\ker(\chi_i)) = \lfloor n/2 \rfloor - 1$. As a result, $|\{v_1, \ldots, v_t\} \setminus \ker(\chi_i)\rangle| \ge t - 2^{\lfloor n/2 \rfloor - 1}$. Hence there are at least $t - 2^{\lfloor n/2 \rfloor - 1}$ sets in \mathcal{A}' which have odd-sized intersection with A_i . Since we can do this for all $t < i \le m$, it follows that

$$\operatorname{op}(\mathcal{A}) \ge \left(t - 2^{\lfloor n/2 \rfloor - 1}\right) \cdot (m - t).$$
(3.2)

The result then follows by an optimization problem and considering all positive integers $t: 2^{\lfloor n/2 \rfloor - 1} < t \leq 2^{\lfloor n/2 \rfloor}.$

3.2.2 The case of s = 2

Let $\mathcal{A} = \{A_1, \ldots, A_m\} \subseteq 2^{[n]}$ be a collection of even-sized subsets with $m = 2^{\lfloor n/2 \rfloor} + 2$. Let $\mathcal{A}' \subseteq \mathcal{A}$ be a maximal (with respect to \mathcal{A}) eventown subfamily of \mathcal{A} of maximum size. If $|\mathcal{A}'| \leq 2^{\lfloor n/2 \rfloor - 2} + 1$, then we may partition \mathcal{A} by $\mathcal{A} = \mathcal{A}'_1 \sqcup \mathcal{A}'_2 \sqcup \cdots \sqcup \mathcal{A}'_l$ with $l \geq 4$ so that for each $A_i \in \mathcal{A}_i$ and each j < i, there exists $B_j \in \mathcal{A}_j$ with $|A_i \cap B_j|$ odd and hence

$$op(\mathcal{A}) \ge |\mathcal{A} \setminus \mathcal{A}'_1| + |\mathcal{A} \setminus (\mathcal{A}'_1 \sqcup \mathcal{A}'_2)| + \dots + |\mathcal{A}'_l| \ge \frac{3}{4} \cdot 2^{\lfloor n/2 \rfloor} + \frac{1}{2} \cdot 2^{\lfloor n/2 \rfloor} > 2^{\lfloor n/2 \rfloor}.$$

If $|\mathcal{A}'| \in [2^{\lfloor n/2 \rfloor - 2} + 2, 2^{\lfloor n/2 \rfloor - 1} - 1]$ or $|\mathcal{A}'| \ge 2^{\lfloor n/2 \rfloor - 1} + 2$, then we are done by (3.2).

If $|\mathcal{A}'| = 2^{\lfloor n/2 \rfloor - 1} + 1 = t$, then without loss of generality assume that the maximal eventown subfamily is $\mathcal{A}' = \{A_1, \ldots, A_t\}$. Setting $v_i = v_{A_i} \in \mathbb{F}_2^n$ as the characteristic vector of A_i , we let $W = \operatorname{span}\{v_1, \ldots, v_t\}$. As in Section 3.2.1, we note that $\dim(W) = \lfloor n/2 \rfloor$ and will consider the linear functional $\chi_i : W \to \mathbb{F}_2$ given by $w \mapsto \langle w, v_i \rangle$ with $W_i := \ker(\chi_i)$. We then set $\mathcal{B}' = \mathcal{A} \setminus \mathcal{A}'$ and further consider $\mathcal{B}'' \subseteq \mathcal{B}'$ to be the sets $B \in \mathcal{B}'$ for which $\operatorname{op}(\mathcal{A}' \cup \{B\}) = 1$. We now consider a few cases based on $|\mathcal{B}''|$:

Case 1: $|\mathcal{B}''| \leq 1$

For each $B \in \mathcal{B}' \setminus B''$, there exists at least two sets in \mathcal{A}' with odd-sized intersections with B, so

$$\operatorname{op}(\mathcal{A}) \ge |\mathcal{B}''| + 2 \cdot |\mathcal{B}' \setminus \mathcal{B}''| \ge 1 + 2(2^{\lfloor n/2 \rfloor - 1}) > 2^{\lfloor n/2 \rfloor}.$$

Case 2: $2 \le |\mathcal{B}''| \le 2^{\lfloor n/2 \rfloor - 2}$

As $\mathcal{B}'' \neq \emptyset$, there exists an i > t where $W_i = \ker(\chi_i)$ is so that $|\{v_1, \ldots, v_t\} \setminus W_i| = 1$. For

each $A_j \in \mathcal{B}' \setminus \mathcal{B}''$, we know that $|\{v_1, \ldots, v_t\} \setminus W_j| \ge 2$ and in particular $W_i \neq W_j$. As W_i and W_j are vector subspaces of \mathbb{F}_2^n of dimension $\lfloor n/2 \rfloor - 1$, it follows that $|W_i \cap W_j| \le 2^{\lfloor n/2 \rfloor - 2}$ and hence $|\{v_1, \ldots, v_t\} \setminus W_j| \ge |W_i \setminus W_j| \ge 2^{\lfloor n/2 \rfloor - 2}$. As such, we get

$$\operatorname{op}(\mathcal{A}) \ge |\mathcal{B}''| + (2^{\lfloor n/2 \rfloor - 2}) \cdot |\mathcal{B}' \setminus \mathcal{B}''| > 2^{\lfloor n/2 \rfloor}.$$
(3.3)

Case 3: $|\mathcal{B}''| \ge 2^{\lfloor n/2 \rfloor - 2}$

Let $\mathcal{A}'' = \{A \in \mathcal{A}' : \exists B \in \mathcal{B}'' \text{ so that } |A \cap B| \text{ is odd}\}$. We first consider the case where $|\mathcal{A}''| \leq n$ and then will show that we actually necessarily have $|\mathcal{A}''| \leq n$. In this case, there exists $A_i \in \mathcal{A}''$ for which $\operatorname{op}(\mathcal{B}'' \cup \{A_i\}) \geq \frac{1}{n} \cdot 2^{\lfloor n/2 \rfloor - 2}$. Let $B_1'', B_2'' \in \mathcal{B}''$ be so that $|A_i \cap B_1''|$ and $|A_i \cap B_2''|$ are both odd. Then $|B_1'' \cap B_2''|$ is odd as otherwise $\mathcal{A}' \setminus \{A_i\} \cup \{B_1'', B_2''\}$ is a eventown subfamily of \mathcal{A} of larger size. As a result,

$$\operatorname{op}(\mathcal{A}) \ge {\binom{\frac{1}{n} \cdot 2^{\lfloor n/2 \rfloor - 2}}{2}} > 2^{\lfloor n/2 \rfloor}.$$
(3.4)

Seeking a contradiction, suppose that $|\mathcal{A}''| \geq n+1$. As each element from \mathcal{B}'' has an odd number of elements in common with exactly one set from \mathcal{A}'' , we may find distinct $X_1, \ldots, X_{n+1} \in \mathcal{A}''$ and $Y_1, \ldots, Y_{n+1} \in \mathcal{B}''$ so that $|X_i \cap Y_i|$ is odd for all $i \in [n+1]$ and $|X_i \cap Y_j|$ is even for all $i \neq j \in [n+1]$. As $\mathcal{A} \subseteq 2^{[n]}$, this violates Theorem 1.4.3 as there are n+1 such set pairs.

If $|\mathcal{A}'| = 2^{\lfloor n/2 \rfloor}$ and $\dim(W) = \lfloor n/2 \rfloor$, then we may argue in a similar fashion as above. If $\dim(W) < \lfloor n/2 \rfloor$, then the result follows by (3.2). Thus we have shown $\operatorname{op}(\mathcal{A}) \geq 2^{\lfloor n/2 \rfloor}$ in all cases.

This chapter 3 contains material from: J. O'Neill, "Towards supersaturation for

oddtown and eventown", arxiv:2109.09925, 2021. The dissertation author was the sole author of this paper.

Chapter 4

On k-wise oddtown problems

Let $K_{n:k}$ denote the Kneser graph with $V(K_{n:k}) = {\binom{[n]}{k}}$ and $(A, B) \in E(K_{n:k})$ if and only if $A \cap B = \emptyset$. Given pairwise disjoint subsets X_1, \ldots, X_k of [n], let $\prod_{i=1}^k X_i$ denote the complete k-partite k-graph with edges consisting of exactly one vertex from each part X_1, \ldots, X_k and $X_1 \sqcup \cdots \sqcup X_k$ denote their disjoint union. For k = 2, we refer to complete bipartite graphs as *bicliques*.

4.1 Hypergraph Covering Problems

In this section, we relate the problem at hand to a decomposition problem of a particular hypergraph. We first define the corresponding collection of set families.

Definition 1. Let $\mathcal{A}_j = \{A_{j,i} : 1 \leq i \leq m\}$ for $j \in [k]$. Then $(\mathcal{A}_1, \mathcal{A}_2, \dots, \mathcal{A}_k)$ forms a Bollobás set (k, t)-tuple modulo 2 if

$$\left|\bigcap_{i \in [k]} A_{j,i_j}\right| = 0 \pmod{2} \iff |\{i_1, \dots, i_k\}| < t.$$

Let $b_{k,t}(n)$ be the size of the largest Bollobás set (k,t)-tuple modulo 2 with ground set [n].

In the non-modular setting, Bollobás set (k, t)-tuples were first explored by the authors [OV21a] and are a generalization of the well-studied Bollobás set pairs [Bol65] when k = t = 2. Using the language of Definition 1, Theorems 1.4.4 and 1.4.5 (i.e. the "reverse" versions) say $b_{k,k}(n) = \Theta(n^{1/\lfloor k/2 \rfloor})$ and $b_{k,t}(n) = \Theta(n^{1/(t-1)})$ respectively. In order to determine the asymptotics of $b_{k,t}(n)$, we consider the following related hypergraph covering problem: **Definition 2.** Set $X_i = \{x_{i,j} : j \in [n]\}$ for $i \in [k]$. For $H \subseteq \prod_{i \in [k]} X_i$, a modulo 2 cover of H is a collection of complete k-partite subgraphs of $\prod_{i \in [k]} X_i$ which cover each edge of H an odd number of times and each non-edge an even number of times. We denote the minimum size of a modulo 2 cover of H by $f_k(H)$.

Let $H_{k,t}(n)$ be the subhypergraph of $\prod_{i \in [k]} X_i$ defined by

$$H_{k,t}(n) = \{\{x_{1,i_1}, x_{2,i_2}, \dots, x_{k,i_k}\} : |\{i_1, \dots, i_k\}| \ge t\}.$$

Then, as in [OV21a, Section 1.2] there is a one to one correspondence between a modulo 2 cover of $H_{k,t}(n)$ with m complete k-partite k-graphs and a Bollobás set (k, t)-tuple modulo 2 consisting of subsets of [m]. Hence,

$$f_{k,t}(n) := f_k(H_{k,t}(n)) = \min\{m : b_{k,t}(m) \ge n\}.$$
(4.1)

The problem in Definition 2 is closely related to the hypergraph extension of the Graham-Pollak [GP72] problem. Given a hypergraph $H \subseteq {\binom{[n]}{k}}$, let $g_k(H)$ denote minimum number of complete k-partite subgraphs of H needed to cover each edge of H exactly once and let $g_k(n) = g_k({\binom{[n]}{k}})$.

Graham and Pollak [GP72] proved for all $n \ge 1$, $g_2(n) = n-1$. Alon [Alo86] proved that $g_3(n) = n - 2$ for $n \ge 2$ and proved that $g_k(n) = \Theta(n^{\lfloor k/2 \rfloor})$. Relating the problem to that of a minimal biclique cover of the Kneser graph $K_{n:k}$, Cioabă et al. [CKV09] improved the bounds on $g_k(n)$ to:

$$\frac{2\binom{n-1}{k}}{\binom{2k}{k}} \le g_{2k}(n) \le \binom{n-k}{k}.$$
(4.2)

Currently Leader, Milićević, and Tan [LMT18] have the best general bounds. A key ingredient in our proof of Theorem 1.4.4 is to similarly relate this hypergraph covering problem to a graph covering problem as in Cioabă et al. [CKV09]. To this end, define the *ordered* Kneser graph $OK_{n:k}$ as follows:

$$V(OK_{n:k}) = \{(i_1, i_2, \dots, i_k) : \{i_1, \dots, i_k\} \in {\binom{[n]}{k}}\}$$
$$E(OK_{n:k}) = \{((i_1, \dots, i_k), (j_1, \dots, j_k)) : \{i_1, \dots, i_k\} \cap \{j_1, \dots, j_k\} = \emptyset\}.$$

The ordered Kneser graph is a k! blowup of the Kneser graph $K_{n:k}$ and our interest in it is the following lemma which relates modulo 2 covers of $H_{2k,2k}(n)$ and biclique covers of $OK_{n:k}$.

Lemma 4.1.1. Let $\{C_i : i \in [t]\}$ be a modulo 2 cover of $H_{2k,2k}(n)$. Then, there exists a modulo 2 biclique cover $\{C'_i : i \in [t]\}$ of $OK_{n:k}$ and in particular $f_{2k,2k}(n) \ge f_2(OK_{n:k})$.

Proof. For each complete 2k-partite 2k-graph $C_i := \prod_{j=1}^{2k} X_{i,j}$ in a modulo 2 cover of $H_{2k,2k}(n)$, define the complete bipartite graph C'_i with parts $\prod_{j=1}^k X_{i,j} \cap V(OK_{n:k})$, and $\prod_{j=k+1}^{2k} X_{i,j} \cap V(OK_{n:k})$. Observe that $\{C'_i\}$ is a modulo 2 biclique cover of $OK_{n:k}$.

Our next lemma shows a connection between the k-graphs $H_{k,k}(n)$ and $\binom{[n]}{k}$.

Lemma 4.1.2. Let $k \ge 2$. Then $g_k(H_{k,k}(n)) \le k!g_k(n)$.

Proof. Let $m = g_k(n)$ and $\{\mathcal{C}_i : i \in [m]\}$ be such a minimal k-partite, k-uniform cover of $\binom{[n]}{k}$. Then, given $\mathcal{C}_i = \prod_{i \in [k]} C_i$ and a bijection $\pi : [k] \to [k]$, define

$$\mathcal{C}_{i,\pi} = \prod_{i \in [k]} C_{\pi(i)} \subseteq H_{k,k}(n).$$

It follows that $\{C_{i,\pi}\}$ forms a decomposition of $H_{k,k}(n)$ and implies the bound.

4.2 Proof of Theorem 1.4.4

4.2.1 Two families

In the setting where t = k = 2, using a straightforward argument, we are able to prove a "reverse" bipartite oddtown theorem:

Proposition 4.2.1. For all $n \ge 1$,

m

$$b_{2,2}(n) = \begin{cases} n & n \text{ is odd} \\ \\ n+1 & n \text{ is even.} \end{cases}$$

Proof. For n even, and $i \in [n]$ consider $A_i = \{i\}$ with $B_i = \{i\}^c$ and $B_{n+1} = A_{n+1} = [n]$. This is a Bollobás set pair modulo 2 of size n + 1 and thus $b_{2,2}(n) = n + 1$ for even n by the bounds in (1.2).

For n odd, let $m = b_{2,2}(n)$ and $(\mathcal{A}, \mathcal{B})$ be a Bollobás set pair modulo 2 of size m. If $\{v_{A_i} : A_i \in \mathcal{A}\}$ is linearly independent in \mathbb{F}_2^n , then $m \leq n$. Thus, we may assume that there exists a non-trivial solution to $\sum_{i=1}^m \epsilon_i v_{A_i} = 0$ in \mathbb{F}_2^n . For all $j \in [m]$, it follows that

$$0 = \left\langle \sum_{i=1}^{m} \epsilon_i v_{A_i}, v_{B_j} \right\rangle = \sum_{\epsilon_i = 1; i \neq j} \left\langle v_{A_i}, v_{B_j} \right\rangle = \sum_{\epsilon_i = 1; i \neq j} |A_i \cap B_j| = |\{i : \epsilon_i = 1, i \neq j\}|$$

over the field \mathbb{F}_2 . Since this is true for all $j \in [m]$, the existence of one non-zero ϵ_i implies they are all equal to one. In particular, (m-1) is even and hence $m \in \{n, n+1\}$ is odd and thus m = n. For n odd, taking $A_i = \{i\}$ with $B_i = \{i\}^c$ for all $i \in [n]$ shows this is best possible.

4.2.2 More than two families

In this section, for $k \geq 2$, we will prove that

$$\left(\frac{1}{2} + o(1)\right) \binom{n}{k} \leq f_{2k,2k}(n) \leq (2k)! \binom{n}{k}$$

$$(4.3)$$

$$\left(\frac{1}{2} + o(1)\right) \binom{n}{k-1} \leq f_{2k-1,2k-1}(n) \leq (2k-1)! \binom{n}{k-1}$$
(4.4)

which imply Theorem 1.4.4 by (4.1) and that the bound in the theorem is the correct order of magnitude. Using Lemma 4.1.2 with the upper bound in (4.2), we get the upper bounds in (4.3) and (4.4).

By considering the link of the vertex $x_{1,1}$ in $H_{2k,2k}(n)$ and in each complete 2kpartite 2k-graph in a minimal modulo 2 cover of $H_{2k,2k}(n)$, it follows that $f_{2k,2k}(n) \ge f_{2k-1,2k-1}(n-1)$. As a result, the lower bound in (4.4) follows by the corresponding lower
bound in (4.3). As such, it suffices to prove the lower bound in (4.3). By Lemma 4.1.1,
it then suffices to prove that $f_2(OK_{n:k}) \ge (1/2 + o(1)) {n \choose k}$.

As the adjacency matrix of each biclique has rank at most 2, the subadditivity of matrix rank and taking the minor of $OK_{n:k}$ which corresponds to the Kneser graph $K_{n:k}$ give

$$f_2(OK_{n:k}) \ge \frac{1}{2} \operatorname{rank}(A(OK_{n:k})) \ge \frac{1}{2} \operatorname{rank}(A(K_{n:k}))$$
 (4.5)

where the rank of the matrices is over \mathbb{F}_2 . Let $M_{n,k,l} \in \{0,1\}^{\binom{n}{k} \times \binom{n}{l}}$ where the (K, L) entry

for $K \in {\binom{[n]}{k}}$ and $L \in {\binom{[n]}{l}}$ is the indicator of $K \subseteq L$ and note that $M_{n,k,n-k} = A(K_{n:k})$.

Wilson [Wil90, Theorem 1] determined the rank for matrices $M_{n,k,l}$ over \mathbb{F}_p for $k \leq \min\{l, n-l\}$:

$$\operatorname{rank}(M_{n,k,l}) = \sum_{i: p \nmid \binom{l-i}{k-i}} \binom{n}{i} - \binom{n}{i-1}.$$
(4.6)

Applying (4.6) when p = 2 and l = n - k gives the following for the rank over \mathbb{F}_2 :

$$\operatorname{rank}(A(K_{n:k})) \ge \binom{n}{k} - \binom{n}{k-1} = (1+o(1))\binom{n}{k}.$$
(4.7)

Hence, using (4.5) and (4.7) give

$$f_{2k,2k}(n) \ge f_2(OK_{n:k}) \ge \left(\frac{1}{2} + o(1)\right) \binom{n}{k}$$

Taking the bounds from (4.3) and (4.4) and noting in (4.1) that $f_{k,k}(n) = \min\{m : b_{k,k}(m) \ge n\}$, we get the following bounds on $b_{2k,2k}(n)$ and $b_{2k-1,2k-1}(n)$:

$$\left(\frac{1}{2k} + o(1)\right) n^{\frac{1}{k}} \leq b_{2k,2k}(n) \leq \left(k2^{\frac{1}{k}}\right) n^{\frac{1}{k}}$$
$$\left(\frac{1}{2k^2} + o(1)\right) n^{\frac{1}{k-1}} \leq b_{2k-1,2k-1}(n) \leq \left((k-1)2^{\frac{1}{k-1}} + o(1)\right) n^{\frac{1}{k-1}}.$$

4.3 Proof of Theorem 1.4.5

In this section, we will prove that if $2t - 2 \le k$, then

$$f_{k,t}(n) \le \left(\frac{t^k}{t!} + o(1)\right) n^{t-1}$$
 and $b_{k,t}(n) \le \left(t - 1 + o(1)\right) n^{\frac{1}{t-1}}$. (4.8)

Since $f_{k,t}(n) = \min\{m : b_{k,t}(m) \ge n\}$, the upper bound on $f_{k,t}(n)$ in (4.8) implies a lower bound on $b_{k,t}(n)$ which together with the upper bound on $b_{k,t}(n)$ in (4.8) proves Theorem 1.4.5 as they imply

$$\left(\left(\frac{t!}{t^k}\right)^{\frac{1}{t-1}} + o(1)\right)n^{\frac{1}{t-1}} \le b_{k,t}(n) \le \left(t - 1 + o(1)\right)n^{\frac{1}{t-1}}.$$

4.3.1 Upper bound on $f_{k,t}(n)$

Let $\phi : [n]^k \to \{\pi : \pi \text{ set partition of } [k]\}$ be the correspondence of an edge of $[n]^k$ to a set partition of [k] where each part consists of the coordinates with equal indices. For a set partition π of [k], define $H_k^{\pi}(n) = \{e \in [n]^k : \phi(e) = \pi\}$ so that

$$H_{k,t}(n) = \bigsqcup_{|\pi| \ge t} H_k^{\pi}(n).$$
(4.9)

It follows that $f_k(H_k^{\pi}(n)) \leq n^{|\pi|}$ as the number of edges in $H_k^{\pi}(n)$ is at most $n^{|\pi|}$ and one may take a modulo 2 cover consisting of single edges. Moreover, adding the complete k-partite k-graph $[n]^k$ to a minimal cover, it follows that

$$f_{k,t}(n) \le 1 + f_k([n]^k \setminus H_{k,t}(n)).$$
 (4.10)

Since $f_k(\cdot)$ is subadditive with respect to disjoint unions, using (4.9) gives

$$f_k([n]^k \setminus H_{k,t}(n)) \le \sum_{|\pi| \le t-1} f_k(H_k^{\pi}(n)) \le \sum_{|\pi| \le t-1} n^{|\pi|} = (S(k,t-1) + o(1))n^{t-1}$$
(4.11)

where S(k, t) is the Stirling number of the second kind. We then get the upper bound on $f_{k,t}(n)$ in (4.8) by using the bound $S(k, t-1) \leq S(k, t) \leq t^k/t!$ together with (4.10) and (4.11).

4.3.2 Upper bound on $b_{k,t}(n)$

We first prove the following lemma which relates $b_{k,t}(n)$ to $b_{2,2}(n)$.

Lemma 4.3.1. Given $n \ge 1$, let $\alpha = \alpha(k, t) = 2t - k + 2$. Then letting $m = b_{k,t}(n)$,

$$b_{2,2}(n) \ge \begin{cases} \binom{m}{t-1} & \text{if } 2t - 2 \le k \\ \binom{m-\alpha}{k-t+1} & \text{if } 2t - 2 > k \end{cases}$$

In particular, both quantities are less than or equal to n + 1 by (1.2).

Proof. Let $m = b_{k,t}(n)$ for $2t - 2 \leq k$ and let $(\mathcal{A}_1, \ldots, \mathcal{A}_k)$ be a Bollobás set (k, t)tuple modulo 2 of size m where $\mathcal{A}_i = \{A_{j,i} \subseteq [n] : i \in [m]\}$ for all $j \in [k]$. For each $I = \{i_1 < \cdots < i_{t-1}\} \in {[m] \choose t-1}$, define the families $\mathcal{A} := \{A_I\}$ and $\mathcal{B} := \{B_I\}$ where:

$$A_I := A_{1,i_1} \cap \cdots \cap A_{t-1,i_{t-1}}$$
$$B_I := A_{t,i_1} \cap \cdots \cap A_{2t-2,i_{t-1}}$$

Note that $(\mathcal{A}, \mathcal{B})$ is a Bollobás set pair modulo 2 and hence the result follows.

Similarly, for 2t - 2 > k, let $m = b_{k,t}(n)$ and let $(\mathcal{A}_1, \ldots, \mathcal{A}_k)$ be a Bollobás set (k, t)-tuple modulo 2 of size m where $\mathcal{A}_i := \{A_{j,i} \subseteq [n] : i \in [m]\}$ for all $j \in [k]$. Consider the index set

$$\mathcal{F} := \binom{[\alpha+1,m]}{k-t+1}$$

For $I = \{i_1 < \cdots < i_{k-t+1}\} \in \mathcal{F}$, define the families $\mathcal{A} := \{A_I\}$ and $\mathcal{B} := \{B_I\}$ where

$$A_I := A_{1,1} \cap \cdots \cap A_{\alpha,\alpha} \cap A_{\alpha+1,i_1} \cap \cdots \cap A_{t-1,i_{k-t+1}}$$
$$B_I := A_{k-t+2,i_1} \cap \cdots \cap A_{k,i_{t-1}}.$$

Note that $(\mathcal{A}, \mathcal{B})$ is a Bollobás set pair modulo 2 and hence the result follows.

We are now able to prove the upper bound on $b_{k,t}(n)$ in (4.8) when $2t - 2 \leq k$. Letting $m = b_{k,t}(n)$, Lemma 4.3.1 yields that $\binom{m}{t-1} \leq n+1$. Using the bound $\binom{m}{t-1} \geq (m/(t-1))^{t-1}$, it follows that

$$m \le \left(t - 1 + o(1)\right) n^{\frac{1}{t-1}}.$$

This chapter contains material from: J. O'Neill, J. Verstraëte, "A note on k-wise oddtown problems", arxiv:2011.09402, 2020. The dissertation author was one of the primary investigators and authors of this paper.

Chapter 5

A generalization of the Bollobás set

pairs inequality

5.1 Proof of Theorem 1.5.2

Given a Bollobás set (k, t)-tuple $(\mathcal{A}_1, \ldots, \mathcal{A}_k)$ with $\mathcal{A}_j = \{A_{j,i} : 1 \le i \le m\}$ and a surjection $\phi : [k] \to [t]$, consider $\mathcal{A}_{\ell}(\phi) : 1 \le \ell \le t$ where $\mathcal{A}_{\ell}(\phi) = \{A_{\ell,i}(\phi) : 1 \le i \le m\}$ and

$$A_{\ell,i}(\phi) = \bigcap_{h:\phi(h)=\ell} A_{h,i}.$$

It follows that $(\mathcal{A}_1(\phi), \ldots, \mathcal{A}_t(\phi))$ is a Bollobás set (t, t)-tuple and hence it suffices to prove Theorem 1.5.2 in the case where t = k. In this setting, surjections $\phi : [k] \to [k]$ simply permute the k families and as such we suppress the notation of ϕ for the remainder of this section. The proof of Theorem 1.5.1, given a Bollobás set pair, defines a collection of permutations \mathcal{C}_i for $i \in [m]$ and shows that these chains are necessarily disjoint. Similarly, given a Bollobás set (k, k)-tuple, we will define a collection of chains \mathcal{C}_{σ} for every ordered collection σ of (k-1) distinct indices of [m] and show these chains are pairwise disjoint.

Let $(\mathcal{A}_1, \ldots, \mathcal{A}_k)$ with $\mathcal{A}_j = \{A_{j,i} : 1 \le i \le m\}$ be a Bollobás set (k, k)-tuple, and

 set

$$X = \bigcup_{i=1}^{m} (A_{1,i} \cup A_{2,i} \cup \dots \cup A_{k,i})$$

with |X| = n. For $\sigma \in [m]_{(k-1)}$, define a subset \mathscr{C}_{σ} of permutations $\pi : X \to [n]$ by

$$\mathscr{C}_{\sigma} := \left\{ \pi : X \to [n] : \max_{x \in A_{1,\sigma}} \pi(x) < \min_{y \in A_{2,\sigma}} \pi(y) \le \max_{y \in A_{2,\sigma}} \pi(y) < \dots < \min_{z \in A_{k,\sigma}} \pi(z) \right\}.$$

Letting $U_{\sigma} := A_{1,\sigma} \cup \cdots \cup A_{k,\sigma}$, elementary counting methods give

$$|\mathscr{C}_{\sigma}| = \binom{n}{|U_{\sigma}|} |A_{1,\sigma}|! \cdots |A_{k,\sigma}|! (n - |U_{\sigma}|)! = n! \cdot \binom{|U_{\sigma}|}{|A_{1,\sigma}| \cdots |A_{k,\sigma}|}^{-1}.$$
 (5.1)

We will now prove a lemma which states that $\{\mathscr{C}_{\sigma}\}_{\sigma \in [m]_{(k-1)}}$ forms a disjoint collection of a permutations. The general proof only works for $k \geq 4$, so we first consider k = 3.

Lemma 5.1.1. If $\sigma_1, \sigma_2 \in [m]_{(2)}$ are distinct, then $\mathscr{C}_{\sigma_1} \cap \mathscr{C}_{\sigma_2} = \emptyset$.

Proof. Seeking a contradiction, suppose there exists $\pi \in \mathscr{C}_{\sigma_1} \cap \mathscr{C}_{\sigma_2}$. After relabeling, it suffices to consider the following five cases.

(1)
$$\sigma_1 = \{1,3\}$$
 and $\sigma_2 = \{2,4\}$ (2) $\sigma_1 = \{1,3\}$ and $\sigma_2 = \{2,3\}$
(3) $\sigma_1 = \{1,2\}$ and $\sigma_2 = \{1,3\}$ (4) $\sigma_1 = \{1,2\}$ and $\sigma_2 = \{2,3\}$
(5) $\sigma_1 = \{1,2\}$ and $\sigma_2 = \{3,1\}$.

In case (1), without loss of generality, $\max\{\pi(x) : x \in A_{1,1}\} \le \max\{\pi(x) : x \in A_{1,2}\}$ and thus $\pi \in \mathscr{C}_{\sigma_2}$ yields

$$\max_{x \in A_{1,1}} \pi(x) \le \max_{x \in A_{1,2}} \pi(x) < \min_{y \in A_{2,4} \setminus A_{1,2}} \pi(y).$$

Then as $A_{1,1} \cap A_{2,4} \cap A_{3,2} \neq \emptyset$, there exists $w \in A_{1,1} \cap A_{2,4} \cap A_{3,2}$. It follows that $w \notin A_{1,2}$ since if $w \in A_{1,2}$, then $w \in A_{1,2} \cap A_{2,4} \cap A_{3,2} \neq \emptyset$; a contradiction. But this yields a contradiction as

$$\pi(w) \le \max_{x \in A_{1,1}} \pi(x) \le \max_{x \in A_{1,2}} \pi(x) < \min_{y \in A_{2,4} \setminus A_{1,2}} \pi(y) \le \pi(w).$$

In case (2), without loss of generality, $\max\{\pi(x) : x \in A_{1,1}\} \leq \max\{\pi(x) : x \in A_{1,2}\}$ and we recover a similar contradiction as case (1) by noting that there exists $w \in A_{1,1} \cap A_{2,3} \cap A_{3,2}$ with $w \notin A_{1,2}$.

In case (3) we may assume $\max\{\pi(x) : x \in A_{2,2} \setminus A_{1,1}\} \le \max\{\pi(x) : x \in A_{2,3} \setminus A_{1,1}\}$ and $\pi \in \mathscr{C}_{1,3}$ yields $\max\{\pi(x) : x \in A_{2,3} \setminus A_{1,1}\} < \min\{\pi(x) : x \in A_{3,1} \setminus (A_{1,1} \cup A_{2,3})\}.$ Thus

$$\max\{\pi(x) : x \in A_{2,2} \setminus A_{1,1}\} < \min\{\pi(x) : x \in A_{3,1} \setminus (A_{1,1} \cup A_{2,3})\}$$

and there exists $w \in A_{1,3} \cap A_{2,2} \cap A_{3,1}$ with $w \notin A_{1,1}$ and $w \notin A_{2,3}$. It follows that $\pi(w) < \pi(w)$, a contradiction.

In case (4), if $\max\{\pi(x) : x \in A_{1,1}\} \leq \max\{\pi(x) : x \in A_{1,2}\}$, then using $w \in A_{1,1} \cap A_{2,3} \cap A_{3,2}$ and noting $w \notin A_{1,2}$, we get a contradiction. Thus, we may assume otherwise and $\pi \in C_{1,2}$ gives

$$\max_{x \in A_{1,2}} \pi(x) < \max_{x \in A_{1,1}} \pi(x) < \min_{z \in A_{3,1} \setminus (A_{1,1} \cup A_{2,2})} \pi(z).$$

This is a contradiction as there exists $w \in A_{1,2} \cap A_{2,3} \cap A_{3,1}$ with $w \notin A_{1,1}$ and $w \notin A_{2,2}$.

In case (5), if $\max\{\pi(x) : x \in A_{1,1}\} \leq \max\{\pi(x) : x \in A_{1,3}\}$, then we may proceed as in the latter part of case (4) using $w \in A_{1,1} \cap A_{2,2} \cap A_{3,3}$ and $w \notin A_{2,1}$ and $w \notin A_{1,3}$ to get a contradiction. Otherwise, proceeding as in case (1) and noting there exists $w \in A_{1,3} \cap A_{2,2} \cap A_{3,1}$, but $w \notin A_{1,1}$ yields a contradiction. \Box

A similar argument yields the analog of Lemma 5.1.1 to the case where $k \ge 4$.

Lemma 5.1.2. Let $k \ge 4$. If $\sigma_1, \sigma_2 \in [m]_{(k-1)}$ are distinct, then $\mathscr{C}_{\sigma_1} \cap \mathscr{C}_{\sigma_2} = \emptyset$.

Proof. Since $\sigma_1, \sigma_2 \in [m]_{(k-1)}$ are distinct, there exists minimal $h \in [k-1]$ so that $\sigma_1(h) \neq \sigma_2(h)$. Seeking a contradiction, suppose there exists a $\pi \in \mathscr{C}_{\sigma_1} \cap \mathscr{C}_{\sigma_2}$. Without

loss of generality,

$$\max\{\pi(x) : x \in A_{h,\sigma_1}\} \le \max\{\pi(x) : x \in A_{h,\sigma_2}\} < \min\{\pi(z) : z \in A_{k,\sigma_2}\}.$$

Now, consider a bijection $\tau : [k-1] \setminus \{h\} \to [k-1] \setminus \{1\}$ which has no fixed points. As in Lemma 5.1.1, we want to show that there exists a $w \in A_{h,\sigma_1} \cap A_{k,\sigma_2}$ and consider two separate cases.

First, suppose that $\sigma_1(h) \notin \sigma_2([k-1])$. As $|\{\sigma_1(h), \sigma_2(1), \dots, \sigma_2(k-1)\}| = k$, there exists

$$w \in A_{h,\sigma_1(h)} \cap A_{k,\sigma_2(1)} \cap \bigcap_{l \in [k-1] \setminus \{h\}} A_{l,\sigma_2(\tau(l))}.$$
(5.2)

Next, suppose that $\sigma_1(h) = \sigma_2(x)$ for some x. We now claim that $x \neq 1$. If h = 1, then this is trivial. If h > 1, then $\sigma_1(1) = \sigma_2(1)$, so $\sigma_1(h) \neq \sigma_2(1)$ since $\sigma_1(h) \neq \sigma_1(1)$. For τ as above, there exists $y \in [k-1] \setminus \{h\}$ so that $\tau(y) = x$. Taking γ distinct from $\{\sigma_2(1), \ldots, \sigma_2(k-1)\} \setminus \{\sigma_2(x)\}, |\{\sigma_1(h), \gamma, \sigma_2(1), \ldots, \sigma_2(k-1)\} \setminus \{\sigma_2(x)\}| = k$ and hence there exists

$$w \in A_{h,\sigma_1(h)} \cap A_{k,\sigma_2(1)} \cap A_{y,\gamma} \cap \bigcap_{l \in [k-1] \setminus \{y,h\}} A_{l,\sigma_2(\tau(l))}.$$
(5.3)

By construction, $w \in A_{h,\sigma_1(h)} \cap A_{k,\sigma_2(1)}$. Suppose there exists a $t \in [k-1] \setminus \{h\}$ so that $w \in A_{t,\sigma_2(t)}$. As τ has no fixed points, replacing the set in the k-wise intersection corresponding to \mathcal{A}_t with $A_{t,\sigma_2(t)}$ in either (5.2) or (5.3), w is an element of this new k-wise intersection with (k-1) distinct indices; a contradiction. If $w \in A_{h,\sigma_2(h)}$, then we may similarly replace $A_{h,\sigma_1(h)}$ with $A_{h,\sigma_2(h)}$ in the k-wise intersection in either (5.2) or (5.3) to get a contradiction. Thus, $w \notin A_{1,\sigma_2(1)} \cup \cdots \cup A_{k-1,\sigma_2(k-1)}$ and hence $w \in A_{h,\sigma_1} \cap A_{k,\sigma_2}$ so that $\pi(w) < \pi(w)$; a contradiction.

Using Equation (5.1), Lemma 5.1.1, and Lemma 5.1.2, we are now able to prove Theorem 1.5.2 in the case where t = k. There are n! total permutations, and Lemma 5.1.1 and Lemma 5.1.2 yield that each of which appears in at most one of the sets \mathscr{C}_{σ} for $\sigma \in [m]_{(k-1)}$. Hence, using $|\mathscr{C}_{\sigma}|$ in Equation (5.1),

$$\sum_{\sigma \in [m]_{(k-1)}} |\mathscr{C}_{\sigma}| = \sum_{\sigma \in [m]_{(k-1)}} n! \cdot \binom{|A_{1,\sigma} \cup \dots \cup A_{k,\sigma}|}{|A_{1,\sigma}| \cdots |A_{k,\sigma}|}^{-1} \le n!$$

and thus the result follows by dividing through by n!.

Sharpness of Theorem 1.5.2:

We give a simple construction establishing the sharpness of Theorem 1.5.2 for $k \ge t = 2$. Let $n \ge 4k$ and using addition modulo n, define $A_{1,i} = \{i\}^c$, $A_{j,i} = \{i - (j-1), i + (j-1)\}^c$ for $j \in [2, k-1]$, and $A_{k,i} = \{i - k + 2, i - k + 3, \dots, i + k - 2\}$. Letting $\mathcal{A}_j = \{A_{j,i}\}_{i \in [n]}$ for all $j \in [k]$, we will show $(\mathcal{A}_1, \dots, \mathcal{A}_k)$ is a Bollobás (k, 2)-tuple. Since $|A_{1,i}| = n - 1$ and $|A_{2,i} \cap \dots \cap A_{k,i}| = 1$, Theorem 1.5.2 with t = 2 and surjection $\phi : [k] \rightarrow [2]$ with $\phi(1) = 1$ and $\phi(i) = 2$ for $i \neq 1$ gives

$$1 \geq \sum_{i=1}^{n} \left(\frac{|A_{1,i}| + |A_{2,i} \cap \dots \cap A_{k,i}|}{|A_{1,i}|} \right)^{-1} = \sum_{i=1}^{n} \frac{1}{n} = 1.$$

By construction, for all $i \in [n]$, $A_{1,i} \cap A_{2,i} \cap \cdots \cap A_{k,i} = \emptyset$. It thus suffices to show these are the only empty k-wise intersections. To this end, for $\mathbf{i} = (i_1, \dots, i_{k-1})$, define

$$A(\mathbf{i}) := A_{1,i_1} \cap \dots \cap A_{k-1,i_{k-1}}$$

Lemma 5.1.3. Let $\mathbf{i} = (i_1, \ldots, i_{k-1})$. If $A(\mathbf{i})^c = A_{k,i_k}$, then $i_1 = \cdots = i_k$.

Proof. We proceed by induction on k where the result is trivial when k = 2. In the case where k > 2, $i_{k-1} - k + 2 = i_k + x$ for some x such that $-(k-2) \le x \le (k-2)$ and thus $i_{k-1} + (k-2) = i_{k-1} - (k-2) + (2k-4) = i_k + x + (2k-4).$

Next, there is a y such that $-(k-2) \le y \le (k-2)$ with $i_{k-1} + (k-2) = i_k + y$, and since $n \ge 4k$, x+2k-4 = y with equality over \mathbb{Z} and moreover $i_{k-1} + (k-2) = i_k + (k-2)$ over \mathbb{Z} and hence $i_k = i_{k-1}$. Removing these elements from each set, the result then follows by induction.

If
$$A_{1,i_1} \cap \dots \cap A_{k,i_k} = \emptyset$$
, then as $A(i) = A_{1,i_1} \cap A_{2,i_2} \cap \dots \cap A_{k-1,i_{k-1}}$,
 $\emptyset = A_{1,i_1} \cap A_{2,i_2} \cap \dots \cap A_{k-1,i_{k-1}} \cap A_{k,i_k} = A(i) \cap A_{k,i_k}$.

The result follows by noting $|A(\mathbf{i})| \ge n - (2k - 3)$, $|A_{k,i_k}| = 2k - 3$, and using Lemma 5.1.3.

An Explicit Construction:

Let $k \geq 3$. An explicit construction of a Bollobás (k, 2)-tuple $(\mathcal{A}_1, \mathcal{A}_2, \ldots, \mathcal{A}_k)$ where $|\mathcal{A}_i| = 2^n$ and each \mathcal{A}_i consists of subsets of X for |X| = kn may be described as follows. Let $I_j := \{x_{j,1}, x_{j,2}, \ldots, x_{j,k}\}$ and consider $X = I_1 \sqcup \cdots \sqcup I_n$. Now, for each $f : [n] \rightarrow [2]$ and $j \in [k]$, define

$$A_{j,f} := \{x_{1,f(1)+j-1}, \ldots, x_{n,f(n)+j-1}\}^c$$

where we work modulo k within the subscripts of I_j . It is straightforward to check that $(\mathcal{A}_1, \mathcal{A}_2, \ldots, \mathcal{A}_k)$ is a Bollobás (k, 2)-tuple. This establishes the lower bound on $\beta_{k,2}(n)$ in Equation (1.6) and hence the upper bound on $f_{k,2}(n)$ in Equation (1.11).

5.2 Proof of Theorem 1.5.4

5.2.1 Upper bound on $f_{k,t}(n)$

We wish to find a covering of $H_{k,t}(n)$ with complete k-partite k-graphs and assume the parts of $H_{k,t}(n)$ are X_1, X_2, \ldots, X_k . For each subset T of [k] of size t, consider the uniformly random coloring $\chi_T : [n] \to T$. Given such a χ_T , let $Y_i \subseteq X_i$ be the vertices of color i for $i \in T$; that is $Y_i := \{x_{ij} : \chi(j) = i\}$ and $Y_i = X_i$ for $i \notin T$. Denote by $H(T, \chi)$ the (random) complete k-partite hypergraph with parts Y_1, Y_2, \ldots, Y_k , and note that $H(T, \chi) \subseteq H_{k,t}(n)$. We place each $H(T, \chi)$ a total of N times independently and randomly where

$$N = \left\lfloor \frac{(t+1)t^t \log_2 n}{(k-t+1)\log_2 e} \right\rfloor$$

and produce $\binom{k}{t}N$ random subgraphs $H(T, \chi)$. For a set partition π of [k], let $|\pi|$ denote the number of parts in the partition and index the parts by $[|\pi|]$. Given a set partition $\pi = (P_1, P_2, \ldots, P_s)$, let

$$f(\pi, t) = \sum_{T \in [s]^{(t)}} \prod_{i \in T} |P_i|.$$

If U is the number of edges of $H_{k,t}(n)$ not in any of these subgraphs, then

$$\mathbb{E}(U) \le \sum_{|\pi| \ge t} n^{|\pi|} (1 - t^{-t})^{Nf(\pi,t)} = \sum_{t \le s \le k} n^s \sum_{|\pi| = s} (1 - t^{-t})^{Nf(\pi,t)}.$$
 (5.4)

For sufficiently large n, we claim that $\mathbb{E}(U) < 1$, which implies there exists a covering of $H_{k,t}(n)$ with at most $\binom{k}{t}N$ complete k-partite k-graphs, as required. The following technical lemma states that f is a decreasing function in the set partition lattice,

and that $f(\pi, t)$ increases when we merge all but one element of a smaller part of π with a larger part of π :

Lemma 5.2.1. Let $k \ge s \ge t \ge 2$, and let $\pi = (P_1, P_2, \ldots, P_s)$ be a partition of [k].

- (i) If π' is a refinement of π with $|\pi'| = s + 1$, then $f(\pi, t) \leq f(\pi', t)$.
- (ii) If |P₁| ≥ |P₂| ≥ 2 and a ∈ P₂, and π' is the partition (P'₁, P'₂,..., P'_s) of [k] with P'₁ = P₁ ∪ P₂ \ {a} and P'₂ = {a} and with P'_i = P_i for 3 ≤ i ≤ s, then f(π', t) ≤ f(π, t).

Proof. We will only prove (i) as the proof of (ii) is similar. Let $\pi = P_1|P_2|\cdots|P_s$ and without loss of generality, $\pi' = P_x|P_y|P_2|\cdots|P_s$. Setting $\mathcal{T}(\overline{1}) = \{T \in [s]^{(t)} : 1 \notin T\}$ and $\mathcal{T}'(\overline{x}, \overline{y}) = \{T \in \{x, y, 2, \dots, s\}^{(t)} : x, y \notin T\}$, it follows that

$$\sum_{T \in \mathcal{T}(\bar{1})} \prod_{i \in T} |P_i| = \sum_{T \in \mathcal{T}'(\bar{x}, \bar{y})} \prod_{i \in T} |P_i|.$$

Now, letting $\mathcal{T}(1) = \{T \in [s]^{(t)} : 1 \in T\}$ and $\mathcal{T}'(x, \overline{y}) = \{T \in \{x, y, 2, \dots, s\}^{(t)} : x \in T, y \notin T\}$ and $\mathcal{T}'(\overline{x}, y) = \{T \in \{x, y, 2, \dots, s\}^{(t)} : x \notin T, y \in T\}$, we see that

$$\sum_{T \in \mathcal{T}(1)} \prod_{i \in T} |P_i| = \sum_{T \in \mathcal{T}'(\overline{x}, y)} \prod_{i \in T} |P_i| + \sum_{T \in \mathcal{T}'(x, \overline{y})} \prod_{i \in T} |P_i|$$

since $|P_1| = |P_x| + |P_y|$. Thus letting $\mathcal{T}'(x, y) = \{T \in \{x, y, 2, \dots, s\}^{(t)} : x \in T, y \in T\},\$

$$f(\pi',t) - f(\pi,t) = \sum_{T \in \mathcal{T}'(x,y)} \prod_{i \in T} |P'_i|$$

and in particular $f(\pi, t) \leq f(\pi', t)$.

74

By Lemma 5.2.1, a set partition of [k] into s parts which minimizes $f(\pi, t)$ consists of one part of size k - s + 1 and s - 1 singleton parts and hence

$$\min\{f(\pi,t): |\pi|=s\} = (k-s+1)\binom{s-1}{t-1} + \binom{s-1}{t}.$$
(5.5)

In what follows, we denote a set partition of [k] into s parts which minimizes $f(\pi, t)$ by π_s . For n large enough, and all s where $t \leq s \leq k$, we will show

$$\frac{\sum_{|\pi|=t} (1-t^{-t})^{Nf(\pi,t)}}{\sum_{|\pi|=s} (1-t^{-t})^{Nf(\pi,t)}} \ge n^{s-t}.$$

Replacing the numerator with its largest term and each term in denominator with its largest term,

$$\frac{\sum_{|\pi|=t} (1-t^{-t})^{Nf(\pi,t)}}{\sum_{|\pi|=s} (1-t^{-t})^{Nf(\pi,t)}} \ge \frac{(1-t^{-t})^{Nf(\pi_t,t)}}{S(k,s)(1-t^{-t})^{Nf(\pi_s,t)}} = \frac{1}{S(k,s)} (1-t^{-t})^{N(f(\pi_t,t)-f(\pi_s,t))}$$

where S(k, s) is the Stirling number of the second kind. Taking $n \ge S(k, s)$, we will now show the following:

Claim 5.2.2.

$$\frac{1}{S(k,s)} (1 - t^{-t})^{N(f(\pi_t,t) - f(\pi_s,t))} \ge n^{s-t}.$$
(5.6)

Proof. First, we recall that

$$N = \left\lfloor \frac{(t+1)t^t \log_2 n}{(k-t+1)\log_2 e} \right\rfloor \text{ and } f(\pi_s, t) = (k-s+1)\binom{s-1}{t-1} + \binom{s-1}{t}.$$

As a result, when $t \leq s < k$, a calculation yields that

$$f(\pi_{s+1}, t) - f(\pi_s, t) = (k-s) \binom{s-1}{t-2}.$$
(5.7)

Letting $n \geq S(k,t)$, after taking $\log_2(\cdot)$ on both sides of (5.6), it suffices to prove that

$$N \cdot \frac{f(\pi_s, t) - f(\pi_t, t)}{t^t} \left(-t^t \log_2(1 - t^{-t}) \right) \ge (s - t + 1) \log_2(n).$$
(5.8)

Using the fact that $(1 - t^{-t})^{t^t} \le e^{-1}$ and our choice of N, it suffices to show that

$$f(\pi_s, t) - f(\pi_t, t) \ge \frac{(s - t + 1)(k - t + 1)}{t + 1}.$$
(5.9)

The inequality in (5.9) holds for all $k \ge s > t \ge 3$ by using (5.7).

Therefore, the index s = t maximizes the right hand side of Equation (5.4), and hence

$$\mathbb{E}[U] \le (k-t+1)(n^t) \sum_{|\pi|=t} (1-t^{-t})^{Nf(\pi,t)} < (k-t+1)n^t S(k,t)(1-t^{-t})^{N(k-t+1)} < 1$$

for our choice of N provided $n \ge kS(k, t)$. Thus,

$$f_{k,t}(n) \le \binom{k}{t} \frac{(t+1)t^t \log_2 n}{(k-t+1)\log_2 e} = \frac{(t+1)t^{t-1}}{\log_2 e} \binom{k}{t-1} \log_2 n.$$

5.2.2 Lower bound on $f_{k,2}(n)$

In this section, we show

$$f_{k,2}(n) \ge \min\{m : \binom{m}{\lceil m/k \rceil} \ge n\}.$$
(5.10)

Let $\{H_1, H_2, \ldots, H_m\}$ be a covering of $H_{k,2}(n)$ with $m = f_{k,2}(n)$ complete k-partite k-graphs. We recall $H_{k,2}(n) = K_{n,n,\dots,n} \setminus M$, where M is a perfect matching of $K_{n,n,\dots,n}$. For $i \in [k]$ and $j \in [n]$, define $A_{i,j} = \{H_r : x_{ij} \in V(H_r)\}$ and $\mathcal{A}_i = \{A_{i,j} : 1 \leq j \leq n\}$. As in (1.9), $(\mathcal{A}_1, \mathcal{A}_2, \dots, \mathcal{A}_k)$ is a Bollobás (k, 2)-tuple of size n. For convenience, for each $i \in [k]$, let $\phi_i : [k] \to [2]$ be so that $\phi_i^{-1}(1) = \{i\}$. Taking the sum of inequality from Theorem 1.5.2 with t = 2 over all $i \in [k]$,

$$\sum_{i=1}^{k} \sum_{j=1}^{n} \left(\frac{|A_{1,j}(\phi_i) \cup A_{2,j}(\phi_i)|}{|A_{1,j}(\phi_i)|} \right)^{-1} \le k.$$
(5.11)

We use this inequality to give a lower bound on $f_{k,2}(n) = m$. First we observe

$$\sum_{r=1}^{m} |V(H_r)| = \sum_{j=1}^{n} \sum_{i=1}^{k} |A_{i,j}| = \sum_{j=1}^{n} \sum_{i=1}^{k} |A_{1,j}(\phi_i)|.$$
(5.12)

Let ∂H denote the set of (k-1)-tuples of vertices contained in some edge of a hypergraph H. Then

$$\sum_{r=1}^{m} |\partial H_r \cap \partial M| = \sum_{j=1}^{n} \sum_{i=1}^{k} |A_{2,j}(\phi_i)|.$$
(5.13)

Putting the above identities together,

$$\sum_{r=1}^{m} |V(H_r)| + \sum_{r=1}^{m} |\partial H_r \cap \partial M| = \sum_{j=1}^{n} \sum_{i=1}^{k} (|A_{1,j}(\phi_i)| + |A_{2,j}(\phi_i)|).$$
(5.14)

We note $|\partial H_r \cap \partial M| \leq |V(H_r)|/(k-1)$, and therefore

$$\sum_{r=1}^{m} |\partial H_r \cap \partial M| \le \frac{1}{k-1} \sum_{r=1}^{m} |V(H_r)|.$$
 (5.15)

It follows that

$$\sum_{j=1}^{n} \sum_{i=1}^{k} (|A_{1,j}(\phi_i)| + |A_{2,j}(\phi_i)|) \le \frac{k}{k-1} \sum_{r=1}^{m} |V(H_r)|.$$
(5.16)

Subject to the linear inequalities (5.12) and (5.16), the left side of (5.11) is minimized when $kn|A_{1,j}(\phi_i)| = \sum_{r=1}^m |V(H_r)|$ and $kn(|A_{1,j}(\phi_i)| + |A_{2,j}(\phi_i)|) = (k-1)|A_{1,j}(\phi_i)|$. Since $|V(H_r)| \le (k-1)n$ for all $r \in [m]$, (5.11) implies $\binom{m}{\lceil m/k \rceil} \ge n$, which gives (5.10).

5.2.3 Lower bound on $f_{k,k}(n)$

Let $\mathcal{H} = \{H_1, H_2, \dots, H_m\}$ be a minimal covering of $H_{k,k}(n)$ with complete kpartite k-graphs, so $m = f(H_{k,k}(n))$. Given a k-partite k-graph H, consider its 2-shadow $\partial_2(H) = \{R \subseteq V(H) : |R| = k - 2, R \subseteq e \text{ for some } e \in H\}$. Let $\partial_2(\mathcal{H}) = \bigcup_{i=1}^m \partial_2(H_i)$.

Given $R \in \partial_2(\mathcal{H})$ and $H_i \in \mathcal{H}$, let $H_i(R) := \{e \in \binom{V(H_i)}{2} : e \cup R \in H_i\}$ be the possibly empty link graph of the edge R in the hypergraph H_i and let $V(H_i(R))$ be the set of vertices in the link graph. Observe that double counting yields

$$\sum_{R\in\partial_2(\mathcal{H})} \left(\sum_{i=1}^m |V(H_i(R))|\right) = \sum_{i=1}^m \left(\sum_{R\in\partial_2(H_i)} |V(H_i(R))|\right).$$
(5.17)

An optimization argument yields $|\partial_2(H_i)|$ is maximized when the parts of H_i are of equal or nearly equal maximal size. Since $|V(H_i(R))| \leq 2(n-k+2)$, the right hand side of Equation (5.17) is bounded above by

$$\sum_{i=1}^{m} \left(\sum_{R \in \partial_2(H_i)} |V(H_i(R))| \right) \le m \cdot {\binom{k}{2}} \cdot \left(\frac{n}{k}\right)^{k-2} \cdot 2(n-k+2).$$
(5.18)

For a lower bound on the left hand side of Equation (5.17), fix $R \in \partial_2(\mathcal{H})$ and without loss of generality suppose that $R = \{x_{1,1}, \ldots, x_{k-2,k-2}\}$. Let Y = [k-1,n]. Let $K_{Y,Y}$ be the complete bipartite graph with two distinct copies of Y and $\mathcal{M} = \{(x_{k-1,i}, x_{k,i} : i \in Y\}$ be a perfect matching in $K_{Y,Y}$. Then, $\{H_1(R), \ldots, H_m(R)\}$ forms a biclique cover of $K_{Y,Y} \setminus \mathcal{M}$. Applying the convexity result of Tarjan [Tar75, Lemma 5],

$$\sum_{i=1}^{m} |V(H_i(R))| \ge (n-k+2)\log_2(n-k+2)$$

Noting that $|\partial_2(\mathcal{H})| = {k \choose 2} (n)_{(k-2)}$, the left hand side of Equation (5.17) is bounded below by

$$\sum_{R \in \partial_2(\mathcal{H})} \left(\sum_{i=1}^m |V(H_i(R))| \right) \ge {\binom{k}{2}} (n)_{(k-2)} (n-k+2) \log_2(n-k+2).$$
(5.19)

Comparing the bounds from Equation (5.18) and Equation (5.19),

$$m \ge \frac{(n)_{(k-2)} \log_2(n-k+2)}{2\left(\frac{n}{k}\right)^{k-2}} \ge \frac{k^{k-2}}{2} \log_2 n$$

provided that n is large enough.

For $t \geq 3$ and t < k, the lower bound on $f_{k,t}(n)$ in Theorem 1.5.4 is obtained from the lower bounds on $f_{t-1,t-1}(n-1)$ as follows: Let $\mathcal{H} = \{H_1, H_2, \ldots, H_m\}$ be a minimal covering of $H_{k,t}(n)$ with complete k-partite k-graphs, so $m = f(H_{k,t}(n))$. Given $T \in {\binom{[k]}{k-t+1}}$, define $H_T \subseteq H_{k,t}(n)$ by

$$H_T := \{ \{ x_{1,i_1}, \dots, x_{k,i_k} \} \in H_{k,t}(n) : i_j = 1 \ \forall \ j \in T \}.$$

It follows that at least $f_{t-1,t-1}(n-1)$ of the complete k-partite k-graphs in \mathcal{H} are needed to cover H_T . Moreover, for distinct $T, T' \in {\binom{[k]}{k-t+1}}$, the corresponding complete k-partite k-graphs from \mathcal{H} are necessarily pairwise disjoint and hence

$$f_{k,t}(n) \ge \binom{k}{k-t+1} f_{t-1,t-1}(n-1) \ge \binom{k}{t-1} \frac{(t-1)^{t-3}}{2} \log_2 n$$

provided that n is large enough.

This chapter contains material from: J. O'Neill, J. Verstraëte, "A generalization of the Bollobás set pairs inequality", The Electronic Journal of Combinatorics, 28(3), 2021. The dissertation author was one of the primary investigators and authors of this paper.

Chapter 6

Intersection saturated set systems

6.1 Clique partitions

In this section, we explore clique partitions. Given integers $2 \leq t < k$, a *clique* partition of $\binom{[k]}{t}$ is a collection of proper subsets $X_1, \ldots, X_m \subset [k]$ such that $\bigcup_i \binom{X_i}{t}$ is a partition of $\binom{[k]}{t}$. Let cp(k, t) denote the minimum size of a clique partition of $\binom{[k]}{t}$.

When t = 2, the classical theorem of de Bruijn and Erdős [dBE48] states that cp(k,t) = k. In the study of (n, k, t)-saturated set systems, we are interested in the minimum clique size in any clique partition and use the following result of Alon, Mellinger, Mubayi and Verstraëte [AMMV12]:

Theorem 6.1.1 (Alon et al.[AMMV12]). Let $t \ge 2$ and $k \in \mathbb{N}$. Define $q \in \mathbb{R}$ by $k = q^2 + q + t - 1$. Then,

$$cp(k,t) \ge \frac{\binom{k}{t}}{\binom{q+t-1}{t}}.$$

In particular, in any such clique partition there exists X_i with $|X_i| \leq \lfloor q+t-1 \rfloor$.

6.2 Rödl Nibble

In this section, we mention a generalization of the Rödl Nibble by Frankl and Rödl [FR85]. For a hypergraph H, let $\nu(H)$ be the maximum number of pairwise disjoint edges in H and t(H) to be the minimum t for which t edges in H cover all of V(H).

Theorem 6.2.1 (Frankl-Rödl [FR85]). Suppose $\epsilon > 0$ is arbitrary, H is a d-uniform hypergraph on X, |X| = n, and a > 3 is a real number. Then there exists positive real

 δ such that if for some D one has $(1 - \delta)D < d_H(v) < (1 + \delta)D$ for all $x \in X$ and $d_H(\{x, y\}) < D/(\log(n)^a)$ holds for all distinct $x, y \in X$. Then for all $n > n_0(\delta)$,

$$t(H) \le \frac{n(1+\epsilon)}{d}$$
 and $\nu(H) \ge \frac{n}{d}(1-d\epsilon).$

Lemma 6.2.2. Let S = S(n, 7, 5). Then there exists $S_1 \subset \partial_6(S(n, 7, 5))$ for which each 4-tuple $T \in \binom{n}{4}$ is contained in at most one edge in S_1 and $|S_1| = (1 - o(1))\binom{n}{4} / \binom{6}{4}$.

Proof. Let $V(H) = {\binom{[n]}{4}}$ with $d = {\binom{6}{4}}$ and $E(H) = \{\{A \in {\binom{[n]}{4}} : A \subset B\} : B \in \partial_6(S)\}$. For each $A \in {\binom{[n]}{4}}$, it follows that the link of A in S is a 3-uniform perfect matching for which each edge contributes three to the degree of A in H and hence $d_H(A) = 3 \cdot \frac{n-4}{3} = n-4$. Moreover, for $A_1, A_2 \in {\binom{[n]}{4}}$, it follows that $d_H(\{A_1, A_2\}) \leq 2$ and hence we satisfy the conditions of Theorem 6.2.1. As such $\nu(H) \geq (1-o(1))\frac{|V|}{d} = (1-o(1)){\binom{n}{4}}/{\binom{6}{4}}$ which corresponds to such an S_1 as desired.

6.3 Large Sets of Designs

Given $n \ge k \ge t$, a large set of designs is a partition of $\binom{[n]}{k}$ into pairwise edge disjoint Steiner systems S(n, k, t). It is not difficult to see that we necessarily need $\binom{k-i}{t-i} \binom{n-i}{t-i}$ and $\binom{k}{t}^{-1} \binom{n}{t} \binom{n}{k}$ in order for a large set of designs to exist. Completing the work of Lu [JX83], Tierlinck [Tie91] proved that large sets of designs exists when t = 2and k = 3. Keevash [Kee18] then proved that large sets of designs exist for $t \le k$ and nsufficiently large: **Theorem 6.3.1** (Keevash [Kee18]). Suppose $k \ge t$ is fixed and $n \ge n_0(k, t)$ is large and so that it satisfies the above divisibility conditions. Then there exists a large set of designs.

6.4 Proof of Theorem 1.6.3

Let $T_{n,k-1}$ denote the largest such complete (k-1)-partite graph on [n]. We will use the following stability theorem from Füredi $[F\ddot{1}5]$:

Theorem 6.4.1 (Füredi [FÏ5]). Suppose H is an n-vertex K_k -free graph with $|E(H)| \ge |T_{n,k-1}| - s$. Then there exists a (k-1)-chromatic G with V(G) = V(H), $E(G) \subseteq E(H)$, $|E(G)| \ge |E(H)| - s$.

Let k > t = 2 and suppose $\mathcal{F} \subset {\binom{[n]}{k}}$ is (n, k, 2)-saturated with

$$\binom{k}{2} \cdot |\mathcal{F}| \le (1+\epsilon) \cdot \frac{1}{k-1} \binom{n}{2}$$

where $\epsilon = \epsilon(k)$ is some constant to be chosen later. Observe that $\partial_2(\mathcal{F})^c$ is K_k -free, and

$$\left|\partial_2(\mathcal{F})^c\right| \ge \left(1 - \frac{1}{k-1}\right)\binom{n}{2} - \frac{\epsilon}{k-1}\binom{n}{2} \ge \left|T_{n,k-1}\right| - \frac{\epsilon}{k-1}\binom{n}{2}$$

since $|\partial_2(\mathcal{F})| \leq {\binom{k}{2}} \cdot |\mathcal{F}|$. As such, applying Theorem 6.4.1 to $\partial_2(\mathcal{F})^c$ with $s = \epsilon {\binom{n}{2}}/(k-1)$ gives pairwise disjoint $X_1, \ldots, X_{k-1} \subset [n]$ and $G \subseteq K(X_1, \ldots, X_{k-1})$ with $E(G) \subseteq \partial_2(\mathcal{F})^c$ and so that $|E(G)| \geq |\partial_2(\mathcal{F})^c| - \epsilon {\binom{n}{2}}/(k-1)$.

Definition. An edge $A \in \mathcal{F}$ is called good if $A \subset X_i$ for some $i \in [k-1]$ and $d_{\mathcal{F}}(e) = 1$ for all $e \in \binom{A}{2}$. Otherwise, we refer to A as a bad edge. Set $\mathcal{B} \subset \mathcal{F}$ as the collection of bad edges.

Seeking to show there are few bad edges in \mathcal{F} , we further consider $\mathcal{B}_1 = \{A \in \mathcal{B} :$

 $\exists i \neq j \text{ with } A \cap X_i \neq \emptyset, A \cap X_j \neq \emptyset \}$ and $\mathcal{B}_2 = \{A \in \mathcal{B} : \exists i \text{ with } A \subset X_i \}.$

Claim. $|\mathcal{B}| = |\mathcal{B}_1| + |\mathcal{B}_2| \le \frac{4\epsilon}{3(k-1)} {n \choose 2}.$

Proof. Let $\mathcal{F}_i = \mathcal{F}|_{X_i}$ be the trace of \mathcal{F} on X_i so that $|\mathcal{B}_2| = |\mathcal{F}_1 \cap \mathcal{B}_2| + \dots + |\mathcal{F}_{k-1} \cap \mathcal{B}_2|$.

$$\binom{k}{2} \cdot \left(|\mathcal{F}_1| + \dots + |\mathcal{F}_{k-1}| + |\mathcal{B}_1|\right) = \binom{k}{2} |\mathcal{F}| \le (1+\epsilon) \cdot \frac{1}{k-1} \binom{n}{2} \tag{6.1}$$

and for each $i \in [k-1]$ it follows that

$$\binom{k}{2}|\mathcal{F}_i| = \sum_{e \in \partial_2(\mathcal{F}_i)} d_{\mathcal{F}_i}(e) = |\partial_2(\mathcal{F}_i)| + \sum_{e \in \partial_2(\mathcal{F}_i)} (d_{\mathcal{F}_i}(e) - 1).$$
(6.2)

Since \mathcal{F} is (n, k, 2)-saturated, for each $A \in \mathcal{B}_2 \cap \mathcal{F}_i$ there exists $A \neq B \in \mathcal{B}_2 \cap \mathcal{F}_i$

which contribute at least three to the sum on the right hand side of (6.2) as $|A \cap B| \ge 3$. Therefore,

$$\sum_{e \in \partial_2(\mathcal{F}_i)} (d_{\mathcal{F}_i}(e) - 1) \ge \frac{3}{2} \cdot |\mathcal{F}_i \cap \mathcal{B}_2|.$$
(6.3)

Note that each edge $A \in \mathcal{B}_1$ intersects at most $\binom{k-1}{2}$ edges in $K(X_1, \ldots, X_{k-1})^c$ and moreover that $|\partial_2(\mathcal{F})^c \cap K(X_1, \ldots, X_{k-1})^c| \leq \epsilon \binom{n}{2}/(k-1)$ since each edge in the intersection needs to be removed from $\partial_2(\mathcal{F})^c$ in order to create G. It thus follows that

$$|\partial_2(\mathcal{F}_1)| + \dots + |\partial_2(\mathcal{F}_{k-1})| + \binom{k-1}{2} |\mathcal{B}_1| + \frac{\epsilon}{k-1} \binom{n}{2} \ge |K(X_1, \dots, X_{k-1})^c| \ge |T_{n,k-1}^c|.$$
(6.4)

As a result, using (6.2), (6.3) and (6.4), we recover

$$\binom{k}{2}(|\mathcal{F}_1| + \dots + |\mathcal{F}_{k-1}| + |\mathcal{B}_1|) \ge \frac{1}{k-1}\binom{n}{2} - \frac{\epsilon}{k-1}\binom{n}{2} + (k-1)|\mathcal{B}_1| + \frac{3}{2}|\mathcal{B}_2|.$$

Hence, using (6.1) it follows that

$$(1+\epsilon) \cdot \frac{1}{k-1} \binom{n}{2} \ge \frac{1}{k-1} \binom{n}{2} - \frac{\epsilon}{k-1} \binom{n}{2} + (k-1)|\mathcal{B}_1| + \frac{3}{2}|\mathcal{B}_2|.$$

We then recover the desired bound by noting $k \ge 3$ and rearranging to get

$$\frac{2\epsilon}{k-1}\binom{n}{2} \ge (k-1)|\mathcal{B}_1| + \frac{3}{2}|\mathcal{B}_2| \ge \frac{3}{2}\left(|\mathcal{B}_1| + |\mathcal{B}_2|\right).$$

Lemma 6.4.2. Let $A \in \mathcal{F}$ be a good edge and $A' \subset A \subset X_i$ with |A'| = k - 1 and $v \in [n] \setminus X_i$. Then there exists $u \in A'$ with $\{u, v\} \in \partial_2(\mathcal{F})$.

Proof. It follows that $A' \cup \{v\} \notin \mathcal{F}$ as A is a good edge. As such, there must be a set $B \in \mathcal{F}$ which intersects $A' \cup \{v\}$ in exactly two places. Using that A is good, we recover the desired claim.

Let $e(X_i, X_j)$ be the number of edges in $\partial_2(\mathcal{F})$ with one vertex in X_i and the other vertex in X_j . Using Lemma 6.4.2, for each good edge $A \in \mathcal{F}$ and vertex v from another block, there are at least two edges between v and a vertex from A in $\partial_2(\mathcal{F})$. Note that $\mathcal{F}_i \cap (\mathcal{F} \setminus \mathcal{B})$ forms a partial Steiner system and hence vertices have degree at most $|X_i| - 1$. An optimization argument then gives

$$\sum_{\{i,j\}\in \binom{[k]}{2}} e(X_i, X_j) \ge \frac{(|\mathcal{F}| - |\mathcal{B}|) \cdot 2\frac{(k-2)n}{r-1}}{\frac{n}{k-1}} \ge 2(k-2)(|\mathcal{F}| - |\mathcal{B}|).$$

We reach a contradiction if $2(k-2)(|\mathcal{F}|-|\mathcal{B}|) > \epsilon {n \choose 2}/(k-1)$ which is equivalent

$$2(k-2)\left(\frac{2(1+\epsilon)}{k(k-1)} - \frac{4\epsilon}{3}\right) > \epsilon.$$

It then follows that taking $\epsilon=k^{-2}$ gives a contradiction.

6.5 Proof of Theorem 1.6.4

to

In this section, we show that at least one of the following set families is (n, k, t)saturated.

Construction 1. Let $n \ge k > t \ge 1$ be integers and consider $q \in \mathbb{R}$ where $k = q^2 + q + t - 1$. For $0 \le r \le \lfloor q - 1 \rfloor$, consider $\partial_k(\mathcal{S}(n, k + r, t))$.

Claim. If $\partial_k(\mathcal{S}(n, k+r, t))$ is not (n, k, t)-saturated, then there exists a clique partition of $\binom{[k]}{t}$ into cliques of size at least t + r + 1.

Proof. Suppose $\partial_k(\mathcal{S}(n, k+r, t))$ is not (n, k, t)-saturated, then there exists an $R \in {\binom{[n]}{k}}$ which may be added without creating an intersection of size t. However, since each $T \in {\binom{[n]}{t}}$ is covered by an edge of $\mathcal{S}(n, k+r, t)$, it follows that the trace $\partial_k(\mathcal{S}(n, k+r, t))|_R$ is a clique partition of ${\binom{R}{t}}$. Suppose there exists a set X_i in the clique partition of ${\binom{R}{t}}$ with $|X_i| \leq t+r$. This means that there exists $A \in \mathcal{S}(n, k+r, t)$ with $|A \cap R| \leq r+t$. Then, by appropriately choosing a subset $A'' \subset A$ of size k, we recover an intersection of size t; a contradiction.

By Theorem 6.1.1, each clique partition of $\binom{[k]}{t}$ into proper subcliques has a clique of size at most $t + \lfloor q - 1 \rfloor$ and hence the above claim gives that at least one of the set families in Construction 1 is (n, k, t)-saturated which then implies Theorem 1.6.4 by taking the largest such family.

6.6 Proof of Theorem 1.6.5

In this section, we construct a (n, 5, 4)-saturated set system which gives $w(n, 5, 4) \le 5/8$:

Construction 2. Let $n = 0 \pmod{12}$ and consider pairwise disjoint X and Y with |X| = n - 3 and |Y| = n and further let $X = S_X \sqcup X'$ and $Y = S_Y \sqcup Y'$ with $|S_X| = 9$ and $|S_Y| = 15$. For convenience, let $S = S_X \sqcup S_Y$. By Theorem 6.3.1, we can decompose $\binom{X'}{4}$ and $\binom{Y}{4}$ as follows:

$$\binom{X'}{4} = \bigsqcup_{i \in [n-15]} \mathcal{S}_i(X', 4, 3) \quad and \quad \binom{Y}{4} = \bigsqcup_{i \in [n-3]} \mathcal{S}_i(Y, 4, 3)$$

Letting $X = \{x_1, \ldots, x_{n-3}\}$ and $Y' = \{y_1, \ldots, y_{n-15}\}$, we now define $\mathcal{F} = \mathcal{F}_1 \sqcup \mathcal{F}_2$ where

$$\mathcal{F}_1 = \{\{y_i\} \cup T : T \in \mathcal{S}_i(X', 4, 3), i \in [n-15]\}$$
$$\mathcal{F}_2 = \{\{x_i\} \cup T : T \in \mathcal{S}_i(Y, 4, 3), i \in [n-3]\}.$$

Claim. There exists (2n-3,5,4)-saturated $\mathcal{F}' \subset {\binom{[2n-3]}{5}}$ with $\mathcal{F} \subset \mathcal{F}'$ and $|\mathcal{F}'| = |\mathcal{F}| + O(n^3)$.

Proof. As \mathcal{F} is not (2n-3, 5, 4)-saturated, we greedily add edges until our new set system \mathcal{F}' is (2n-3, 5, 4)-saturated. It then suffices to show $|\mathcal{F}'| = |\mathcal{F}| + O(n^3)$. First, it follows that we may not add any A with $A \cap S = \emptyset$ by considering the cases of $|A \cap X'|$ and $|A \cap Y|$. Next, there are $\Theta(n^3)$ edges with $|A \cap S| \ge 2$, and so it suffices to consider the case where $|A \cap S| = 1$. It then follows that the only edges we may add are so that $|A \cap S| = 1$, $|A \cap X'| = 2$, and $|A \cap Y'| = 2$. For $s \in S$ and $e \in {X' \choose 2}$, let $G_{s,e} = \{f \in {Y' \choose 2} : \{s\} \cup e \cup f \in \mathcal{F}'\}$ be the link graph of $\{s\} \cup e$ in \mathcal{F}' . Since \mathcal{F}' does not contain an intersection of size four, $G_{s,e}$ is necessarily a graph matching so

$$|\mathcal{F}' \setminus \mathcal{F}| = \sum_{s \in S} \sum_{e \in \binom{X'}{2}} |G_{s,e}| \le |S| \cdot \binom{n}{2} \cdot \frac{n}{2} = \Theta(n^3).$$

When $n = i \pmod{12}$, letting $\mathcal{F} \subset \mathcal{F}' \subset {\binom{[n-i]}{4}}$ be as in Construction 2 and the above claim, and take $\mathcal{F}'' \subset {\binom{[n]}{5}}$ to be an (n, 5, 4)-saturated set system which contains \mathcal{F}' . Setting S = [n - i + 1, n], there are $O(n^3)$ edges in \mathcal{F}'' with $|A \cap S| \geq 2$ and for each $s \in S$, it follows that $|\mathcal{F}(\{s\})| = O(n^3)$ as it is (n - i, 4, 3)-saturated by the result of Frankl-Füredi [FF87].

6.7 Proof of Theorem 1.6.6

In this section, we construct a (n, 7, 5)-saturated set system which gives $w(n, 7, 5) \le 1/2$:

Construction 3. Let X_1 and X_2 be a balanced partition of [n]. Set $\mathcal{F}_0 = \mathcal{S}(X_1, 7, 5) \sqcup$ $\mathcal{S}(X_2, 7, 5)$. Let $\mathcal{S}_1 \subset \partial_6(\mathcal{S}(X_1, 7, 5))$ and $\mathcal{S}_2 \subset \partial_6(\mathcal{S}(X_2, 7, 5))$ be as in Lemma 6.2.2. Then define \mathcal{F}_1 by

$$\mathcal{F}_1 := \{\{x\} \cup A : x \in X_1; B \in \mathcal{S}_2\} \sqcup \{\{x\} \cup A : x \in X_2; B \in \mathcal{S}_1\}.$$

Claim. There exists a (n, 7, 5)-saturated $\mathcal{F} \supset \mathcal{F}_0 \cup \mathcal{F}_1$ with $|\mathcal{F}| \leq |\mathcal{F}_0| + |\mathcal{F}_1| + o(n^5)$.

Proof. As $\mathcal{F}_0 \cup \mathcal{F}_1$ is not (n, 7, 5)-saturated, we greedily add edges until our new set system \mathcal{F} is (n, 7, 5)-saturated. The only sets $A \in {\binom{[n]}{7}}$ which we could have added to $\mathcal{F}_0 \cup \mathcal{F}_1$ are so that $|A \cap X_1| \in \{1, 3, 4, 6\}$. It suffices to show that $|\mathcal{F}| \leq |\mathcal{F}_0| + |\mathcal{F}_1| + o(n^5)$.

By symmetry, it suffices to consider the cases where $|A \cap X_1| = 1$ and $|A \cap X_1| = 4$. Fix $x \in X_1$, then $\mathcal{F} \setminus (\mathcal{F}_0 \cup \mathcal{F}_1)(x) = o(n^4)$ by the construction of \mathcal{S}_2 . Hence, the number of edges with $|A \cap X_1| = 1$ we can add to $\mathcal{F}_0 \cup \mathcal{F}_1$ is $o(n^5)$. We now consider edges of the form $|A \cap X_1| = 4$. If $A \cap X_1 \in \partial_4(\mathcal{S}_1)$, then it is not hard to see that there exists a $B \in \mathcal{F}_1$ with $|A \cap B| = 5$. If $A \cap X_1 \notin \partial_4(\mathcal{S}_2)$, then consider the link of A in \mathcal{F} . This is a 3-uniform set system which does not contain an intersection of size one. By oddtown rules, $|\mathcal{F}(A)| \leq n$ and as there are $o(n^4)$ sets $A' \notin \partial_4(\mathcal{S}_1)$, it follows that there are at most $o(n^5)$ edges of the form $|A \cap X_1| = 4$ we can add to $\mathcal{F}_0 \cup \mathcal{F}_1$.

For other large values of n, we may argue similarly as in the end of Section 6.6 to get the desired upper bound for w(n, 7, 5). This construction can also be adapted when (k, t) = (6, 4) which gives $w^*(n, 6, 4) \leq 3/4$.

This chapter contains material from: J. O'Neill, J. Verstraëte, "A note on intersection saturation", which is currently being prepared for submission for publication. The dissertation author was one of the primary investigators and authors of this paper.

References

- [ABS91] Noga Alon, László Babai, and Nevin M. Singhi, Multilinear polynomials and Frankl-Ray-Chaudhuri-Wilson type intersection theorems, J. Combin. Theory Ser. A 58 (1991), 165–180.
- [AK85] Noga Alon and Gil Kalai, A simple proof of the upper bound theorem, European J. Combin. 6 (1985), 211–214.
- [Alo86] Noga Alon, Decomposition of the complete r-graph into complete r-partite r-graphs, Graphs and Combinatorics **2** (1986), 95–100.
- [AMMV12] Noga Alon, Keith Mellinger, Dhruv Mubayi, and Jacques Verstraëte, The de Bruijn-Erdős theorem for hypergraphs, Designs, Codes and Cryptography 65 (2012), 233– 245.
- [BD71] Alan Brace and David E. Daykin, A finite set covering theorem, Bulletin of the Australian Mathematical Society 5 (1971), 197–202.
- [Ber69] Elwyn R. Berlekamp, On subsets with intersections of even cardinality, Canad. Math. Bull **12** (1969), 471–477.
- [BF20] László Babai and Peter Frankl, *Linear algebra methods in combinatorics*, vol. Preliminary Version 2.1, The University of Chicago, 2020.
- [Bol65] Bela Bollobás, On generalized graphs, Acta Math. Acad. Sci. Hungar. 16 (1965), 447–452.
- [CKV09] Sebastian Cioabă, André Kündgen, and Jacques Verstraëte, On decompositions of complete hypergraphs, Journal of Combin. Ser. A (2009), 1232–1234.
- [dBE48] Nicolaas Govert de Bruijn and Paul Erdős, On a combinatorial problem, Nederl. Akad. Wetensch., Proc. **51** (1948), 421–423.
- [DEF78] Michel Deza, Paul Erdős, and Peter Frankl, Intersection properties of systems of finite sets, Proc. London Math. Soc. **36** (1978), 369–384.

- [DFS83] Michel Deza, Peter Frankl, and Nevin M. Singhi, On functions of strength t, Combinatorica **3** (1983), 331–339.
- [EHM64] Paul Erdős, András Hajnal, and John Moon, A problem in graph theory, Amer. Math. Monthly 71 (1964), 1107–1110.
- [EKR61] Paul Erdős, Chao Ko, and Richard Rado, Intersection theorems for systems of finite sets, The Quarterly Journal of Mathematics 12 (1961), 313–320.
- [Erd62] Paul Erdős, On a theorem of Rademacher-Turán, Illinois J. Math. (1962), 122–127.
- [F84] Zoltán Füredi, Geometrical solution of a problem for two hypergraphs, European J.
 Combin. 5 (1984), 133–136.
- [F15] Zoltán Füredi, A proof of the stability of extremal graphs, Simonovits' stability from Szemerédi's regularity, J. Combinatorial Theory Ser. B 115 (2015), 66–71.
- [FF87] Peter Frankl and Zoltán Füredi, Exact solution of some Turán-type problems, J. Combin. Theory Ser. A 45 (1987), 226–262.
- [FFS11] Jill Faudree, Ralph Faudree, and John Schmitt, A survey of minimum saturated graphs, Electronic J. of Combinatorics 18 (2011), ds19.
- [FHHZ18] Peter Frankl, Hao Huang, Jie Han, and Yi Zhao, A degree version of the Hilton-Milner theorem, J. Combin. Theory Ser. A 155 (2018), 493–502.
- [FJS14] Zoltán Füredi, Tao Jiang, and Robert Seiver, *Exact solution of the hypergraph Turán* problem for k-uniform linear paths, Combinatorica **34** (2014), 299–322.
- [FK84] Michael Fredman and János Komlós, On the size of separating systems and families of perfect hash functions, SIAM Journal on Algebraic Discrete Methods 5 (1984), 61–68.
- [FK18] Peter Frankl and Andrey Kupavskii, *Counting intersecting and pairs of crossintersecting families*, Combin. Probab. Comput. **27** (2018), 60–68.
- [FO83] Peter Frankl and Andrew M. Odlyzko, On subsets with cardinalities of intersections divisible by a fixed integer, European J. Combin. 4 (1983), 215–220.
- [FR85] Peter Frankl and Vojtěch Rödl, Near perfect coverings in graphs and hypergraphs, European J. Combinatorics 6 (1985), 317–326.
- [Fra82] Peter Frankl, An extremal problem for two families of sets, European J. Combin. **3** (1982), 125–127.
- [Fra13] _____, Improved bounds on the Erdős matching conjecture, J. Combin. Theory Ser. A **120** (2013), 1068–1072.
- [FS04] Zoltan Füredi and Benny Sudakov, Extremal set systems with restricted k-wise intersections, J. Combin. Theory Ser. A 105 (2004), 143–159.

- [FT18] Peter Frankl and Norihide Tokushige, *Extremal problems for finite sets*, American Mathematical Soc., 2018.
- [Für83] Zoltán Füredi, On finite set-systems whose every intersection is a kernel of a star, Discrete Math. 47 (1983), 129–132.
- [Für14] _____, Linear trees in uniform hypergraphs, European J. Combin. **35** (2014), 264–272.
- [FW81] Peter Frankl and Richard Wilson, Intersection theorems with geometric consequences, Combinatorica 1 (1981), 357–368.
- [GKLO20] Stefan Glock, Daniela Kuhn, Allan Lo, and Deryk Osthus, The existence of designs via iterative absorption: hypergraph f-designs for arbitrary f, Memoirs of the American Mathematical Society (2020), 1–95.
- [GP72] Ronald L. Graham and Henry O. Pollak, On embedding graphs in squashed cubes, In: Graph Theory and Appl., Springer Lecture Notes in Math. 303 (1972), 99–110.
- [GR19] Venkatesan Guruswami and Andrii Riazanov, Beating Fredman-Komlós for perfect k-hashing, ICALP, 2019.
- [Gra75] Jack Graver, Boolean designs and self-dual matroids, Linear Algebra and Appl. 10 (1975), 111–128.
- [GS02] Vince Grolmusz and Benny Sudakov, On k-wise set-intersections and k-wise hamming-distances, J. Combin. Theory Ser. A **99** (2002), 180–190.
- [Han64] Georges Hansel, Nombre minimal de contacts de fermature nécessaires pour réaliser une fonction booléenne symétrique de n variables, C. R. Acad. Sci. Paris (1964), 6037–6040.
- [HK17] Jie Han and Yoshiharu Kohayakawa, The maximum size of a non-trivial intersecting uniform family that is not a subfamily of the Hilton-Milner family, Proc. Amer. Math. Soc. 145 (2017), 73–87.
- [HM67] Anthony Hilton and Eric C. Milner, Some intersection theorems for systems of finite sets, Quart. J. Math. Oxford Ser. 18 (1967), 369–384.
- [JP71] Francois Jaeger and Charles Payan, Nombre maximal d'arêtes d'un hypergraphe critique de rang h, C. R. Acad. Sci. Paris (1971), 221–223.
- [JX83] Lu Jia-Xi, On large sets of disjoint Steiner triple systems i, ii, and iii, J. Combinatorial Theory Ser. A 34 (1983), 147–155.
- [Kat72] Gyula Katona, A simple proof of the Erdős-Ko-Rado theorem, Journal of Combinatorial Theory, Series B 13 (1972), 183–184.
- [Kat74] Gyula O. H. Katona, Solution of a problem of Ehrenfeucht and Mycielski, J. Combin. Theory, Ser. A (1974), 265–266.

- [Kee11] Peter Keevash, *Hypergraph turán problems*, In R. Chapman (Ed.), Surveys in Combinatorics (London Mathematical Society Lecture Note Series (2011), 83–140.
- [Kee18] _____, The existence of designs i and ii, arxiv:1401.3665 and arxiv:1802.05900, 2018.
- [KKK15] Dong Yeap Kang, Jaehoon Kim, and Younjin Kim, On the Erdős-Ko-Rado theorem and the Bollobás theorem for t-intersecting families, European J. Combin. 47 (2015), 68–74.
- [KM88] János Körner and Katalin Marton, New bounds for perfect hashing via information theory, European J. Combin. (1988), 523–530.
- [KM17] Alexandr Kostochka and Dhruv Mubayi, The structure of large intersecting families, Proc. Amer. Math. Soc. 145 (2017), 2311–2321.
- [Kup19] Andrey Kupavskii, Structure and properties of large intersecting families, arxiv:1810.00920, 2019.
- [LMT18] Imre Leader, Luka Milićević, and Ta Shang Tan, *Decomposing the complete r-graph*, Journal of Combinatorial Theory, Series A **154** (2018), 21–31.
- [Lov77a] László Lovász, Flats in matroids and geometric graphs, Combinatorial Surveys (P. J. Cameron, ed.), Academic Press, New York (1977), 45–86.
- [Lov77b] _____, Topological and algebraic methods in graph theory, Graph Theory and Related Topics (J. A. Bondy and U. S. R. Murty, eds.) (1977), 1–15.
- [Lov07] _____, Combinatorial problems and exercises, AMS Chelsea Pub., 2007.
- [Man07] Willem Mantel, *Problem 28*, Wiskundige Opgaven (907), 60–61.
- [MS15] Guy Moshkovitz and Asaf Shapira, *Exact bounds for some hypergraph saturation* problems, J. Combin. Theory Ser. B **111** (2015), 242–248.
- [MV16] Dhruv Mubayi and Jacques Verstraete, A survey of Turán problems for expansions. in: Beveridge a., griggs j., hogben l., musiker g., tetali p. (eds) recent trends in combinatorics., vol. 159, The IMA Volumes in Mathematics and its Applications, 2016.
- [O'N21] Jason O'Neill, Towards supersaturation for oddtown and eventown, arxiv:2109.09925, 2021.
- [Orl77] James Orlin, Contentment in graph theory: Covering graphs with cliques, Indagationes Mathematicae (Proceedings) 80 (1977), 406–424.
- [OV20a] Jason O'Neill and Jacques Verstraëte, Intersection saturated set systems, arxiv:2011.09402 (2020).
- [OV20b] _____, A note on k-wise oddtown problems, arxiv:2011.09402, 2020.

[OV21a]	, A generalization of the Bollobás set pairs inequality, Electronic Journal of Combinatorics (2021), 3.8.
[OV21b]	, Non-trivial d-wise intersecting families, Journal of Combinatorial Theory, Ser. A 178 (2021).
[Pik01]	Oleg Pikhurko, Construction designs straightforwardly: Worst arising cases, J. Combin. Designs 9 (2001), 100–106.
[Pyb86]	László Pyber, A new generalization of the Erdős–Ko–Rado theorem, J. Combin. Theory, Ser. A 43 (1986), 85–90.
[Sid81]	Alexander Sidorenko, Systems of sets that have the t-property, Moscow University Mathematics Bulletin 36 (1981), 22–26.
[Sid95]	, What we know and what we do not know about Turán numbers, Graphs and Combinatorics 11 (1995), 179–199.
[Sid21]	, On Turán numbers of the complete 4-graphs, arxiv:2009.12955, 2021.
[Sim95]	Gabor Simonyi, <i>Graph entropy: A survey</i> , Combinatorial Optimization, W. Cook, L. Lovász, P. Seymour, Eds., DIMACS Series in Discrete Mathematics and Theoretical Computer Science (1995), 399–441.
[Spe28]	Emanuel Sperner, <i>Ein satz "uber untermengen einer endlichen menge</i> , Mathema- tische Zeitschrift 27 (1928), 544–548.
[SV05]	Tibor Szabó and Van Vu, <i>Exact k-wise intersection theorems</i> , Graphs and Combinatorics 21 (2005), 247–261.
[SV18]	Benny Sudakov and Pedro Vieira, <i>Two remarks on eventown and oddtown problems</i> , SIAM J. of Discrete Math. 32 (2018), 280–295.
[Tal04]	John Talbot, A new Bollobás-type inequality and applications to t-intersecting fam- ilies of sets, Discrete Mathematics 285 (2004), 349–353.
[Tar75]	Tamás G. Tarjan, Complexity of lattice-configurations, Studia Sci. Math. Hungar. 10 (1975), 203–211.
[Tie91]	Luc Tierlinck, A completion of Lu's determination of the spectrum for large sets of disjoint Steiner triple systems, J. Combinatorial Theory Ser. A 57 (1991), 302–305.
[Tur41]	Pál Turán, On an extremal problem in graph theory, Mat. Fiz. Lapok 48 (1941), 436–452.
[Tuz85]	Zsolt Tuza, Critical hypergraphs and intersecting set-pair systems, J. Combin. The- ory Ser. B 39 (1985), 134–145.
[Tuz94]	, Applications of the set-pair method in extremal hypergraph theory, Extremal Problems for Finite Sets, Bolyai Society Mathematical Studies, Vol. 3, János Bolyai Mathematical Society (1994), 479–514.

[Tuz96]	, Applications of the set-pair method in extremal problems, ii, Combinatorics, Paul Erdős is Eighty, Bolyai Society Mathematical Studies 2 (1996), 459–490.
[Vu97]	Van Vu, Extremal systems with upper-bounded odd intersections, Graphs and Combinatorics 13 (1997), 197–208.
[Vu99]	<u> </u>
[Wil72]	Richard Wilson, An existence theory for pairwise balanced designs, i and ii, J. Combinatorial Theory Ser. A 13 (1972), 220–273.
[Wil90]	, A diagonal form for the incidence matrices of t-subsets vs. k-subsets, European Journal of Combinatorics 11 (1990), 609–615.